UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

-

FORM 10-Q

[**ü**]Quarterly Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 **For the Quarterly Period Ended <u>June 30, 2008</u>.**

[]Transition Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 For the Transition Period From $_$ to $_$.

Commission file number <u>1-8400</u>.

AMR Corporation

(Exact name of registrant as specified in its charter)

 Delaware
 75-1825172

 (State or other jurisdiction of incorporation or organization)
 (I.R.S. Employer Identification No.)

 4333 Amon Carter Blvd. Fort Worth, Texas
 76155

 (Address of principal executive offices)
 (Zip Code)

 Registrant's telephone number, including area code
 (817) 963-1234

 Not Applicable
 Image: Commer address and former fiscal year , if changed since last report)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. \square Yes \square No

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, or a non-accelerated filer. See definition of "accelerated filer" and "large accelerated filer" in Rule 12b-2 of the Exchange Act. 🛛 Large Accelerated Filer 🗌 Accelerated Filer 🗌 Non-accelerated Filer

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). 🗆 Yes 🗵 No

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

Common Stock, \$1 par value - 251,147,538 shares as of July 14, 2008.

INDEX

AMR CORPORATION

PART I:FINANCIAL INFORMATION

Item 1. Financial Statements

Consolidated Statements of Operations -- Three and six months ended June 30, 2008 and 2007

Condensed Consolidated Balance Sheets -- June 30, 2008 and December 31, 2007

Condensed Consolidated Statements of Cash Flows -- Six months ended June 30, 2008 and 2007

Notes to Condensed Consolidated Financial Statements -- June 30, 2008

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Item 3. Quantitative and Qualitative Disclosures about Market Risk

Item 4. Controls and Procedures

PART II: OTHER INFORMATION

- Item 1. Legal Proceedings
- Item 4. Submission of Matters to a Vote of Security Holders
- Item 5. Other Information
- Item 6. Exhibits

SIGNATURE

Item 1. Financial Statements

AMR CORPORATION

CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited) (In millions, except per share amounts)

	Three Months Ended June 30,					Six Months Ended June 30,		
		2008	2	007		2008		2007
Revenues								
Passenger – American Airlines	\$	4,735	\$	4,525	\$	9,114	\$	8,701
- Regional Affiliates		683		658		1,264		1,216
Cargo		233		200		448		401
Other revenues		528		496		1,050		988
Total operating revenues		6,179		5,879		11,876		11,306
Expenses								
Aircraft fuel		2,423		1,644		4,473		3,054
Wages, salaries and benefits		1,658		1,655		3,302		3,326
Other rentals and landing fees		318		313		641		642
Depreciation and amortization		324		295		633		585
Maintenance, materials and repairs		323		268		638		516
Commissions, booking fees and credit card expense		259		268		516		517
Aircraft rentals		125		152		250		303
Food service		133		133		260		260
Special charges		1,164		-		1,164		-
Other operating expenses		742		684		1,476		1,388
Total operating expenses		7,469		5,412		13,353		10,591
Operating Income (Loss)		(1,290)		467		(1,477)		715
Other Income (Expense)								
Interest income		48		90		101		167
Interest expense		(185)		(235)		(379)		(476)
Interest capitalized		8		5		13		14
Miscellaneous – net		(29)		(10)		(34)		(22)
		(158)		(150)		(299)		(317)
Income (Loss) Before Income Taxes Income tax		(1,448)		317		(1,776)		398
	¢	(1.440)	¢	-	¢	(1.770)	¢	
Net Earnings (Loss)	\$	(1,448)	\$	317	\$	(1,776)	\$	398
Earnings (Loss) Per Share								
Basic	\$	(5.77)	\$	1.28	\$	(7.10)	\$	1.65
		()				(
Diluted	\$	(5.77)	\$	1.08	\$	(7.10)	\$	1.38

The accompanying notes are an integral part of these financial statements.

AMR CORPORATION CONDENSED CONSOLIDATED BALANCE SHEETS (Unaudited) (In millions)

	June 30, 2008	December 31, 2007
Assets		
Current Assets		
Cash	\$ 287	\$ 148
Short-term investments	4,782	4,387
Restricted cash and short-term investments	434	428
Receivables, net	1,166	1,027
Inventories, net	702	601
Fuel derivative contracts	1,282	416
Other current assets	419	222
Total current assets	9,072	7,229
Equipment and Property		
Flight equipment, net	12,702	13,977
Other equipment and property, net	2,390	2,413
Purchase deposits for flight equipment	483	241
	15,575	16,631
Equipment and Property Under Capital Leases		
Flight equipment, net	340	686
Other equipment and property, net	67	77
	407	763
Route acquisition costs and airport operating and gate lease rights, net	1,123	1,156
Other assets	2,748	2,792
	\$ 28,925	\$ 28,571
Linkiliting and Stankholdows' Equity		
Liabilities and Stockholders' Equity Current Liabilities		
Accounts payable	\$ 1,389	\$ 1,182
Accrued liabilities	2,804	2,267
Air traffic liability	4,889	3,985
Current maturities of long-term debt	1,425	902
Current obligations under capital leases	133	147
Total current liabilities	10,640	8,483
Long-term debt, less current maturities	8,708	9,413
Obligations under capital leases, less current obligations	623	680
Pension and postretirement benefits	3,670	3,620
Other liabilities, deferred gains and deferred credits	3,624	3,718
Stockholders' Equity		
Preferred stock	-	-
Common stock	257	255
Additional paid-in capital	3,491	3,489
Treasury stock	(367)	(367)
Accumulated other comprehensive income	1,445	670
Accumulated deficit	(3,166)	(1,390)
	1,660	2,657
	\$ 28,925	\$ 28,571
The accompanying notes are an integral part of these financial statements.		

AMR CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (<u>Unaudited</u>) (In millions)

	Six Months Ended June 30				
	2008			2007	
Net Cash Provided by Operating Activities	\$	1,154	\$	1,743	
Cash Flow from Investing Activities:					
Capital expenditures		(473)		(364)	
Net increase in short-term investments		(395)		(1,091)	
Net increase in restricted cash and short-term investments		(6)		(2)	
Proceeds from sale of equipment and property		9		23	
Other		8		5	
Net cash used by investing activities		(857)		(1,429)	
Cash Flow from Financing Activities:					
Payments on long-term debt and capital lease obligations		(379)		(862)	
Proceeds from:					
Issuance of long-term debt		70		-	
Sale leaseback transactions		151		-	
Issuance of common stock, net of issuance costs		-		497	
Reimbursement from construction reserve account		-		59	
Exercise of stock options		-		86	
Net cash used for financing activities		(158)		(220)	
Net increase in cash		139		94	
Cash at beginning of period		148		121	
Cash at end of period	\$	287	\$	215	

The accompanying notes are an integral part of these financial statements.

AMR CORPORATION NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

- 1. The accompanying unaudited condensed consolidated financial statements have been prepared in accordance with generally accepted accounting principles for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements. In the opinion of management, these financial statements contain all adjustments, consisting of normal recurring accruals, necessary to present fairly the financial position, results of operations and cash flows for the periods indicated. Results of operations for the periods presented herein are not necessarily indicative of results of operations for the entire year. The condensed consolidated financial statements include the accounts of AMR Corporation (AMR or the Company) and its wholly owned subsidiaries, including (i) its principal subsidiary American Airlines, Inc. (American) and (ii) its regional airline subsidiary, AMR Eagle Holding Corporation and its primary subsidiaries, American Eagle Airlines, Inc. and Executive Airlines, Inc. (collectively, AMR Eagle). The condensed consolidated financial statements also include the accounts of variable interest entities for which the Company is the primary beneficiary. For further information, refer to the consolidated financial statements and footnotes thereto included in the AMR Annual Report on Form 10-K for the year ended December 31, 2007 (2007 Form 10-K).
- 2. Beginning in the first quarter of 2008, AMR reclassified revenues associated with the marketing component of AAdvantage program mileage sales from Passenger revenue to Other revenue. As a result of this change, approximately \$148 million and \$298 million of revenue was reclassified from Passenger revenue to Other revenue for the three and six months ended June 30, 2007, respectively, to conform to the current presentation.
- 3. As of June 30, 2008, the Company had commitments to acquire 33 Boeing 737-800 aircraft in 2009, seven Boeing 737-800 aircraft in 2010 and an aggregate of 20 Boeing 737 aircraft and seven Boeing 777 aircraft in 2013 through 2016. Payments will approximate \$278 million in the remainder of 2008, \$682 million in 2009, \$107 million in 2010, \$102 million in 2011, \$310 million in 2012, and \$1.3 billion for 2013 and beyond. These amounts are net of purchase deposits currently held by the manufacturer. However, if as anticipated, the Company commits to accelerating the delivery dates of a significant number of aircraft in the future, a significant portion of the \$1.7 billion commitment from 2011 and beyond will be accelerated into 2008, 2009 and 2010. In addition, any incremental aircraft orders will increase the Company's commitments.

On December 18, 2007, the European Commission issued a Statement of Objection ("SO") against 26 airlines, including the Company. The SO alleges that these carriers participated in a conspiracy to set surcharges on cargo shipments in violation of EU law. The SO states that, in the event that the allegations in the SO are affirmed, the Commission will impose fines against the Company. The Company intends to vigorously contest the allegations and findings in the SO under EU laws, and it intends to cooperate fully with all other pending investigations. The evaluation of these allegations is still in the early stages, but based on the information to date, the Company has not recorded any reserve for this exposure for the quarter ended June 30, 2008. In the event that the SO is affirmed or other investigations uncover violations of the U.S. antitrust laws or the competition laws of some other jurisdiction, or if the Company were named and found liable in any litigation based on these allegations, such findings and related legal proceedings could have a material adverse impact on the Company.

4. Accumulated depreciation of owned equipment and property at June 30, 2008 and December 31, 2007 was \$9.9 billion and \$11.9 billion, respectively. Accumulated amortization of equipment and property under capital leases at June 30, 2008 and December 31, 2007 was \$591 million and \$1.2 billion, respectively. During the second quarter of 2008, the Company recorded an impairment charge to write down its McDonnell Douglas MD80 and Embraer RJ-135 fleets and certain related long-lived assets to their estimated fair values. As a result \$2.8 billion of accumulated depreciation and amortization was eliminated as a new cost basis was established for these aircraft. See Note 9 to the condensed consolidated financial statements for more information regarding the impairment charges.

- 5. As discussed in Note 7 to the consolidated financial statements in the 2007 Form 10-K, the Company has a valuation allowance against the full amount of its net deferred tax asset. The Company currently provides a valuation allowance against deferred tax assets when it is more likely than not that some portion, or all of its deferred tax assets, will not be realized. The Company's deferred tax asset valuation allowance increased approximately \$324 million during the six months ended June 30, 2008 to \$949 million as of June 30, 2008, including the impact of comprehensive loss for the six months ended June 30, 2008 and changes from other adjustments.
- 6. As of June 30, 2008, AMR had issued guarantees covering approximately \$1.4 billion of American's tax-exempt bond debt and American had issued guarantees covering approximately \$1.1 billion of AMR's unsecured debt. In addition, as of June 30, 2008, AMR and American had issued guarantees covering approximately \$327 million of AMR Eagle's secured debt and AMR has issued guarantees covering an additional \$2.2 billion of AMR Eagle's secured debt.

As discussed in Note 6 to the consolidated financial statements in the 2007 Form 10-K, the Company also has outstanding \$324 million principal amount of its 4.50 percent senior convertible notes due 2024 (the 4.50 Notes) and \$300 million principal amount of its 4.25 percent senior convertible notes due 2023 (the 4.25 Notes). On each of February 15, 2009 for the 4.50 Notes, and on September 23, 2008 for the 4.25 Notes, and then again at certain later dates, the holders may require the Company to purchase all or a portion of their notes at a price equal to 100% of their principal amount plus unpaid interest which may be paid in cash, common stock or a combination of cash and common stock. Accordingly, the Company has classified both the \$324 million principal and the \$300 million principal amounts of the 4.50 Notes and the 4.25 Notes, respectively, into Current maturities of long term debt as of June 30, 2008 as a result of the existence of these put provisions. The Company is evaluating various payment and refinancing alternatives for the outstanding 4.25% notes upon the expected exercise of the put provision in September 2008. Based on current economic and market conditions, the Company currently expects to settle the 4.25 Notes in cash.

In May 2008, the Financial Accounting Standards Board (FASB) affirmed the consensus of FASB Staff Position APB 14-1 (FSP APB 14-1), "Accounting for Convertible Debt Instruments That May Be Settled in Cash upon Conversion (Including Partial Cash Settlement)", which applies to all convertible debt instruments that have a "net settlement feature", which means that such convertible debt instruments, by their terms, may be settled either wholly or partially in cash upon conversion. FSP APB 14-1 requires issuers of convertible debt instruments that may be settled wholly or partially in cash upon conversion to separately account for the liability and equity components in a manner reflective of the issuers' nonconvertible debt borrowing rate. FSP APB 14-1 is effective for financial statements issued for fiscal years beginning after December 15, 2008 and interim periods within those fiscal years. Early adoption is not permitted and retroactive application to all periods presented is required. The Company is currently evaluating the earnings impact that adoption of FSP APB 14-1 will have on its consolidated financial statements; however the Company expects the adoption to reduce reported earnings.

American has a secured bank credit facility which consists of an undrawn \$255 million revolving credit facility, with a final maturity on June 17, 2009, and a fully drawn \$438 million term loan facility, with a final maturity on December 17, 2010 (the Revolving Facility and the Term Loan Facility, respectively, and collectively, the Credit Facility). The Credit Facility contains a covenant (the EBITDAR Covenant) requiring AMR to maintain a ratio of cash flow to fixed charges. In May 2008, AMR and American entered into an amendment to the Credit Facility which waived compliance with the EBITDAR Covenant for periods ending on any date from and including June 30, 2008 through March 31, 2009, and which reduced the minimum ratios AMR is required to satisfy thereafter. The required ratio will be 0.90 to 1.00 for the one quarter period ending June 30, 2009 and will increase to 1.15 to 1.00 for the four quarter period ending September 30, 2010.

On July 8, 2008, the Company raised approximately \$500 million under a loan secured by aircraft, due in installments through 2015.

7. In September 2006, the Financial Accounting Standards Board (FASB) issued Statement of Financial Accounting Standards No. 157 "Fair Value Measurements" (SFAS 157). SFAS 157 introduces a framework for measuring fair value and expands required disclosure about fair value measurements of assets and liabilities. SFAS 157 for financial assets and liabilities is effective for fiscal years beginning after November 15, 2007, and the Company has adopted the standard for those assets and liabilities as of January 1, 2008 and the impact of adoption was not significant.

SFAS 157 defines fair value as the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. SFAS 157 also establishes a fair value hierarchy which requires an entity to maximize the use of observable inputs and minimize the use of unobservable inputs when measuring fair value. The standard describes three levels of inputs that may be used to measure fair value:

Level 1 - Quoted prices in active markets for identical assets or liabilities.

Level 2 - Observable inputs other than Level 1 prices such as quoted prices for similar assets or liabilities; quoted prices in markets that are not active; or other inputs that are observable or can be corroborated by observable market data for substantially the full term of the assets or liabilities. The Company's short-term investments primarily utilize broker quotes in a non-active market for valuation of these securities. The Company's fuel derivative contracts, which primarily consist of commodity options and collars, are valued using energy and commodity market data which is derived by combining raw inputs with quantitative models and processes to generate forward curves and volatilities.

Level 3 - Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets or liabilities.

The Company utilizes the market approach to measure fair value for its financial assets and liabilities. The market approach uses prices and other relevant information generated by market transactions involving identical or comparable assets or liabilities.

Assets and liabilities measured at fair value on a recurring basis are summarized below:

(in millions)	Fair Value Measurements as of June 30, 2008					1	
Description	Total	Total		Level 2		Level 3	
Short term investments ¹	\$ 4,78	2 \$	1,525	\$	3,257	\$	-
Restricted cash and short-term investments ¹	43	4	434		-		-
Fuel derivative contracts ¹	1,28	2	-		1,282		-
				_			
Total	\$ 6,49	8 \$	1,959	\$	4,539	\$	-

¹Unrealized gains or losses on short term investments, restricted cash and short-term investments and derivatives are recorded in Accumulated other comprehensive income (loss) at each measurement date.

8. The following tables provide the components of net periodic benefit cost for the three and six months ended June 30, 2008 and 2007 (in millions):

		Pension Benefits						
	Th	Three Months Ended June 30,			Six Months Ended June 30,			ed
	2008			2007		2008		2007
Components of net periodic benefit cost								
Service cost Interest cost Expected return on assets Amortization of:	\$	81 171 (197)	\$	93 168 (187)	\$	162 342 (395)	\$	185 336 (374)
Prior service cost Unrecognized net loss		4 1		4 6		8 1		8 13
Net periodic benefit cost	\$	60	\$	84	\$	118	\$	168
	Retiree Medical an				nd Other Benefits			
	Three Months Ended June 30,		Six Months E June 30,			ed		
	2008			2007		2008		2007

Components of net periodic benefit cost				
Service cost	\$ 14 \$	18 \$	27 \$	35
Interest cost	43	49	86	96
Expected return on assets	(5)	(5)	(10)	(9)
Amortization of:				
Prior service cost	(3)	(3)	(7)	(7)
Unrecognized net (gain) loss	 (6)	(2)	(12)	(4)
Net periodic benefit cost	\$ 43 \$	57 \$	84 \$	111

The Company's 2008 minimum required contribution to its defined benefit pension plan is \$78 million. As of June 30, 2008, the Company has contributed \$75 million to its defined benefit pension plans. On July 15, the Company made an additional contribution of \$3 million bringing year to date contributions to the minimum contribution of \$78 million. Absent a significant change in the industry environment, it is unlikely that additional contributions will be made to the Company's defined benefit pension plan in 2008.

9. In accordance with Statement of Financial Accounting Standards No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets" (SFAS 144), the Company records impairment charges on long-lived assets used in operations when events and circumstances indicate that the assets may be impaired, the undiscounted cash flows estimated to be generated by those assets are less than the carrying amount of those assets and the net book value of the assets exceeds their estimated fair value. In May 2008, the Company announced capacity reductions due to unprecedented high fuel costs and the other challenges facing the industry. In connection with these capacity reductions, the Company concluded that a triggering event had occurred and required a test for impairment. As a result of this test, the Company concluded the carrying values of its McDonnell Douglas MD-80 and the Embraer RJ-135 aircraft fleets were no longer recoverable. Consequently, during the second quarter of 2008, the Company recorded an impairment charge of \$1.1 billion to write these and certain related long-lived assets down to their estimated fair values. No portion of the impairment charge will result in future cash expenditures. All other fleet types were tested for impairment but were concluded to be recoverable with projected undiscounted cash flows or had fair values at levels above current carrying value. Included in the charge for the Embraer RJ-135 fleet were write downs on 29 aircraft that were considered held for sale as of June 30, 2008. The McDonnell Douglas MD-80 aircraft will be depreciated over their remaining useful lives averaging approximately five years.

In determining the asset recoverability, management estimated the undiscounted future cash flows utilizing models used by the Company in making fleet and scheduling decisions. In determining fair market value, the Company utilized recent external appraisals of its fleets, a published aircraft pricing survey and recent transactions involving sales of similar aircraft, adjusted based on estimates of maintenance status and to consider the impact of recent industry events on these values. As a result of the write down of these aircraft to fair value, as well as the acceleration of the retirement dates, depreciation expense is expected to decrease by approximately \$156 million on an annualized basis.

In conjunction with the capacity reductions, the Company estimates that it will reduce its workforce commensurate with the announced system-wide capacity reductions. This reduction in workforce will be accomplished through various measures, including voluntary programs, part-time work schedules, furloughs in accordance with collective bargaining agreements, and other reductions. As a result of this reduction in workforce the Company will incur employee charges of approximately \$70 million for severance related costs of which \$55 million was recorded as of June 30, 2008 in accordance with Statement of Financial Accounting Standards No. 112, "Employers Accounting for Postemployment Benefits" (SFAS 112), based on probable expectations of involuntary terminations.

The Company expects to record other accounting charges relating to these capacity reductions, including disposal costs, additional voluntary severance, facility exit costs, costs of grounding leased Airbus A300 aircraft prior to lease expiration and other associated costs, but, at this time is not able to reasonably estimate the amount and timing of the charges or the portion, if any, of these charges that would result in future cash expenditures.

The following table summarizes the components of the Company's special charges, the remaining accruals for these charges and the June 30, 2008 capacity reduction related charges (in millions):

	Aircra	aft Charges	Facility Exit Costs	Employee Charges	Other	Total
Remaining accrual at December 31, 2007	\$	126	\$ 18	\$ -	\$ -	\$ 144
Capacity reduction charges		1,084	-	55	25	1,164
Non-cash charges		(1,084)	-	-	(25)	(1,109)
Adjustments		-	(5)	-	-	(5)
Payments		(22)	-	-	-	(22)
Remaining accrual at June 30, 2008	\$	104	\$ 13	\$ 55	\$ _	\$ 172

Cash outlays related to the accruals for aircraft charges and facility exit costs will occur through 2017 and 2018, respectively. Cash outlays for the employee charges will be incurred over the next several quarters.

10. The Company includes changes in the fair value of certain derivative financial instruments that qualify for hedge accounting and unrealized gains and losses on available-forsale securities in comprehensive income. For the three month periods ended June 30, 2008 and 2007, comprehensive income (loss) was \$(826) million and \$317 million, respectively, and for the six month periods ended June 30, 2008 and 2007, comprehensive income (loss) was \$(1.0) billion and \$470 million, respectively. Total comprehensive income for the year ended December 31, 2007 was \$2.5 billion. The difference between net earnings (loss) and comprehensive income (loss) for the three and six month periods ended June 30, 2008 and 2007 is due primarily to the accounting for the Company's derivative financial instruments. Due to the current value of the Company's derivative contracts, some agreements with counterparties require collateral to be deposited with the Company. As of June 30, 2008 the collateral held in Shortterm investments by AMR from such counterparties was \$835 million, an increase of \$671 million from December 31, 2007, which is included in cash flows from operations. Ineffectiveness is inherent in hedging jet fuel with derivative positions based in crude oil or other crude oil related commodities. As required by Statement of Financial Accounting Standard No. 133, "Accounting for Derivative Instruments and Hedging Activities", the Company assesses, both at the inception of each hedge and on an on-going basis, whether the derivatives that are used in its hedging transactions are highly effective in offsetting changes in cash flows of the hedged items. In doing so, the Company uses a regression model to determine the correlation of the change in prices of the commodities used to hedge jet fuel (NYMEX Heating oil) to the change in the price of jet fuel. The Company also monitors the actual dollar offset of the hedges' market values as compared to hypothetical jet fuel hedges. The fuel hedge contracts are generally deemed to

In March of 2008, the FASB issued Statement of Financial Accounting Standards No. 161, "Disclosures about Derivative Instruments and Hedging Activities, an amendment of FASB Statement No. 133" (SFAS 161). SFAS 161 requires entities to provide greater transparency about how and why the entity uses derivative instruments, how the instruments and related hedged items are accounted for under SFAS 133, and how the instruments and related hedged items affect the financial position, results of operations, and cash flows of the entity. SFAS 161 is effective for fiscal years beginning after November 15, 2008. The principal impact to the Company will be to require the Company to expand its disclosure regarding its derivative instruments.

11. The following table sets forth the computations of basic and diluted earnings (loss) per share (in millions, except per share data):

	Three Months Ended June 30,			Six Months Ended June 30,				
		2008		2007	2008			2007
Numerator:								
Net earnings (loss) - numerator for basic earnings per share Interest on senior convertible notes	\$	(1,448)	\$	317 7	\$	(1,776)	\$	398 14
Net earnings (loss) adjusted for interest on senior convertible notes - numerator for diluted earnings per share	\$	(1,448)	\$	324	\$	(1,776)	\$	412
Denominator:								
Denominator for basic earnings per share – weighted-average shares Effect of dilutive securities:		251		246		250		241
Senior convertible notes		-		32		-		32
Employee options and shares		-		33		-		40
Assumed treasury shares purchased Dilutive potential common shares		-		(12) 53		-		(14) 58
Denominator for diluted earnings per share - adjusted weighted-average shares	_	251	_	299		250		299
Basic earnings (loss) per share	\$	(5.77)	\$	1.28	\$	(7.10)	\$	1.65
Diluted earnings per share	\$	(5.77)	\$	1.08	\$	(7.10)	\$	1.38

Approximately 39 million shares related to employee stock options, performance share plans, convertible notes and deferred stock were not added to the denominator for the three months ended June 30, 2008 because inclusion of such shares would have been antidilutive. For the three months ended June 30, 2008 and 2007, approximately 16 million and six million shares, respectively, related to employee stock options were not added to the denominator because the options' exercise prices were greater than the average market price of the common shares.

Approximately 42 million shares related to employee stock options, performance share plans, convertible notes and deferred stock were not added to the denominator for the six months ended June 30, 2008 because inclusion of such shares would have been antidilutive. For the six months ended June 30, 2008 and 2007, approximately 14 million and four million shares, respectively, related to employee stock options were not added to the denominator because the options' exercise prices were greater than the average market price of the common shares.

12. On April 16, 2008, the Company announced that it had reached a definitive agreement with Lighthouse Holdings, Inc., which is owned by investment funds affiliated with TPG Capital, L.P. and Pharos Capital Group, LLC for the sale of American Beacon Advisors, Inc. (American Beacon), its wholly-owned asset management subsidiary. AMR expects to receive total consideration of approximately \$480 million, of which a substantial portion will result in a gain. While primarily a cash transaction, AMR will retain a minority equity stake in the business. The sale is expected to close in the third quarter of 2008 subject to customary closing conditions and final approvals. American Beacon is classified as held for use as of June 30, 2008, as final approvals from the Board of Trustees of the American Beacon family of mutual funds, and consents from other American Beacon clients are pending. The Company expects to have significant continuing involvement with American Beacon post-sale and therefore does not expect to account for American Beacon as discontinued operations.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

Forward-Looking Information

Statements in this report contain various forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, which represent the Company's expectations or beliefs concerning future events. When used in this document and in documents incorporated herein by reference, the words "expects," "plans," "anticipates," "indicates," "believes," "forecast," "guidance," "outlook," "may," "will," "should," "seeks", "targets" and similar expressions are intended to identify forward-looking statements. Similarly, statements that describe the Company's objectives, plans or goals are forward-looking statements. Forward-looking statements include, without limitation, the Company's expectations concerning operations and financial conditions, including changes in capacity, revenues, and costs, future financing plans and needs, fleet plans, overall economic conditions, plans and objectives for future operations, and the impact on the Company of its results of operations in recent years and the sufficiency of its financial resources to absorb that impact. Other forward-looking statements which do not relate solely to historical facts, such as, without limitation, statements which discuss the possible future effects of known trends or uncertainties, or which indicate that the Company on the date of this report. The Company undertakes no obligation to publicly update or revise any forward-looking statement, whether as a result of new information, future events, or otherwise.

Forward-looking statements are subject to a number of factors that could cause the Company's actual results to differ materially from the Company's expectations. The following factors, in addition to other possible factors not listed, could cause the Company's actual results to differ materially from those expressed in forward-looking statements: the materially weakened financial condition of the Company, resulting from its significant losses in recent years; the ability of the Company to generate additional revenues and reduce its costs; changes in economic and other conditions beyond the Company's control, and the volatile results of the Company's operations; the Company's substantial indebtedness and other obligations; the ability of the Company to satisfy existing financial or other covenants in certain of its credit agreements; continued high and volatile fuel prices and further increases in the price of fuel, and the availability of fuel; the fiercely and increasingly competitive business environment faced by the Company; industry consolidation, competition with reorganized carriers; low fare levels by historical standards and the Company's reduced pricing power; the Company's busines; conflicts or everages in the Company's corporate or business strategy; government regulation of the Company's busines; conflicts or costs that are higher than those of the Company's competitors; uncertainties with respect to the Company's network; and potential reductions of available insurance coverage; the Company's ability to retain key management personnel; potential failures or disruptions of the Company's control, and the price of the Company to generate additional operations; the ability of the Company to reach acceptable agreements with third parties. Additional information concerning these and other factors is contained in the Company's Securities and Exchange Commission filings, including but not limited to the Company's 2007 Form 10-K (see in particular Item 1A "Risk Factors" in the 2007 Form 10-K).

Overview

The Company recorded a net loss of \$1.4 billion in the second quarter of 2008 compared to net earnings of \$317 million in the same period last year. The Company's second quarter 2008 results include a non-cash impairment charge of \$1.1 billion to write the McDonnell Douglas MD-80 and Embraer RJ-135 fleets and certain related long-lived assets down to their estimated fair values, and a \$55 million accrual for employee severance costs expected to be paid over the next several quarters. These charges are described in footnote 9 to the condensed consolidated financial statements.

Underlying these charges, the Company's loss reflects an historic year-over-year increase in fuel prices from an average of \$2.09 per gallon in the second quarter 2007 to an average of \$3.19 per gallon in the second quarter of 2008. Fuel expense has become the Company's largest single expense category and the price increase resulted in \$838 million in incremental year-over-year fuel expense in the second quarter 2008 (based on the year-over-year increase in the average price per gallon multiplied by gallons consumed).

The unprecedented rise in fuel price was partially offset by higher unit revenues (passenger revenue per available seat mile). Mainline passenger unit revenues increased 7.0 percent for the second quarter due to an 8.5 percent increase in passenger yield (passenger revenue per passenger mile) partially offset by a 1.1 point load factor decrease compared to the same period in 2007. Although passenger yield showed year-over-year improvement, passenger yield remains low by historical standards and below the Company's peak yield set in the year 2000.

In addition, the Company's operating and financial results were negatively affected by a substantial number of McDonnell Douglas MD-80 second quarter flight cancellations. These cancellations resulted in an approximate two percent decrease in the Company's scheduled mainline departures during the second quarter 2008.

The Company made several announcements during the second quarter. On April 16, 2008, the Company announced that it had reached a definitive agreement for the sale of American Beacon. See Note 12 to the condensed consolidated financial statements for more information. On May 21, 2008, the Company announced capacity reductions resulting in revisions to the Company's fleet plan and the impairment of the McDonnell Douglas MD80 and the Embraer RJ-135 fleets. See Note 9 to the condensed consolidated financial statements for more information.

As of June 30, 2008, the Company had commitments to acquire 33 Boeing 737-800 aircraft in 2009, seven Boeing 737-800 aircraft in 2010 and an aggregate of 20 Boeing 737 aircraft and seven Boeing 777 aircraft in 2013 through 2016 as a part of its fleet replacement strategy. On July 1, 2008, the Company committed to acquire an incremental Boeing 737-800 in 2009. See subsection entitled "Significant Indebtedness and Future Financing" under Item 2. Management's Discussion and Analysis and Note 3 to the condensed consolidated financial statements for more information.

In November 2007, the Company announced the intended divesture of AMR Eagle, its wholly-owned regional carrier. Given the current industry environment, the Company has decided to place on hold its planned divestiture until industry conditions are more stable and favorable. The Company continues to believe that a divestiture of AMR Eagle makes sense in the long term for the Company, American, AMR Eagle and their stakeholders but the Company also believes that a divestiture is not sensible amid current conditions.

The Company's ability to become consistently profitable and its ability to continue to fund its obligations on an ongoing basis will depend on a number of factors, many of which are largely beyond the Company's control. Certain risk factors that affect the Company's business and financial results are referred to under "Forward-Looking Information" above and are discussed in the Risk Factors listed in Item 1A (on pages 11-17) in the 2007 Form 10-K, and as amended in Item 1A in this Quarterly Report on Form 10-Q for the quarter ended June 30, 2008. In addition, four of the Company's largest domestic competitors and several smaller carriers have filed for bankruptcy in the last several years and have used this process to significantly reduce contractual labor and other costs. In order to remain competitive and to improve its financial condition, the Company must continue to take steps to generate additional revenues and to reduce its costs. Although the Company has a number of initiatives underway to address its cost and revenue challenges, the ultimate success of these initiatives is not known at this time and cannot be assured. If the overall industry revenue environment does not improve substantially and fuel prices remain at historically high levels for an extended period, it will be very difficult for the Company to fund its obligations on an ongoing basis and to become consistently profitable.

LIQUIDITY AND CAPITAL RESOURCES

Significant Indebtedness and Future Financing

The Company remains heavily indebted and has significant obligations (including substantial pension funding obligations), as described more fully under Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations" in the 2007 Form 10-K. As of the date of this Form 10-Q, the Company believes it should have sufficient liquidity to fund its operations for the near term, including repayment of debt and capital leases, capital expenditures and other contractual obligations. However, to maintain sufficient liquidity as the Company continues to implement its restructuring initiatives, and because the Company has significant debt, lease and other obligations in the next several years, including commitments to purchase aircraft, as well as substantial pension funding obligations, the Company will need access to significant additional funding.

The Company's possible financing sources primarily include: (i) a limited amount of additional secured aircraft debt (a very large majority of the Company's owned aircraft, including the Company's Section 1110-eligible aircraft, are encumbered) or sale-leaseback transactions involving owned aircraft; (ii) debt secured by new aircraft deliveries; (iii) debt secured by other assets; (iv) securitization of future operating receipts; (v) the sale or monetization of certain assets; (vi) unsecured debt; and (vii) issuance of equity and/or equity-like securities. Besides unencumbered aircraft, some of the Company's particular assets and other sources of liquidity that could be sold or otherwise used as sources of financing include AAdvantage program miles, route authorities and takeoff and landing slots, and certain of the Company's business units and subsidiaries, such as American Beacon (the sale of which is pending) and AMR Eagle. The availability and level of the financing sources described above cannot be assured, particularly in light of the Company's and American's recent financial results, substantial indebtedness, reduced credit ratings, extraordinarily high fuel prices, revenues that are weak by historical standards, the financial difficulties being experienced in the airline industry and recent disruptions in the credit markets. The inability of the Company to obtain necessary funding on acceptable terms would have a material adverse impact on the Company.

The Company's substantial indebtedness and other obligations have important consequences. For example, they: (i) limit the Company's ability to obtain additional financing for working capital, capital expenditures, acquisitions and general corporate purposes, and adversely affect the terms on which such financing could be obtained; (ii) require the Company to dedicate a substantial portion of its cash flow from operations to payments on its indebtedness and other obligations, thereby reducing the funds available for other purposes; (iii) make the Company more vulnerable to economic downturns; (iv) limit the Company's ability to withstand competitive pressures and reduce its flexibility in responding to changing business and economic conditions; and (v) limit the Company's flexibility in planning for, and reacting to, changes in its business and the industry in which it operates.

Future payments for all aircraft that the Company was committed to acquire as of June 30, 2008, including the estimated amounts for price escalation, are currently estimated to be approximately \$2.8 billion, with the majority currently scheduled to be made in 2011 through 2016. These payments are currently scheduled to be approximately \$278 million in the remainder of 2008, \$682 million in 2009, \$107 million in 2010, \$102 million in 2011, \$310 million in 2012, and \$1.3 billion for 2013 and beyond. These amounts are net of purchase deposits currently held by the manufacturer. However, the Company intends to accelerate the delivery of certain aircraft that it is committed to purchase and order additional aircraft, as described below.

As a part of the Company's efforts to improve the cost and fuel efficiency of its fleet, as well as lessen the Company's impact on the environment, AMR announced on April 16, 2008 its intention to further accelerate the replacement of its MD-80 aircraft fleet with Boeing 737-800 aircraft. The Company now intends to take delivery of a total of 36 Boeing 737-800 aircraft in 2009 and expects to take delivery of 34 Boeing 737-800s in 2010. These expected deliveries would be pursuant to both accelerated orders and incremental orders. If, as anticipated, the Company commits to accelerating the delivery dates of additional aircraft, the related capital expenditure commitments will be accelerated, and any incremental aircraft orders will increase the Company's obligation. This proposed fleet replacement plan would increase the Company's 2008-2010 commitments described above by an estimated \$1.1 billion. The Company's continued replacement strategy, and its execution of that strategy, will depend on such factors as future economic and industry conditions and the financial condition of the Company. The Company also continues to evaluate the economic benefits and other aspects of replacing some of the other aircraft in its fleet with new aircraft.

The Company currently has no committed financing for any aircraft that it is committed to purchase or that it may order. The Company will need additional financing to continue to execute its fleet replacement plan, and the Company expects that it will seek to finance aircraft that it acquires through financing methods it has used in the past, such as aircraft mortgages and sale leaseback transactions. However, the Company can provide no assurance that such financing will be available.

In the quarter ended June 30, 2008, American raised approximately \$220 million through a series of transactions including loans secured by aircraft and sale leasebacks of certain aircraft. In addition, on July 8, 2008, the Company raised approximately \$500 million under a loan secured by aircraft, due in installments through 2015.

As discussed in Note 6 to the condensed consolidated financial statements, the Company also has outstanding \$300 million principal amount of its 4.25 percent senior convertible notes due 2023 (the 4.25 Notes). On September 23, 2008 for the 4.25 Notes the holders may require the Company to purchase all or a portion of their notes at a price equal to 100% of their principal amount plus unpaid interest which may be paid in cash, common stock or a combination of cash and common stock. The Company is evaluating various payment and refinancing alternatives for the outstanding 4.25% notes upon the expected exercise of the put provision in September 2008. Based on current economic and market conditions, the Company currently expects to settle the 4.25 Notes in cash.

Credit Facility Covenants

American has a secured bank credit facility which consists of a \$255 million revolving credit facility, with a final maturity on June 17, 2009, and a fully drawn \$438 million term loan facility, with a final maturity on December 17, 2010 (the Revolving Facility and the Term Loan Facility, respectively, and collectively, the Credit Facility). In 2007, American paid in full the \$255 million principal balance of the Revolving Facility and as of June 30, 2008, it remained undrawn. American's obligations under the Credit Facility are guaranteed by AMR.

The Credit Facility contains a covenant (the Liquidity Covenant) requiring American to maintain, as defined, unrestricted cash, unencumbered short term investments and amounts available for drawing under committed revolving credit facilities of not less than \$1.25 billion for each quarterly period through the life of the Credit Facility. AMR and American were in compliance with the Liquidity Covenant as of June 30, 2008 and expect to be able to continue to comply with this covenant. In addition, the Credit Facility contains a covenant (the EBITDAR Covenant) requiring AMR to maintain a ratio of cash flow (defined as consolidated net income, before interest expense (less capitalized interest), income taxes, depreciation and amortization and rentals, adjusted for certain gains or losses and non-cash items) to fixed charges (comprising interest expense (less capitalized interest) and rentals). In May 2008, AMR and American entered into an amendment to the Credit Facility which waived compliance with the EBITDAR Covenant for periods ending on any date from and including June 30, 2008 through March 31, 2009, and which reduced the minimum ratios AMR is required to satisfy thereafter. The required ratio will be 0.90 to 1.00 for the one quarter period ending June 30, 2009 and will increase to 1.15 to 1.00 for the four quarter period ending September 30, 2010. Given fuel prices and revenues, it is difficult to assess whether the Company will be able to comply with these covenants, and there are no assurances that the Company will be able to do so. Failure to comply with these covenants would result in a default under a significant amount of other debt and lease obligations, and otherwise have a material adverse impact on the Company.

Credit Card Processing and Other Reserves

American has agreements with a number of credit card companies and processors to accept credit cards for the sale of air travel and other services. Under certain of American's current credit card processing agreements, the related credit card company or processor may hold back, under certain circumstances, a reserve from American's credit card receivables. American is not currently required to maintain any reserve under these agreements.

Under one such agreement, the amount of such reserve may be based on, among other things, the amount of unrestricted cash held by American and American's debt service coverage ratio, as defined in the agreement. American expects that if fuel prices remain high by historical standards and are not adequately offset by fare increases, American will be required to maintain a reserve under this agreement in future periods. Given the volatility of fuel prices and revenues, it is difficult to forecast the required amount of such reserve could range between \$200 million to \$300 million by the end of 2008, and that if current business conditions persist for some time, such required amount could increase significantly from that level in 2009.

Pension Funding Obligation

The Company's 2008 minimum required contribution to its defined benefit pension plan is \$78 million. As of June 30, 2008, the Company has contributed \$75 million to its defined benefit pension plans. On July 15, the Company made an additional contribution of \$3 million bringing year to date contributions to the minimum contribution of \$78 million. Absent a significant change in the industry environment, it is unlikely that additional contributions will be made to the Company's defined benefit pension plan in 2008.

Cash Flow Activity

At June 30, 2008, the Company had \$5.1 billion in unrestricted cash and short-term investments, compared with \$4.5 billion as of December 31, 2007, and \$255 million available under the Revolving Facility. The Company's unrestricted cash and short-term investments included \$835 million and \$164 million, respectively, in collateral deposits received from the counterparties of the Company's fuel derivative instruments. See Note 10 to the condensed consolidated financial statements for more information. Net cash provided by operating activities in the six month period ended June 30, 2008 was \$1.2 billion, a decrease of \$589 million over the same period in 2007 primarily due to a dramatic year-over-year increase in average fuel prices from \$2.09 per gallon the second quarter 2007 to \$3.19 per gallon for the same period in 2008. The fuel price increase resulted in \$838 million in incremental year-over-year expense in the second quarter 2008 (based on the year-over-year increase in the average price per gallon multiplied by gallons consumed). The Company contributed \$75 million to its defined benefit pension plans in the first six months of 2008 compared to \$180 million during the first six months of 2007.

Capital expenditures for the first six months of 2008 were \$473 million and primarily included aircraft purchase deposits and aircraft modifications.

In the past, the Company has from time to time refinanced, redeemed or repurchased its debt and taken other steps to reduce its debt or lease obligations or otherwise improve its balance sheet. Going forward, depending on market conditions, its cash positions and other considerations, the Company may continue to take such actions.

RESULTS OF OPERATIONS

For the Three Months Ended June 30, 2008 and 2007

REVENUES

The Company's revenues increased approximately \$300 million, or 5.1 percent, to \$6.2 billion in the second quarter of 2008 from the same period last year. American's passenger revenues increased by 4.6 percent, or \$210 million, despite a 2.2 percent decrease in capacity (available seat mile) (ASM). American's passenger load factor decreased 1.1 points to 82.5 percent while passenger yield increased by 8.5 percent to 13.76 cents. This resulted in an increase in passenger revenue per available seat mile (RASM) of 7.0 percent to 11.35 cents. Following is additional information regarding American's domestic and international RASM and capacity:

	Three Months Ended June 30, 2008							
	RASM	Y-O-Y		Y-O-Y				
	(cents)	Change	ASMs (billions)	Change				
DOT Domestic	11.27	5.9%	26.1	(3.6)%				
International	11.49	8.7	15.6	0.3				
DOT Latin America	11.73	10.2	7.4	2.8				
DOT Atlantic	11.29	6.0	6.5	(2.2)				
DOT Pacific	11.20	12.8	1.7	(0.8)				

The Company's Regional Affiliates include two wholly owned subsidiaries, American Eagle Airlines, Inc. and Executive Airlines, Inc. (collectively, AMR Eagle), and two independent carriers with which American has capacity purchase agreements, Trans States Airlines, Inc. (Trans States) and Chautauqua Airlines, Inc. (Chautauqua).

Regional Affiliates' passenger revenues, which are based on industry standard proration agreements for flights connecting to American flights, increased \$25 million, or 3.7 percent, to \$683 million as a result of increased passenger yield. Regional Affiliates' traffic decreased 7.5 percent to 2.4 billion revenue passenger miles (RPMs), while capacity decreased 3.1 percent to 3.3 billion ASMs, resulting in a 3.5 point decrease in the passenger load factor to 73.3 percent.

In the first quarter of 2008, the Company began classifying certain mileage sales revenue to Other revenue, which was previously recognized as a component of Passenger revenue. See Note 2 to the condensed consolidated financial statements for additional information.

OPERATING EXPENSES

The Company's total operating expenses increased 38.0 percent, or \$2.1 billion, to \$7.5 billion for the second quarter 2008 compared to the second quarter of 2007. The Company's operating expenses per ASM for the second quarter 2008 increased 41.2 percent to 16.60 cents compared to the same period in 2007. These increases are largely due to a non-cash impairment charge of \$1.1 billion to write the McDonnell Douglas MD-80 and Embraer RJ-135 fleets and certain related long-lived assets down to their estimated fair values. This impairment charge was triggered by the record increase in fuel prices over the last twelve months. In addition, the Company accrued \$55 million for severance costs related to the capacity reductions. These special charges represented 2.59 cents of the increase in operating expenses per ASM for the second quarter. The remaining increase in operating expense was primarily due to a dramatic year-over-year increase in fuel prices from \$2.09 per gallon in the second quarter of 2007 to \$3.19 per gallon for the second quarter 2008. Fuel expense is the Company's largest single expense category and the price increase resulted in \$838 million in incremental year-over-year fuel expense for the second quarter 2008 (based on the year-over-year increase in the average price per gallon multiplied by gallons consumed). Continuing high fuel prices, additional increases in the price of fuel and/or disruptions in the supply of fuel would further materially adversely affect the Company's financial condition and results of operations.

In addition, the Company's operating and financial results were negatively affected by a substantial number of McDonnell Douglas MD-80 second quarter flight cancellations. These cancellations resulted in an approximate two percent decrease in the Company's scheduled mainline departures during the second quarter 2008.

(in millions) Operating Expenses	E	Three Months Ended June 30, 2008		nge from 2007	Percentage Change
Aircraft Fuel	\$	2,423	\$	779	47.4%(a)
Wages, salaries and benefits		1,658		3	0.2
Other rentals and landing fees		318		5	1.6
Depreciation and amortization		324		29	9.8
Maintenance, materials and repairs		323		55	20.5 (b)
Commissions, booking fees and credit card expense		259		(9)	(3.4)
Aircraft rentals		125		(27)	(17.8)
Food service		133		-	-
Special charges		1,164		1,164	* (c)
Other operating expenses		742		58	8.5
Total operating expenses	\$	7,469	\$	2,057	38.0%

(a) Aircraft fuel expense increased primarily due to a 52.9 percent increase in the Company's price per gallon of fuel (net of the impact of fuel hedging) offset by a 3.6 percent decrease in the Company's fuel consumption, primarily due to reductions in available seat miles.

- (b) Maintenance, materials and repairs expense increased due to a heavier workscope of scheduled airframe maintenance overhauls, repair costs and volume, and contractual engine repair rates, which are driven by aircraft age.
- (c) Special charges are related to a non-cash impairment charge of \$1.1 billion to write down the Company's McDonnell Douglas MD-80 and Embraer RJ-135 fleets and certain related long-lived assets to their estimated fair values. This impairment charge was triggered by the record increase in fuel prices over the last twelve months. In addition, the Company accrued \$55 million for severance costs related to the capacity reductions.

OTHER INCOME (EXPENSE)

Interest income decreased \$42 million due to a decrease in interest rates. Interest expense decreased \$50 million as a result of a decrease in the Company's long-term debt balance.

INCOME TAX

The Company did not record a net tax provision (benefit) associated with its loss for the three months ended June 30, 2008 or its earnings for the three months ended June 30, 2007 due to the Company providing a valuation allowance, as discussed in Note 5 to the condensed consolidated financial statements.

OPERATING STATISTICS

The following table provides statistical information for American and Regional Affiliates for the three months ended June 30, 2008 and 2007.

	Three Months Ende	ed June 30,
	2008	2007
American Airlines, Inc. Mainline Jet Operations		
Revenue passenger miles (millions)	34,399	35,669
Available seat miles (millions)	41,718	42,647
Cargo ton miles (millions)	533	536
Passenger load factor	82.5%	83.6%
Passenger revenue yield per passenger mile (cents)	13.76	12.68
Passenger revenue per available seat mile (cents)	11.35	10.61
Cargo revenue yield per ton mile (cents)	43.74	37.25
Operating expenses per available seat mile, excluding Regional Affiliates (cents) (*)	15.80	11.14
Fuel consumption (gallons, in millions)	688	713
Fuel price per gallon (cents)	317.3	207.5
Operating aircraft at period-end	653	693
Regional Affiliates		
Revenue passenger miles (millions)	2,400	2,595
Available seat miles (millions)	3,274	3,380
Passenger load factor	73.3%	76.8%

(*) Excludes \$904 million and \$710 million of expense incurred related to Regional Affiliates in 2008 and 2007, respectively.

Operating aircraft at June 30, 2008, included:

American Airlines Aircraft		AMR Eagle Aircraft	
Airbus A300-600R	33	Bombardier CRJ-700	25
Boeing 737-800	77	Embraer 135	39
Boeing 757-200	124	Embraer 140	59
Boeing 767-200 Extended Range	15	Embraer 145	108
Boeing 767-300 Extended Range	57	Super ATR	39
Boeing 777-200 Extended Range	47	Saab 340B/340B Plus	34
McDonnell Douglas MD-80	300	Total	304
Total	653		

The average aircraft age for American's and AMR Eagle's aircraft is 15.2 years and 8.1 years, respectively.

Of the operating aircraft listed above, two owned McDonnell Douglas MD-80 aircraft, and eight operating leased Saab 340B Plus aircraft were in temporary storage as of June 30, 2008.

Owned and leased aircraft not operated by the Company at June 30, 2008, included:

American Airlines Aircraft		AMR Eagle Aircraft
Boeing 767-300 Extended Range	1	Embraer 145
Boeing 767-200 Extended Range	1	Saab 340B
Fokker 100	4	Total
McDonnell Douglas MD-80	31	
Total	37	

AMR Eagle leased its ten owned Embraer 145s that are not operated by AMR Eagle to Trans States Airlines, Inc.

For the Six Months Ended June 30, 2008 and 2007

REVENUES

The Company's revenues increased approximately \$570 million, or 5.0 percent, to \$11.9 billion for the six months ended June 30, 2008 from the same period last year. American's passenger revenues increased by 4.7 percent, or \$413 million, while capacity (ASM) decreased by 1.9 percent. American's passenger load factor remained effectively static at 80.8 percent and passenger revenue yield per passenger mile increased by 6.9 percent to 13.63 cents. This resulted in an increase in American's passenger RASM of 6.7 percent to 11.01 cents. Following is additional information regarding American's domestic and international RASM and capacity based on geographic areas defined by the DOT:

10 21 31

	Six Months Ended June 30, 2008			
	RASM	Y-O-Y		Y-O-Y
	(cents)	Change	ASMs (billions)	Change
DOT Domestic	10.91	6.4%	52.0	(3.6)%
International	11.19	7.2	30.8	1.2
DOT Latin America	11.91	9.1	15.5	3.2
DOT Atlantic	10.44	3.1	12.0	(0.5)
DOT Pacific	10.51	11.9	3.3	(1.6)

Regional Affiliates' passenger revenues, which are based on industry standard proration agreements for flights connecting to American flights, increased \$48 million, or 3.9 percent, to \$1.3 billion as a result of increased passenger yield. Regional Affiliates' traffic decreased 6.5 percent to 4.5 billion revenue passenger miles (RPMs), while capacity decreased 4.1 percent to 6.4 billion ASMs, resulting in a 1.8 point decrease in the passenger load factor to 71.2 percent.

In the first quarter of 2008, the Company began classifying certain mileage sales revenue to Other revenue, which was previously recognized as a component of Passenger revenue. See Note 2 to the condensed consolidated financial statements for additional information.

OPERATING EXPENSES

The Company's total operating expenses increased 26.1 percent, or \$2.8 billion, to \$13.4 billion for the six months ended June 30, 2008 compared to the same period of 2007. The Company's operating expenses per ASM increased 28.7 percent to 14.98 cents compared to the same period in 2007. These increases are largely due to a non-cash impairment charge of \$1.1 billion to write the McDonnell Douglas MD-80 and Embraer RJ-135 fleets and certain related long-lived assets down to their estimated fair values. This impairment charge was triggered by the record increase in fuel prices over the last twelve months. In addition, the Company accrued \$55 million for severance costs related to the capacity reductions. These special items represented 1.31 cents of the increase in operating expenses per ASM for the six months ended June 30, 2008. The remaining increase in operating expense was primarily due to a dramatic year-over-year increase in fuel prices from \$1.97 per gallon in the six months ending June 30, 2007 to \$2.97 per gallon for the same period in 2008. Fuel expense is the Company's largest single expense category and the price increase resulted in \$1.5 billion in incremental year-over-year fuel expense for the six months ended June 30, 2008 (based on the year-over-year increase in the average price per gallon multiplied by gallons consumed). Continuing high fuel prices, additional increases in the price of fuel and/or disruptions in the supply of fuel would further materially adversely affect the Company's financial condition and results of operations.

In addition, the Company's operating and financial results were negatively affected by a substantial number of McDonnell Douglas MD-80 second quarter flight cancellations. These cancellations resulted in an approximate two percent decrease in the Company's scheduled mainline departures during the second quarter 2008.

(in millions)

Operating Expenses	-	onths Ended 30, 2008	Change from 2007	Percentage Change
Aircraft Fuel	\$	4,473	1,419	46.5%(a)
Wages, salaries and benefits		3,302	(24)	(0.7)
Other rentals and landing fees		641	(1)	(0.2)
Depreciation and amortization		633	48	8.2
Maintenance, materials and repairs		638	122	23.6 (b)
Commissions, booking fees and credit card expense		516	(1)	(0.2)
Aircraft rentals		250	(53)	(17.5)
Food service		260	-	-
Special charges		1,164	1,164	* (C)
Other operating expenses		1,476	88	6.3
Total operating expenses	\$	13,353	\$ 2,762	26.1%

(a) Aircraft fuel expense increased primarily due to a 50.5 percent increase in the Company's price per gallon of fuel (net of the impact of fuel hedging) offset by a 2.7 percent decrease in the Company's fuel consumption, primarily due to reductions in available seat miles.

- (b) Maintenance, materials and repairs expense increased due to a heavier workscope of scheduled airframe maintenance overhauls, repair costs and volume, and contractual engine repair rates, which are driven by aircraft age.
- (c) Special charges are related to a non-cash impairment charge of \$1.1 billion to write the McDonnell Douglas MD-80 and Embraer RJ-135 fleets and certain related long-lived assets down to their estimated fair values. This impairment charge was triggered by the record increase in fuel prices over the last twelve months. In addition, the Company accrued \$55 million for severance costs related to the capacity reductions.

OTHER INCOME (EXPENSE)

Interest income decreased \$66 million in six months ended June 30, 2008 compared to the same period in 2007 due primarily to decrease in interest rates. Interest expense decreased \$97 million as a result of a decrease in the Company's long-term debt balance.

INCOME TAX

The Company did not record a net tax provision (benefit) associated with its loss for the six months ended June 30, 2008 or its earnings for the six months ended June 30, 2007 due to the Company providing a valuation allowance, as discussed in Note 5 to the condensed consolidated financial statements.

OPERATING STATISTICS

The following table provides statistical information for American and Regional Affiliates for the six months ended June 30, 2008 and 2007.

	Six Months Ende	Six Months Ended June 30,	
	2008	2007	
American Airlines, Inc. Mainline Jet Operations			
Revenue passenger miles (millions)	66,887	68,244	
Available seat miles (millions)	82,770	84,338	
Cargo ton miles (millions)	1,038	1,060	
Passenger load factor	80.8%	80.9%	
Passenger revenue yield per passenger mile (cents)	13.63	12.75	
Passenger revenue per available seat mile (cents)	11.01	10.32	
Cargo revenue yield per ton mile (cents)	43.17	37.80	
Operating expenses per available seat mile, excluding Regional Affiliates (cents) (*)	14.23	11.03	
Fuel consumption (gallons, in millions)	1,368	1,405	
Fuel price per gallon (cents)	295.4	196.0	
Regional Affiliates			
Revenue passenger miles (millions)	4,542	4,857	
Available seat miles (millions)	6,380	6,654	
Passenger load factor	71.2%	73.0%	

(*) Excludes \$1.6 billion and \$1.4 billion of expense incurred related to Regional Affiliates in 2008 and 2007, respectively.

<u>Outlook</u>

As a result of unprecedented fuel prices and growing concerns about the economy, the Company has reduced full year capacity from the previous guidance given in its Annual Report on Form 10-K for the year ended December 31, 2007 and its Quarterly Report on Form 10-Q for the quarter ended March 31, 2008. In May of 2008, the Company announced that it will reduce mainline domestic capacity in the fourth quarter of 2008 by 11 percent to 12 percent, compared to the fourth quarter of 2007.

Capacity for American's mainline jet operations is expected to decline 2.7 percent in the third quarter compared to the third quarter of 2007 and is expected to decline approximately 3.4 percent for the full year 2008 compared to 2007. In addition, as announced by the Company in May 2008, American's fourth quarter mainline jet capacity is expected to decline 11 to 12 percent compared to the fourth quarter of 2007.

The Company's recently announced capacity reductions aim to significantly reduce costs as well as create a more sustainable supply-and-demand balance in the market. As a result of reduced flying, the Company expects to retire 30 McDonnell Douglas MD80 and ten Airbus A300 mainline aircraft from the Company's fleet in 2008. The capacity reductions will also result in the retirement of 26 SAAB turboprop regional aircraft and the retirement or removal from service of 37 regional jets from the Company's regional affiliate fleet. These previously announced capacity changes will also result in workforce reductions at both American Airlines and AMR Eagle and could result in facility closures or facility consolidation. The remaining 24 Airbus A300 aircraft will be retired in 2009, which is expected to result in capacity reductions in 2009. Beyond the expected 2009 capacity reductions resulting from the retirement of the Airbus A300 fleet, given current fuel price and economic trends, the Company expects to make additional capacity reductions in 2009.

The Company currently expects third quarter 2008 mainline unit costs to increase approximately 25.2 percent year over year and full year 2008 mainline unit cost to increase approximately 26.5 percent year over year primarily due to the increase in the price of fuel. The Company's results are significantly affected by the price of jet fuel, which is in turn affected by a number of factors beyond the Company's control. Fuel prices are at historically high levels and are volatile. The Company expects full year fuel expense to increase significantly in 2008 versus 2007.

Critical Accounting Policies and Estimates

The preparation of the Company's financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the amounts reported in the consolidated financial statements and accompanying notes. The Company believes its estimates and assumptions are reasonable; however, actual results and the timing of the recognition of such amounts could differ from those estimates. The Company has identified the following critical accounting policies and estimates used by management in the preparation of the Company's financial statements: accounting for long-lived assets, routes, passenger revenue, frequent flyer program, stock compensation, pensions and retiree medical and other benefits, income taxes and derivatives accounting. These policies and estimates are described in the 2007 Form 10-K, except as updated below.

Fair value – The Company has adopted Statement of Financial Accounting Standard No. 157 "Fair Value Measurements" (SFAS 157) as it applies to financial assets and liabilities effective January 1, 2008. SFAS 157 defines fair value, establishes a framework for measuring fair value under generally accepted accounting principles (GAAP) and enhances disclosures about fair value measurements. Fair value is defined under SFAS 157 as the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. For additional information on the fair value of certain financial assets and liabilities, see Note 7 to the condensed consolidated financial statements for additional information.

Under SFAS 157, AMR utilizes several valuation techniques in order to assess the fair value of the Company's financial assets and liabilities. The Company's fuel derivative contracts, which primarily consist of commodity options and collars, are valued using energy and commodity market data which is derived by combining raw inputs with quantitative models and processes to generate forward curves and volatilities. The Company's short-term investments primarily utilize broker quotes in a non-active market for valuation of these securities.

Long-lived assets – The recorded value of the Company's fixed assets is impacted by a number of estimates made by the Company, including estimated useful lives, salvage values and the Company's determination as to whether aircraft are temporarily or permanently grounded. In accordance with Statement of Financial Accounting Standards No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets" (SFAS 144), the Company records impairment charges on long-lived assets used in operations when events and circumstances indicate that the assets may be impaired, the undiscounted cash flows estimated to be generated by those assets are less than the carrying amount of those assets and the net book value of the assets exceeds their estimated fair value. In making these determinations, the Company uses certain assumptions, including, but not limited to: (i) estimated fair value of the assets; and (ii) estimated future cash flows expected to be generated by the assets, generally evaluated at a fleet level, which are based on additional assumptions such as asset utilization, length of service and estimated salvage values. A change in the Company's fleet plan has been the primary indicator that has resulted in an impairment charge in the past.

The majority of American's fleet types are depreciated over 30 years. It is possible that the ultimate lives of the Company's aircraft will be significantly different than the current estimate due to unforeseen events in the future that impact the Company's fleet plan, including positive or negative developments in the areas described above. For example, operating the aircraft for a longer period will result in higher maintenance, fuel and other operating costs than if the Company replaced the aircraft. At some point in the future, higher operating costs, including higher fuel expense, and/or improvement in the Company's economic condition, could change the Company's analysis of the impact of retaining aircraft versus replacing them with new aircraft.

On April 16, 2008, the Company announced its intention to further accelerate the replacement of certain of its MD-80 aircraft fleet with Boeing 737-800 aircraft. The Company now intends to take delivery of a total of 36 Boeing 737-800 aircraft in 2009 and expects to take delivery of 34 Boeing 737-800s in 2010. See further discussion of aircraft commitments and the fleet replacement plan in subsection entitled "Significant Indebtedness and Future Financing" under Item 2. Management's Discussion and Analysis and Note 3 to the condensed consolidated financial statements.

In the second quarter of 2008, in connection with the May 21, 2008 announcement regarding capacity reductions and related matters, the Company concluded a triggering event had occurred and impairment testing was necessary. As a result of that testing, the Company recorded impairment charges related to its McDonnell Douglas MD-80 aircraft and Embraer RJ-135 aircraft. See Note 9 to the condensed consolidated financial statements for additional information with respect to these impairment charges.

Item 3. Quantitative and Qualitative Disclosures about Market Risk

There have been no material changes in market risk from the information provided in <u>Item 7A. Quantitative and Qualitative Disclosures About Market Risk</u> of the Company's 2007 Form 10-K. The change in market risk for aircraft fuel is discussed below for informational purposes.

The risk inherent in the Company's fuel related market risk sensitive instruments and positions is the potential loss arising from adverse changes in the price of fuel. The sensitivity analyses presented do not consider the effects that such adverse changes may have on overall economic activity, nor do they consider additional actions management may take to mitigate the Company's exposure to such changes. Therefore, actual results may differ. The Company does not hold or issue derivative financial instruments for trading purposes.

Aircraft Fuel The Company's earnings are affected by changes in the price and availability of aircraft fuel. In order to provide a measure of control over price and supply, the Company trades and ships fuel and maintains fuel storage facilities to support its flight operations. The Company also manages the price risk of fuel costs primarily by using jet fuel and heating oil hedging contracts. Market risk is estimated as a hypothetical 10 percent increase in the June 30, 2008 cost per gallon of fuel. Based on projected 2008 and 2009 fuel usage through June 30, 2009, such an increase would result in an increase to aircraft fuel expense of approximately \$909 million in the twelve months ended June 30, 2009, inclusive of the impact of effective fuel hedge instruments outstanding at June 30, 2008, and assumes the Company's fuel hedging program remains effective under Financial Accounting Standard No. 133, "Accounting for Derivative Instruments and Hedging Activities". Comparatively, based on projected 2008 fuel usage, such an increase would have resulted in an increase to aircraft fuel expense of approximately \$649 million in the twelve months ended December 31, 2008, inclusive of the impact of fuel hedge instruments outstanding at December 31, 2007. The change in market risk is primarily due to the increase in fuel prices.

Ineffectiveness is inherent in hedging jet fuel with derivative positions based in crude oil or other crude oil related commodities. As required by Statement of Financial Accounting Standard No. 133, "Accounting for Derivative Instruments and Hedging Activities", the Company assesses, both at the inception of each hedge and on an on-going basis, whether the derivatives that are used in its hedging transactions are highly effective in offsetting changes in cash flows of the hedged items. In doing so, the Company uses a regression model to determine the correlation of the change in prices of the commodities used to hedge jet fuel (e.g. NYMEX Heating oil) to the change in the price of jet fuel. The Company also monitors the actual dollar offset of the hedges' market values as compared to hypothetical jet fuel hedges. The fuel hedge contracts are generally deemed to be "highly effective" if the R-squared is greater than 80 percent and the dollar offset correlation is within 80 percent to 125 percent. The Company discontinues hedge accounting prospectively if it determines that a derivative is no longer expected to be highly effective as a hedge or if it decides to discontinue the hedging relationship.

As of June 30, 2008, the Company had effective hedges, including option contracts and collars, covering approximately 32 percent of its estimated remaining 2008 fuel requirements. The consumption hedged for the remainder of 2008 is capped at an average price of approximately \$2.90 per gallon of jet fuel excluding taxes and transportation costs. A deterioration of the Company's financial position could negatively affect the Company's ability to hedge fuel in the future.

Item 4. Controls and Procedures

The term "disclosure controls and procedures" is defined in Rules 13a-15(e) and 15d-15(e) of the Securities Exchange Act of 1934, or the Exchange Act. This term refers to the controls and procedures of a company that are designed to ensure that information required to be disclosed by a company in the reports that it files under the Exchange Act is recorded, processed, summarized and reported within the time periods specified by the Securities and Exchange Commission. An evaluation was performed under the supervision and with the participation of the Company's management, including the Chief Executive Officer (CEO) and Chief Financial Officer (CFO), of the effectiveness of the Company's disclosure controls and procedures as of June 30, 2008. Based on that evaluation, the Company's management, including the CEO and CFO, concluded that the Company's disclosure controls and procedures were effective as of June 30, 2008. During the quarter ending on June 30, 2008, there was no change in the Company's internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II: OTHER INFORMATION

Item 1. Legal Proceedings

Between April 3, 2003 and June 5, 2003, three lawsuits were filed by travel agents, some of whom opted out of a prior class action (now dismissed) to pursue their claims individually against American, other airline defendants, and in one case against certain airline defendants and Orbitz LLC. The cases, <u>Tam Travel et. al., v. Delta Air Lines et. al.</u>, in the United States District Court for the Northern District of California, San Francisco (51 individual agencies), <u>Paula Fausky d/b/a Timeless Travel v. American Airlines, et. al.</u>, in the United States District Court for the Northern District of Ohio, Eastern Division (29 agencies) and <u>Swope Travel et al. v. Orbitz et. al.</u> in the United States District Court for the Northern District of Ohio, Eastern Division. Collectively, these lawsuits seek damages and injunctive relief alleging that the certain airline defendants and Orbitz LLC: (i) conspired to prevent travel agents from acting as effective competitors in the distribution of airline tickets to passengers in violation of Section 1 of the Sherman Act; (ii) conspired to monopolize the distribution of common carrier air travel between airports in the United States in violation of Section 1 of the Sherman Act; On September 23, 2005, the <u>Fausky</u> plaintiffs dismissed their claims with prejudice. On September 14, 2006, the court dismissed with prejudice 28 of the <u>Swope</u> plaintiffs. On October 29, 2007, the court dismissed all of the <u>Swope</u> plaintiffs' claims. The <u>Swope</u> plaintiffs have appealed the court's decision. American continues to vigorously defend these lawsuits. A final adverse court decision awarding substantial money damages or placing material restrictions on the Company's distribution practices would have a material adverse impact on the Company.

On July 12, 2004, a consolidated class action complaint that was subsequently amended on November 30, 2004, was filed against American and the Association of Professional Flight Attendants (APFA), the union which represents American's flight attendants (Ann M. Marcoux, et al., v. American Airlines Inc., et al. in the United States District Court for the Eastern District of New York). While a class has not yet been certified, the lawsuit seeks on behalf of all of American's flight attendants or various subclasses to set aside, and to obtain damages allegedly resulting from, the April 2003 Collective Bargaining Agreement referred to as the Restructuring Participation Agreement (RPA). The RPA was one of three labor agreements American successfully reached with its unions in order to avoid filing for bankruptcy in 2003. In a related case (Sherry Cooper, et al. v. TWA Airlines, LLC, et al., also in the United States District Court for the Eastern District of New York), the court denied a preliminary injunction against implementation of the RPA on June 30, 2003. The Marcoux suit alleges various claims against the APFA and American relating to the RPA and the ratification vote on the RPA by individual APFA members, including: violation of the Labor Management Reporting and Disclosure Act (RLA) through improper coercion of flight attendants into voting or changing their vote for ratification, and violations of the Racketeer Influenced and Corrupt Organizations Act of 1970 (RICO). On March 28, 2006, the district court dismissed all of various state law claims against the APFA, and the claimed violations of RICO. This leaves the claimed violations of the RIA and the duty of fair representation against American and the APFA (as well as one LMRDA claim and one claim against the APFA of a breach of its constitution). Although the Company believes the case against it is without merit and both American and the APFA are vigorously defending the lawsuit, a final adverse court decision invalidating the RPA and awarding substanti

On February 14, 2006, the Antitrust Division of the United States Department of Justice (the "DOJ") served the Company with a grand jury subpoena as part of an ongoing investigation into possible criminal violations of the antitrust laws by certain domestic and foreign air cargo carriers. At this time, the Company does not believe it is a target of the DOJ investigation. The New Zealand Commerce Commission notified the Company on February 17, 2006 that it is also investigating whether the Company and certain other cargo carriers entered into agreements relating to fuel surcharges, security surcharges, war risk surcharges, and customs clearance surcharges. On February 22, 2006, the Company received a letter from the Swiss Competition Commission informing the Company that it too is investigating whether the Company and certain other cargo carriers entered into agreements relating to fuel surcharges, security surcharges, war risk surcharges, and customs clearance surcharges. On March 11, 2008, the Company received from the Swiss Competition Commission a request for information concerning, among other things, the scope and organization of the Company's activities in Switzerland. On December 19, 2006 and June 12, 2007, the Company received requests for information from the European Commission, seeking information regarding the Company's corporate structure, revenue and pricing announcements for air cargo shipments to and from the European Union. On January 23, 2007, the Brazilian competition authorities, as part of an ongoing investigation, conducted an unannounced search of the Company's cargo facilities in Sao Paulo, Brazil. On April 28, 2008, the Brazilian competition authorities preliminarily charged the Company with violating Brazilian competition laws. The authorities are investigating whether the Company and certain other foreign and domestic air carriers violated Brazilian competition laws by illegally conspiring to set fuel surcharges on cargo shipments. The Company is vigorously contesting the allegations and the preliminary findings of the Brazilian competition authorities. On June 27, 2007 and October 31, 2007, the Company received requests for information from the Australian Competition and Consumer Commission seeking information regarding fuel surcharges imposed by the Company on cargo shipments to and from Australia and regarding the structure of the Company's cargo operations. On December 18, 2007, the European Commission issued a Statement of Objection ("SO") against 26 airlines, including the Company. The SO alleges that these carriers participated in a conspiracy to set surcharges on cargo shipments in violation of EU law. The SO states that, in the event that the allegations in the SO are affirmed, the Commission will impose fines against the Company. The Company intends to vigorously contest the allegations and findings in the SO under EU laws, and it intends to cooperate fully with all other pending investigations. In the event that the SO is affirmed or other investigations uncover violations of the U.S. antitrust laws or the competition laws of some other jurisdiction, or if the Company were named and found liable in any litigation based on these allegations, such findings and related legal proceedings could have a material adverse impact on the Company.

Approximately 44 purported class action lawsuits have been filed in the U.S. against the Company and certain foreign and domestic air carriers alleging that the defendants violated U.S. antitrust laws by illegally conspiring to set prices and surcharges on cargo shipments. These cases, along with other purported class action lawsuits in which the Company was not named, were consolidated in the United States District Court for the Eastern District of New York as In re Air Cargo Shipping Services Antitrust Litigation, 06-MD-1775 on June 20, 2006. Plaintiffs are seeking trebled money damages and injunctive relief. The Company has not been named as a defendant in the consolidated complaint filed by the plaintiffs. However, the plaintiffs have not released any claims that they may have against the Company, and the Company may later be added as a defendant in the litigation. If the Company is sued on these claims, it will vigorously defend the suit, but any adverse judgment could have a material adverse impact on the Company. Also, on January 23, 2007, the Company was served with a purported class action Holdings, et al.). The plaintiff alges that the defendants violated Canadian competition laws by illegally conspiring to set prices and surcharges on cargo shipments. The plaintiff alges that the defendants violated Canadian competition laws by illegally conspiring to set prices and surcharges on cargo shipments. The company could be brought back into the litigation at a future date. If litigation is recommenced against the Company in the Canadian courts, the Company will vigorously defend itself; however, any adverse judgment could have a material adverse impact on the Company.

On June 20, 2006, the DOJ served the Company with a grand jury subpoena as part of an ongoing investigation into possible criminal violations of the antitrust laws by certain domestic and foreign passenger carriers. At this time, the Company does not believe it is a target of the DOJ investigation. The Company intends to cooperate fully with this investigation. On September 4, 2007, the Attorney General of the State of Florida served the Company with a Civil Investigative Demand as part of its investigation of possible violations of federal and Florida antitrust laws regarding the pricing of air passenger transportation. In the event that this or other investigations uncover violations of the U.S. antitrust laws or the competition laws of some other jurisdiction, such findings and related legal proceedings could have a material adverse impact on the Company.

Approximately 52 purported class action lawsuits have been filed in the U.S. against the Company and certain foreign and domestic air carriers alleging that the defendants violated U.S. antitrust laws by illegally conspiring to set prices and surcharges for passenger transportation. On October 25, 2006, these cases, along with other purported class action lawsuits in which the Company was not named, were consolidated in the United States District Court for the Northern District of California as <u>In re International Air</u> <u>Transportation Surcharge Antitrust Litigation</u>, Civ. No. 06-1793 (the "Passenger MDL"). On July 9, 2007, the Company was named as a defendant in the Passenger MDL. On March 13, 2008, and March 14, 2008, two additional purported class action complaints, <u>Turner v. American Airlines, et al.</u>, Civ. No. 08-1079 (E.D.N.Y.), were filed against the Company, alleging that the Company violated U.S. antitrust laws by illegally conspiring to set prices and surcharges for passenger transportation in Japan and Germany, respectively. Plaintiffs in the <u>Turner</u> and <u>LaFlamme</u> cases are seeking trebled money damages and injunctive relief. The Company vigorously will defend these lawsuits, but any adverse judgment in these actions could have a material adverse impact on the Company.

American is defending a lawsuit (Love Terminal Partners, L.P. et al. v. The City of Dallas, Texas et al.) filed on July 17, 2006 in the United States District Court in Dallas. The suit was brought by two lessees of facilities at Dallas Love Field Airport against American, the cities of Fort Worth and Dallas, Southwest Airlines, Inc., and the Dallas/Fort Worth International Airport Board. The suit alleges that an agreement by and between the five defendants with respect to Dallas Love Field violates Sections 1 and 2 of the Sherman Act. Plaintiffs seek injunctive relief and compensatory and statutory damages. On October 31, 2007, the court entered an order dismissing all of the plaintiffs' claims. The plaintiffs have appealed. American will vigorously defend this lawsuit; however, any adverse judgment could have a material adverse impact on the Company.

On August 21, 2006, a patent infringement lawsuit was filed against American and American Beacon Advisors, Inc. (a wholly-owned subsidiary of the Company), in the United States District Court for the Eastern District of Texas (Ronald A. Katz Technology Licensing, L.P. v. American Airlines, Inc., et al.). This case has been consolidated in the Central District of California for pre-trial purposes with numerous other cases brought by the plaintiff against other defendants. The plaintiff alleges that American and American Beacon infringe a number of the plaintiff's patents, each of which relates to automated telephone call processing systems. The plaintiff is seeking past and future royalties, injunctive relief, costs and attorneys' fees. Although the Company believes that the plaintiff's claims are without merit and is vigorously defending the lawsuit, a final adverse court decision awarding substantial money damages or placing material restrictions on existing automated telephone call system operations would have a material adverse impact on the Company.

Item 4. Submission of Matters to a Vote of Security Holders

The owners of 224,332,611 shares of common stock, or 89.93 percent of shares outstanding, were represented at the annual meeting of stockholders on May 21, 2008 at the American Airlines Training & Conference Center, Flagship Auditorium, 4501 Highway 360 South, Fort Worth, Texas.

Stockholders elected the Company's 13 nominees to the 13 director positions by the vote shown below:

Nominees	Votes For	Votes Withheld
Gerard J. Arpey	189,200,053	35,132,558
John W. Bachmann	191,394,830	32,937,781
David L. Boren	181,816,208	42,516,403
Armando M. Codina	181,494,763	42,837,848
Rajat K. Gupta	191,394,263	32,938,348
Alberto Ibargüen	191,439,850	32,892,761
Ann M. Korologos	181,562,784	42,769,827
Michael A. Miles	183,822,422	40,510,189
Philip J. Purcell	187,491,536	36,841,075
Ray M. Robinson	191,148,164	33,184,447
Judith Rodin	189,098,046	35,234,565
Matthew K. Rose	187,387,888	36,944,723
Roger T. Staubach	191,462,397	32,870,214

Stockholders ratified the Audit Committee's decision to retain Ernst & Young LLP as independent auditors for the Company for the 2008 fiscal year. The vote was 214,881,637 in favor, 3,377,883 against, 6,072,911 abstaining and 180 not voting.

Stockholders rejected a proposal to allow cumulative voting in election of outside directors. The proposal was submitted by Mrs. Evelyn Y. Davis. The vote was 52,617,777 in favor, 118,234,294 against, 5,995,165 abstaining and 47,485,375 not voting.

Stockholders rejected a proposal to give holders of any of the Company's outstanding common stock the power to call a special shareholder meeting. The proposal was submitted by Mr. John Chevedden. The vote was 33,771,731 in favor, 136,797,211 against, 6,278,294 abstaining and 47,485,375 not voting.

Stockholders rejected a proposal to require a separation of the roles of CEO and Board Chairman. The proposal was submitted by Ms. Patricia Kennedy, with Mr. Chevedden acting as her proxy. The vote was 59,496,817 in favor, 110,351,618 against, 6,998,801 abstaining and 47,485,375 not voting.

Stockholders rejected a proposal to allow shareholders to vote on a non-binding advisory resolution to ratify the compensation of the Company's named executive officers. The proposal was submitted by The Firefighters' Pension System of the City of Kansas City, Missouri, Trust. The vote was 59,526,061 in favor, 80,412,466 against, 36,908,709 abstaining and 47,485,375 not voting.

Item 5. Other Information

On July 15, 2008, the Compensation Committee of the Company's Board of Directors approved an agreement extending the term of the Employment Agreement dated March 29, 2006, between the Company, American and Thomas W. Horton, the Company's and American's Executive Vice President – Finance and Planning and Chief Financial Officer, through March 29, 2012, on substantially the same terms. The Employment Agreement was filed as Exhibit 10.1 to AMR's current report on Form 8-K dated March 31, 2006, and the agreement extending the term of the Employment Agreement is attached to this Form 10-Q as Exhibit 10.5.

Item 6. Exhibits

The following exhibits are included herein:

- 10.1 Amended and Restated Career Performance Shares Deferred Stock Award Agreement dated as of July 25, 2005 between the Company and Gerard J. Arpey
- 10.2 Purchase Agreement No. 1977 Supplement No. 27 dated May 14, 2008
- 10.3 Purchase Agreement No. 1977 Supplement No. 28 dated June 30, 2008
- 10.4 Stock Purchase Agreement by and among American Beacon Advisors, Inc., AMR Corporation and Lighthouse Holdings, Inc., dated as of April 16, 2008
- 10.5 Letter agreement dated July 20, 2008 between the Company, American and Thomas W. Horton
- 12 Computation of ratio of earnings to fixed charges for the three and six months ended June 30, 2008 and 2007.
- 31.1 Certification of Chief Executive Officer pursuant to Rule 13a-14(a).
- 31.2 Certification of Chief Financial Officer pursuant to Rule 13a-14(a).
- 32 Certification pursuant to Rule 13a-14(b) and section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code).

Signature

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

AMR CORPORATION

Date: July 17, 2008 BY: Thomas W. Horton Executive Vice President and Chief Financial Officer

/s/ Thomas W. Horton

(Principal Financial and Accounting Officer)

AMENDED AND RESTATED CAREER PERFORMANCE SHARES DEFERRED STOCK AWARD AGREEMENT

This Amended and Restated Career Performance Shares Deferred Stock Award Agreement (this "Agreement") is entered into on May 20, 2008 to be effective as of July 25, 2005 and amends and restates in its entirety the Career Performance Shares Deferred Stock Award Agreement dated as of July 25, 2005 (the "Grant Date"), by and between AMR Corporation, a Delaware corporation (the "Corporation") and Gerard J. Arpey ("Arpey").

WHEREAS, the Committee has determined to amend and restate the Agreement in order to ensure compliance with Section 409A of the Internal Revenue Code of 1986, as amended, and the regulations and other guidance issued thereunder, and as otherwise provided herein; and

WHEREAS, the Board of Directors of the Corporation (the "Board") and the Board's Compensation Committee has determined that it is in the best interests of the Corporation and its stockholders to align Arpey's long term interests with those of the Corporation's stockholders and to provide incentives for Arpey to remain with the Corporation as its Chairman, President and/or Chief Executive Officer (collectively, the "CEO"); and

WHEREAS, the Committee has determined to make initial grants to Arpey of deferred stock, as the first steps to induce Arpey to remain as the CEO and to motivate him during his tenure as the CEO.

NOW, THEREFORE, the Corporation and Arpey hereby agree as follows:

1. <u>Grant of Award</u> (a) As of the Grant Date, Arpey is granted 58,000 deferred shares of the Corporation's Common Stock, \$1.00 par value (such shares to be referenced as "Deferred Shares" and the grant to be referenced as the "2005 Award") pursuant to the terms of the 1998 Long Term Incentive Plan, as amended (the "1998 Plan"). The 2005 Award and the Subsequent Awards (as later defined in this Agreement) will be collectively referenced as the "Awards" and may be individually referenced as an "Award".

(b) Any Award will vest in accordance with Sections 2 and 4 of this Agreement. Unless otherwise determined by the Committee, distributions in respect of any vested portion of any Award shall be made in shares of Common Stock authorized for issuance under the 1998 Plan or any subsequently adopted equity compensation plan (within the meaning of the rules of the New York Stock Exchange) or pursuant to arrangement that is exempt from the rules applicable to such equity compensation plans.

2. <u>Performance Period/Vesting</u> The Awards will vest, if at all, on July 25, 2015 (the "Vesting Date")(subject to earlier vesting as detailed in Sections 3 and 4 of this Agreement). Prior to any vesting of the Awards pursuant to this Section 2, but as soon as feasible after the Vesting Date, the Committee will determine that the performance criteria (the "Criteria") established for the Awards have been satisfied, in whole or in part. Based upon the foregoing determination, the number of Deferred Shares for each Award will vest on a percentage basis from 0% to 175%. The Criteria to be used by the Committee in determining the vesting of each Award are set forth in Appendix A to this Agreement. Provided Arpey has paid all applicable taxes with respect to each Award, the shares of Common Stock that vest pursuant to this Section 2 will be issued and delivered to Arpey as soon as feasible following the determination of the Committee as to satisfaction of the Criteria. Upon delivery of the Common Stock to Arpey, this Agreement will terminate. In no case shall payment be made later than 2015.

3. <u>Early Termination</u> (a) For purposes of this Agreement, an Early Termination is the occurrence of one of the following events prior to the Vesting Date:

(i) Arpey ceases to be the Corporation's CEO due to an approved Early Retirement (which is defined as retirement from employment with the Corporation, or a Subsidiary or Affiliate thereof, at or after age 55 but before the age of 60 and with the express approval of the then existing Board);

(ii) Arpey ceases to be the Corporation's CEO due to his death or Disability (as Disability is defined in section 401A(a)(2)(C) of the Internal Revenue Code of 1986, as amended, (the "IRC");

- (iii) The Board replaces Arpey as the Corporation's CEO for reasons other than for Cause;
- (iv) Arpey resigns as CEO for Good Reason (as such term is defined in this Section 3); or
 - (v) A Change in Control (as such term is defined in Section 10 of this Agreement) of the Corporation.

(b) As used in this Agreement, "Good Reason" means one of the following has occurred without Arpey's consent prior to the Vesting Date: (i) his base salary in effect as of the Grant Date is reduced (provided, a reduction in Arpey's base salary that is part of a salary reduction program that affects the other senior officers of the Corporation, will not qualify as Good Reason); (ii) Arpey suffers a significant reduction in the authority, duties and responsibilities as CEO and he concludes in good faith that he can no longer perform the duties of CEO as was contemplated on the Grant Date; and (iii) the material benefits provided Arpey as of the Grant Date are materially reduced. Upon an event of Good Reason occurring, Arpey will provide the Board with written notice of such occurrence. If the Board has not taken action to cure such an event of Good Reason within 30 days following its receipt of Arpey's written notice, then Arpey's subsequent resignation (provided it occurs with 60 days of his written notice to the Board), will be deemed conclusively to be for Good Reason. Any notice to the Board as contemplated by this paragraph, will be sent to the Board via the Corporation's Corporate Secretary.

(c) Upon the occurrence of an Early Termination, the Early Termination Date will be deemed to be, as appropriate: the date of Early Retirement; the date of death; the date of Disability; the date Arpey is replaced as CEO; the date of his resignation for Good Reason; or the date of the Change in Control of the Corporation. Notwithstanding the foregoing, the determination by the Board of the Early Termination Date will in all cases be determinative.

4. <u>Vesting for Early Termination</u> (a) Upon the occurrence of an Early Termination, an Award that has been granted to Arpey prior to the Early Termination Date will be deemed to have vested as of such Early Termination Date. Thereafter, the Committee will review the Criteria to determine whether and to what extent the Criteria have been satisfied as of the Early Termination Date. Based upon the foregoing determination, the Committee may, in its sole discretion, adjust the number of Deferred Shares vesting for each such Award by a percentage factor between 0% and 175% (the vested portion of each such Award as so determined by the Committee will, in the aggregate, be referenced as the "Vested Award").

(b) In the event of an Early Termination on account of Early Retirement (Section 3(a)(i)), replacement without Cause (Section 3(a)(iii)) or termination for Good Reason (Section 3(a)(iv)), and provided that Arpey has paid all applicable taxes with respect to the Vested Award, shares of the Corporation's Common Stock, \$1.00 par value, in an amount equal to the Vested Award, will be delivered to Arpey within 30 days after the sixth month anniversary of the date of Arpey's separation from employment as a result of such Early Termination. "Separation from employment" for such purpose shall mean a "separation from service" under Treasury Regulation 1.409A-1(h) or successor guidance thereto. Upon delivery of the Common Stock to Arpey, this Agreement will terminate.

(c) In the event of an Early Termination on account of death or Disability (Section 3(a)(ii)) or Change in Control of the Corporation (Section 3(a)(v)), and provided that Arpey has paid all applicable taxes with respect to the Vested Award, shares of the Corporation's Common Stock, \$1.00 par value, in an amount equal to the Vested Award, will be delivered to Arpey within 30 days of such Early Termination Date. Upon delivery of the Common Stock to Arpey, this Agreement will terminate.

5. <u>Subsequent Awards</u> Provided Arpey remains an employee of the Corporation, he will receive a minimum of 58,000 Deferred Shares in each of the succeeding four years after 2005 (collectively, the "Subsequent Awards" and individually a "Subsequent Award"). Each Subsequent Award shall be made in accordance with the terms of the 1998 Plan or under such other plan, program or arrangement under which awards of this nature are authorized for issuance (the "Applicable Equity Plan"). The grant date for each Subsequent Award will be no later than December 31 of such succeeding year. In the event the Board has not granted any such Subsequent Award in a succeeding year, the grant date for the Subsequent Award in that year will be deemed to be the last business day of July. Vesting of a Subsequent Award will be in accordance with Sections 2, 3 and 4 of this Agreement and the number of Deferred Shares vesting for each Subsequent Award may range from 0% to 175%.

6. <u>Termination for Cause; Other</u> If prior to the Vesting Date and provided there has been no event of Early Termination, then in the event (a) the Board decides to replace Arpey as the Corporation's CEO for reasons of Cause or (b) Arpey resigns as CEO for reasons other than Good Reason, each Award made prior to such replacement or resignation will be forfeited in its entirety and this Agreement will terminate immediately.

7. <u>Transfer Restrictions</u> This Award is non-transferable otherwise than by will or by the laws of descent and distribution, and may not otherwise be assigned, pledged or hypothecated and will not be subject to execution, attachment or similar process. Upon any attempt by Arpey (or his successor in interest after his death) to effect any such disposition, or upon the commencement of any such process, the Award will immediately become null and void, at the discretion of the Committee.

8. <u>Miscellaneous</u> This Agreement (a) will be binding upon and inure to the benefit of any successor of the Corporation, (b) will be governed by the laws of the State of Texas and any applicable laws of the United States, and (c) may not be amended without the written consent of both the Corporation and Arpey. No contract or right of employment will be implied by this Agreement. If Arpey does not forward to the Corporation, within the applicable period, required taxes with respect to any shares of Common Stock which have vested pursuant to this Agreement, the Corporation may withhold from any payments to be made to him by the Corporation (or any Subsidiary or Affiliate thereof), an amount(s) equal to such taxes or it may withhold the delivery of any shares of the Common Stock, \$1.00 par value, as contemplated by Sections 2 or 4, until such time as such required taxes have been paid.

9. <u>Securities Law Requirements</u> (a) The Corporation will not be required to issue shares pursuant to this Award unless and until (i) such shares have been duly listed upon each stock exchange on which the Corporation's Stock is then registered; and (ii) a registration statement under the Securities Act of 1933 with respect to such shares is then effective.

(b) The Board may require Arpey to furnish to the Corporation, prior to the issuance of any shares of Common Stock, \$1.00 par value, in connection with this Agreement, an agreement, in such form as the Board may from time to time deem appropriate, in which he represents that the shares acquired by him are being acquired for investment and not with a view to the sale or distribution thereof.

10. Incorporation of Applicable Equity Plan Provisions Each Award pursuant to this Agreement shall be made pursuant to the Applicable Equity Plan and is subject to all of the terms and provisions of the Applicable Equity Plan as if the same were fully set forth herein. Capitalized terms not otherwise defined herein will have the meanings set forth for such terms in the 1998 Plan. For purposes of this Agreement, the term "Change in Control" will mean a "change in ownership" or "change in effective control" or "change in ownership of the assets" of the Corporation, as determined pursuant to Treasury Regulation 1.409A-3(i)(5) (or successor guidance thereto) and the 1998 Plan and (b) "Cause" will have the meaning set forth in the 1998 Plan. Notwithstanding the provisions of the 1998 Plan or any other Applicable Equity Plan, (y) Arpey cannot defer payment of an Award and (z) the payment of an Award cannot be accelerated by the Committee or the Corporation, except as provided in this Agreement.

GERARD J. ARPEY

AMR CORPORATION

Corporate Secretary

Kenneth W. Wimberly

Appendix A to that AMENDED AND RESTATED CAREER PERFORMANCE SHARE PROGRAM DEFERRED STOCK AWARD AGREEMENT dated as of July 25, 2005, between AMR Corporation and Gerard J. Arpey (the "Agreement")

The Agreement, Sections 2 and 4, contemplates the existence of performance criteria that will be considered by the Committee when determining the vesting of Award.

In making its vesting determination the Committee will consider the following performance criteria:

- 1. The Corporation's overall cash flow;
- 2. The Corporation's earnings (operating, net or otherwise);
- 3. The per share price of the Common Stock;
- 4. The operating performance of the Corporation and its Subsidiaries (including safety and other issues concerning regulatory compliance);
- 5. The rate of return on investment and/or equity;
- 6. Measures of employee engagement and/ or satisfaction;
- 7. The overall state of relations between the Corporation and the representatives of organized labor groups;
- 8. The Corporation's balance sheet;
- 9. The overall state of relations between the Corporation and its largest shareholders;
- 10. The Corporation's revenues; and
- 11. Such other factors as the Committee may in its discretion deem material.

In making its vesting determination, the Committee may, in its discretion, consider the foregoing factors (a) on a relative basis vis-à-vis the Corporation's major competitors or (b) on a stand-alone basis. Furthermore, the Committee may, in its discretion, consider each criterion equally or may assign greater significance to certain criterion.

PURCHASE AGREEMENT No. 1977 SUPPLEMENT No. 27

PURCHASE AGREEMENT No. 1977 SUPPLEMENT No. 27 (this "Agreement") dated May _____, 2008 between The Boeing Company ("Boeing") and American Airlines, Inc. ("Customer").

$\underline{R} \underline{E} \underline{C} \underline{I} \underline{T} \underline{A} \underline{L} \underline{S}$:

- A. Boeing and Customer have heretofore entered into that certain Purchase Agreement No. 1977, dated October 31, 1997, as amended and supplemented, (capitalized terms used herein without definition shall have the meanings specified therefor in such Purchase Agreement).
- B. Pursuant to Letter Agreement No. 6-1162-AKP-075 titled Aircraft Purchase Rights and Substitution Rights (the "Rights Letter"), Boeing and Customer have agreed to, among other things, treatment of aircraft Purchase Rights.
- C. Customer and Boeing desire to amend and supplement the Purchase Agreement as provided below.

In consideration of the foregoing premises and other good and sufficient consideration, Boeing and Customer hereby agree as follows:

1. Amendment to Reflect Customer's Exercise of MADP Rights.

- 1.1. The Purchase Agreement is amended and supplemented to reflect the exercise of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Rights Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] by Customer with the Scheduled Delivery [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] ("Exercised MADP Aircraft").
- 1.2. The Scheduled Delivery Month and Advanced Payment Schedule for the aircraft is set forth in Table 1A (R6), attached hereto.

1.3. In recognition of the fact the configuration of Customer's Model 737-823 Aircraft is in the process of being updated, the previously selected Optional Features that were selected during initial configuration for the Deferred Aircraft will be incorporated in Table 1A(R6) as a placeholder until Customer selects Optional Features or other desired amendments to the Detail Specification ("Configuration Changes"). The effects of all Configuration Changes, which are mutually agreed upon between Boeing and Customer for incorporation into the Detail Specification will be incorporated into Exhibit A by written amendment no later than [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

- 2. Table of Contents. The Table of Contents dated April 2008 is replaced in its entirety with the revised Table of Contents dated May 2008 (Attachment A hereto) to reflect amendments made to the Purchase Agreement by this Agreement.
- 3. MADP Rights. Pursuant to SA 26, Attachment B (R7) to the Rights Letter is hereby replaced in its entirety with the revised Attachment B (R8) attached hereto and hereby incorporated into the Purchase Agreement. The revised number of certain Customer MADP rights pursuant to this Agreement are reflected in the attached Attachment B (R8) hereto.
- Supplement Exhibit BFE1. Supplement Exhibit BFE1 (R7) is hereby replaced in its entirety with the revised Supplement Exhibit BFE1 (R8) attached hereto and hereby 4. incorporated into the Purchase Agreement. The updated on-dock dates for all contracted firm Aircraft thru December 2013 are reflected in the attached Supplement Exhibit BFE1 (R8).
- Advance Payments for Aircraft. Due at signing of this Agreement, Customer owes Boeing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- Effect on Purchase Agreement. Except as expressly set forth herein, all terms and provisions contained in the Purchase Agreement shall remain in full force and effect. This 6. Agreement contains the entire agreement between the parties with respect to the subject matter hereof and supersedes all previous proposals, and agreements, understandings, commitments or representations whatsoever, oral or written, with respect to the subject matter hereof and may be changed only in writing signed by authorized representatives of the parties.

IN WITNESS WHEREOF, Boeing and Customer have each caused this Agreement to be duly executed as of the day and year first above written.

AMERICAN AIRLINES, INC.

THE BOEING COMPANY By Bv Its VP Corporate Development and Treasurer Attorney-In-Fact Its

Attachments:

Attachment A, Table of Contents (R8) to Purchase Agreement No. 1977

Table 1A (R6) to Purchase Agreement No. 1977, 737-800 Exercised Option Delivery, Description, Price and Advance Payments

Attachment B (R8) to Letter Agreement 6-1162-AKP-075, Aircraft Purchase Rights and Substitution Rights

Supplement Exhibit BFE1 (R8) to Purchase Agreement No. 1977, 737-823 BFE Variables

SA <u>ARTICLES</u>	NUMBER		
1.	Quantity, Model and Description	SA21	
2.	Delivery Schedule		
3.	Price		
4.	Payment		
5.	Miscellaneous		
TABLE			
1(R3) Advance Pa Schedule 1 1A Advance Pa	(R2) SA23 (R6)	Option Aircraft Delivery, Description, Price and	SA27
<u>EXHIBITS</u>			
А.	Aircraft Configuration		
В.	Aircraft Delivery Requirements and Responsibilities		
C.	Defined Terms		
<u>SUPPLEMI</u>	ENTAL EXHIBITS		
AE1	Escalation Adjustment Airframe and Optional Features SA20		
BFE1(R8)		BFE Variables SA27	
CS1	Customer Support Variables		
SLP1	Service Life Policy Components		
EE1	Engine Escalation, Engine Warranty and Patent Indemnity		

LETTER AGREEMENTS

6-1162-AKP-070Miscellaneous Commitments for Model 737, 757, 767 and 777 Aircraft

6-1162-AKP-071

Purchase Obligations

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-074R2 Business Considerations

6-1162-AKP-075Aircraft Purchase Rights and Substitution Rights

- Attachment A

- Attachment B(R8) SA27

- Attachment C(R4) SA26

6-1162-AKP-076 Aircraft Performance Guarantees

6-1162-AKP-077 Spares Matters

6-1162-AKP-078 Model 737 Miscellaneous Commitments

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-080 Installation of Cabin Systems Equipment

6-1162-AKP-081 Model 737 Maintenance Cost Commitment

6-1162-AKP-082 Confidentiality

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] 6-1162-AKP-117 Delivery Schedule

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

BUYER FURNISHED EQUIPMENT VARIABLES

between

THE BOEING COMPANY

and

AMERICAN AIRLINES, INC.

Supplemental Exhibit BFE1 to Purchase Agreement Number 1977

BUYER FURNISHED EQUIPMENT VARIABLES

relating to

BOEING MODEL 737 AIRCRAFT

This Supplemental Exhibit BFE1 contains vendor selection dates, on-dock dates and other variables applicable to the Aircraft.

1. <u>Supplier Selection</u>.

Customer will:

1.1 Select and notify Boeing of the suppliers and part numbers of the following BFE items by the following dates:

Galley System	Complete
Galley Inserts	Complete
Seats (passenger)	Complete
Cabin Systems Equipment	Complete
Miscellaneous Emergency Equipment	Complete
Cargo Handling Systems	Complete
For a new cortification cumplior requires patification [CON	

For a new certification, supplier requires notification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] prior to Cargo Handling System on-dock date.

2. **On-dock** Dates

On or before [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT], Boeing will provide to Customer a BFE Requirements On-Dock/Inventory Document (BFE Document) or an electronically transmitted BFE Report which may be periodically revised, setting forth the items, quantities, on-dock dates and shipping instructions relating to the in-sequence installation of BFE. For planning purposes, a preliminary BFE on-dock schedule is set forth below:

<u>Item</u>

tem <u>Preliminary On-Dock Dates</u> [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Seats

Aircraft Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets

Preliminary On-Dock Dates

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Aircraft

	Aircraft
CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY	
WITH THE COMMISSION PURSUANT TO A REQUEST FOR	
CONFIDENTIAL TREATMENT]	
CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY	
WITH THE COMMISSION PURSUANT TO A REQUEST FOR	
CONFIDENTIAL TREATMENT]	
Seats	
Galleys/Furnishings	
Antennas & Mounting Equipment	
Avionics	
Cabin Systems Equipment	
Miscellaneous Emergency Equipment	
Textiles/Raw Material	
Cargo Systems	
Provision Kits	
Winglets	

Item

Preliminary On-Dock Dates [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

		Aircraft	Aircraft
[CONFIDENTIAL PORTION OMITTE] WITH THE COMMISSION PURSUAN CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTE] WITH THE COMMISSION PURSUAN CONFIDENTIAL TREATMENT] Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets	T TO A REQUEST FOR D AND FILED SEPARATELY		
Item	Preliminary On-Dock Dates		
		Aircraft	Aircraft
Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets			
Item [CONFIDENTIAL PORTION OMITT	<u>Preliminary On-Dock Dates</u> ED AND FILED SEPARATELY W	VITH THE COMMISSION PURSUANT TO A RE	EQUEST FOR CONFIDENTIAL TREATMENT]
		Aircraft	Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Seats Galleys/Furnishings Antennas & Mounting Equipment

Avionics

<u>Item</u>

Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets		Aircraft	Aircraft
Item	Preliminary On-Dock Dates	THE COMMISSION DUDSUANT TO A DI	CLIECT FOD CONFIDENTIAL TDE ATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets

Preliminary On-Dock Dates

Aircraft

		Aircraft	Aircraft
Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets			
Item	Preliminary On-Dock Dates		
Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets		<u>Aircraft</u>	Aircraft
Item	Preliminary On-Dock Dates		
Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets		Aircraft	Aircraft
Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits	<u>Preliminary On-Dock Dates</u>	Aircraft	Aircraft

<u>Item</u>

Seats

Aircraft

Aircraft

Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Item Preliminary On-Dock Dates

Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits

Winglets

3. Additional Delivery Requirements

Customer will insure that Customer's BFE suppliers provide sufficient information to enable Boeing, when acting as Importer of Record for Customer's BFE, to comply with all applicable provisions of the U.S. Customs Service.

Table 1A (R6) to Purchase Agreement No. 1977

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

PURCHASE AGREEMENT No. 1977 SUPPLEMENT No. 28

PURCHASE AGREEMENT No. 1977 SUPPLEMENT No. 28 (this "Agreement") dated June _____, 2008 between The Boeing Company ("Boeing") and American Airlines, Inc. ("Customer").

$\underline{R} \, \underline{E} \, \underline{C} \, \underline{I} \, \underline{T} \, \underline{A} \, \underline{L} \, \underline{S}:$

- A. Boeing and Customer have heretofore entered into that certain Purchase Agreement No. 1977, dated October 31, 1997, as amended and supplemented, (capitalized terms used herein without definition shall have the meanings specified therefor in such Purchase Agreement).
- B. Pursuant to Letter Agreement No. 6-1162-AKP-075 titled Aircraft Purchase Rights and Substitution Rights (the "Rights Letter"), Boeing and Customer have agreed to, among other things, treatment of aircraft Purchase Rights.
- C. Customer and Boeing desire to amend and supplement the Purchase Agreement as provided below.

In consideration of the foregoing premises and other good and sufficient consideration, Boeing and Customer hereby agree as follows:

1. Amendment to Reflect Customer's Exercise of MADP Rights.

- 1.1. The Purchase Agreement is amended and supplemented to reflect the exercise of [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Rights Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] by Customer with the Scheduled Delivery [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to A REQUEST FOR CONFIDENTIAL TREATMENT] by Customer with the Scheduled Delivery [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 1.2. The Scheduled Delivery Month and Advanced Payment Schedule for the aircraft is set forth in Table 1A (R7), attached hereto.

1.3. In recognition of an update to the configuration of Customer's Model 737-823 Aircraft, the revised Detail Specification is incorporated into a new Exhibit A1 attached hereto and made apart hereof. The new Exhibit A1 reflects the Optional Features selected for the Aircraft.

- 2. <u>Table of Contents</u>. The Table of Contents dated May 2008 is replaced in its entirety with the revised Table of Contents dated June 2008 (Attachment A hereto) to reflect amendments made to the Purchase Agreement by this Agreement.
- 3. <u>MADP Rights</u>. Pursuant to the Rights Letter, Attachment B to such letter is hereby replaced in its entirety with the revised Attachment B (R9) attached hereto and hereby incorporated into the Purchase Agreement. The revised number of certain Customer MADP rights pursuant to this Agreement is reflected in the Attachment B (R9) hereto.
- 4. <u>Supplement Exhibit BFE1</u>. Supplement Exhibit BFE1 (R8) is hereby replaced in its entirety with the revised Supplement Exhibit BFE1 (R9) attached hereto and hereby incorporated into the Purchase Agreement. The updated on-dock dates for all contracted firm Aircraft thru December 2013 are reflected in the attached Supplement Exhibit BFE1 (R9).
- 5. [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT
- 6. Advance Payments for Aircraft. Due at signing of this Agreement, Customer owes Boeing [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 7. <u>Effect on Purchase Agreement.</u> Except as expressly set forth herein, all terms and provisions contained in the Purchase Agreement shall remain in full force and effect. This Agreement contains the entire agreement between the parties with respect to the subject matter hereof and supersedes all previous proposals, and agreements, understandings, commitments or representations whatsoever, oral or written, with respect to the subject matter hereof and may be changed only in writing signed by authorized representatives of the parties.

By

Its

IN WITNESS WHEREOF, Boeing and Customer have each caused this Agreement to be duly executed as of the day and year first above written.

AMERICAN AIRLINES, INC.

By

THE BOEING COMPANY

Attornev-In-Fact

Its VP Corporate Development and Treasurer

Attachments:

Exhibit A1, Aircraft Configuration

Attachment A, Table of Contents (R9) to Purchase Agreement No. 1977

Table 1A (R7) to Purchase Agreement No. 1977, 737-800 Exercised Option Delivery, Description, Price and Advance Payments

Attachment B (R9) to Letter Agreement 6-1162-AKP-075, Aircraft Purchase Rights and Substitution Rights

Supplement Exhibit BFE1 (R9) to Purchase Agreement No. 1977, 737-823 BFE Variables

Letter Agreement 6-1162-CLO-1035, [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

SA <u>ARTICLES</u> <u>NUMBER</u>

- 1. Quantity, Model and Description SA21
- 2. Delivery Schedule
- 3. Price
- 4. Payment
- 5. Miscellaneous

TABLE

SLP1

1(R3) Advance Pay -Schedule 1 1A Advance Pay	(R2) SA23 (R7)	Option Aircraft Delivery, Description, Price and	SA28
<u>EXHIBITS</u>			
A. A1	Aircraft Configuration Aircraft Configuration SA28		
В.	Aircraft Delivery Requirements and Responsibilities		
C.	Defined Terms		
<u>SUPPLEME</u>	INTAL EXHIBITS		
AE1	Escalation Adjustment Airframe and Optional Features SA20		
BFE1(R9)		BFE Variables SA28	
CS1	Customer Support Variables		

EE1 Engine Escalation, Engine Warranty and Patent Indemnity

Service Life Policy Components

LETTER AGREEMENTS

6-1162-AKP-070Miscellaneous Commitments for Model 737, 757, 767 and 777 Aircraft

6-1162-AKP-071

Purchase Obligations

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-074R2 Business Considerations

6-1162-AKP-075Aircraft Purchase Rights and Substitution Rights

- Attachment A

- Attachment B(R9) SA28

- Attachment C(R4) SA26

6-1162-AKP-076

Aircraft Performance Guarantees

6-1162-AKP-077 Spares Matters

6-1162-AKP-078 Model 737 Miscellaneous Commitments

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-080 Installation of Cabin Systems Equipment

6-1162-AKP-081 Model 737 Maintenance Cost Commitment

6-1162-AKP-082 Confidentiality

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

6-1162-AKP-117 Delivery Schedule

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]. American Airlines, Inc. P.O. Box 619616 Dallas-Fort Worth Airport, Texas 75261-9616

Subject: [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Reference: Purchase Agreement No. 1977 (the Purchase Agreement) between The Boeing Company (Boeing) and American Airlines, Inc. (Customer) relating to Model 737-823 aircraft (the Aircraft)

This letter agreement (Letter Agreement) amends and supplements the Purchase Agreement. All terms used but not defined in this Letter Agreement have the same meaning as in the Purchase Agreement.

Boeing agrees to provide Customer with the [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] in the Attachment.

The [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] set forth herein are applicable to the Aircraft set forth in the Attachment to this Letter Agreement No. 6-1162-CLO-1035.

Notwithstanding any other provisions of the Purchase Agreement, the rights and obligations described in this Letter Agreement are provided to Customer in consideration of Customer becoming the operator of the Aircraft and cannot be assigned, in whole or in part.

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Very truly yours,

THE BOEING COMPANY

Ву_____

Its Attorney-In-Fact

ACCEPTED AND AGREED TO this Date: _____, 2008

AMERICAN AIRLINES, INC.

Ву_____

Its

FOR AMERICAN AIRLINES, INC.

SECTION	CONTENTS
1	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
	2
	3
	4
	5
	6
	7
	8

- 1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].
- 2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.5 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 2.4.6 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.1.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 4.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUESTAIRCRAFT FOR CONFIDENTIAL TREATMENT]5 CONFIGURATION
- 5.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 5.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.1 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.2 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.3 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.4 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
- 6.5 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL

	TREATMENT]
6.6	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
6.7	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
6.8	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
6.9	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
6.10	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7.1	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7.2	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7.3	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7.4	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7.5	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7.6	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7.7	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7.8	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]
7.9	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

8 [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Exhibit A1 to Purchase Agreement No. 1977

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

Table 1A (R7) to Purchase Agreement No. 1977

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

BUYER FURNISHED EQUIPMENT VARIABLES

between

THE BOEING COMPANY

and

AMERICAN AIRLINES, INC.

Supplemental Exhibit BFE1 to Purchase Agreement Number 1977

BUYER FURNISHED EQUIPMENT VARIABLES

relating to

BOEING MODEL 737 AIRCRAFT

This Supplemental Exhibit BFE1 contains vendor selection dates, on-dock dates and other variables applicable to the Aircraft.

1. <u>Supplier Selection</u>.

Customer will:

1.1 Select and notify Boeing of the suppliers and part numbers of the following BFE items by the following dates:

Galley System	Complete
Galley Inserts	Complete
Seats (passenger)	Complete
Cabin Systems Equipment	Complete
Miscellaneous Emergency Equipment	<u>Complete</u>
Cargo Handling Systems	<u>Complete</u>

For a new certification, supplier requires notification [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] to Cargo Handling System on-dock date.

۱.	
)	
•	

2. <u>On-dock Dates</u>

On or before [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT], Boeing will provide to Customer a BFE Requirements On-Dock/Inventory Document (BFE Document) or an electronically transmitted BFE Report which may be periodically revised, setting forth the items, quantities, on-dock dates and shipping instructions relating to the in-sequence installation of BFE. For planning purposes, a preliminary BFE on-dock schedule is set forth below:

<u>Item</u>	Preliminary On-Dock Dates		
		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft
		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	
Item	Preliminary On-Dock Dates		
		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft	Aircraft
[CONFIDENTIAL PORTION OMITTEI WITH THE COMMISSION PURSUAN CONFIDENTIAL TREATMENT] Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets			
Item	Preliminary On-Dock Dates		
Seats		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft
Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miccollanoous Emergency Equipment			

Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets

<u>Item</u>

[CONFIDENTIAL PORTION OMITTED AND[CONFIDENTIAL PORTION OMITTED ANDFILED SEPARATELY WITH THEFILED SEPARATELY WITH THE

COMMISSION PURSUANT TO A REQUEST COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] FOR CONFIDENTIAL TREATMENT] Aircraft Aircraft [CONFIDENTIAL PORTION OMITTED AND Seats FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets Preliminary On-Dock Dates <u>Item</u> [CONFIDENTIAL PORTION OMITTED AND [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] FOR CONFIDENTIAL TREATMENT] Aircraft Aircraft [CONFIDENTIAL PORTION OMITTED AND Seats FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets Preliminary On-Dock Dates Item [CONFIDENTIAL PORTION OMITTED AND [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] FOR CONFIDENTIAL TREATMENT] Aircraft Aircraft [CONFIDENTIAL PORTION OMITTED AND Seats FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets Item Preliminary On-Dock Dates [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE Seats COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment

Textiles/Raw Material Cargo Systems

Item	Preliminary On-Dock Dates		
		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft
Seats		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	
Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets			
Item	Preliminary On-Dock Dates		
		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft
Seats		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	
Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets			
Item	Preliminary On-Dock Dates		
		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft	[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft
Seats		[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]	
Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets			
Item	Preliminary On-Dock Dates		

FILED SEPARATELY WITH THE

Aircraft

FOR CONFIDENTIAL TREATMENT]

COMMISSION PURSUANT TO A REQUEST

[CONFIDENTIAL PORTION OMITTED AND [CONFIDENTIAL PORTION OMITTED AND

Aircraft

FILED SEPARATELY WITH THE

FOR CONFIDENTIAL TREATMENT]

COMMISSION PURSUANT TO A REQUEST

CONFIDENTIAL TREATMENT] Seats Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets

Item

Preliminary On-Dock Dates

FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

[CONFIDENTIAL PORTION OMITTED AND [CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft

Seats

Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits Winglets

Item

Seats

Winglets

Preliminary On-Dock Dates

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT] Aircraft

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT]

Galleys/Furnishings Antennas & Mounting Equipment Avionics Cabin Systems Equipment Miscellaneous Emergency Equipment Textiles/Raw Material Cargo Systems Provision Kits

3. Additional Delivery Requirements

Customer will insure that Customer's BFE suppliers provide sufficient information to enable Boeing, when acting as Importer of Record for Customer's BFE, to comply with all applicable provisions of the U.S. Customs Service.

Attachment B (R9) to Letter Agreement 6-1162-AKP-075 (Model 737)

[CONFIDENTIAL PORTION OMITTED AND FILED SEPARATELY WITH THE COMMISSION PURSUANT TO A REQUEST FOR CONFIDENTIAL TREATMENT].

STOCK PURCHASE AGREEMENT

BY AND AMONG

AMERICAN BEACON ADVISORS, INC.,

AMR CORPORATION,

AND

LIGHTHOUSE HOLDINGS, INC.

Dated as of April 16, 2008

TABLE OF CONTENTS

ARTICLE I	DEFINITIONS	1
1.1	Definitions	1
	"SEC" means the United States Securities and Exchange Commission10	
ARTICLE II	PURCHASE AND SALE	12
2.1	Purchase and Sale of the Shares	12
	ARTICLE IIIPURCHASE PRICE13	
3.1	Purchase Price	13
3.2	Working Capital Adjustment to Purchase Price	13
3.3	Determination of the Working Capital Adjustment Amount	13
	ARTICLE IVCLOSING14	
4.1	Closing Date	14
4.2	Payment on the Closing Date	14
4.3	Buyer's Additional Closing Date Deliveries	14
4.4	Seller's Closing Date Deliveries	15
	ARTICLE VREPRESENTATIONS AND WARRANTIES OF SELLER16	
5.1	Organization of AMR	16
5.2	Organization; Capital Structure of the Company; Power and Authority	16
5.3	Subsidiaries and Investments	17
5.4	Authority of Seller; Conflicts	17
5.5	Financial Statements	18
5.6	Operations Since Financial Statements Date	18
5.7	Taxes	19
5.8	Governmental Permits	19
5.9	Assets Under Management	19
5.10	Real Property	21
5.11	Personal Property Leases	22
5.12	Intellectual Property	22
5.13	Assets	24
5.14	No Violation, Litigation or Regulatory Action	24
5.15	Contracts	25
5.16	Status of Contracts	26
5.17	ERISA	26
5.18	Employee Relations and Agreements	27
5.19	Insurance	28
5.20	No Brokers	28
5.21	Registration as Investment Adviser	28
5.22	Filing of Reports; Maintenance of Records	30

5.23	Fee Arrangements, Expenses and Custody of Funds	30
5.24	Fund Matters	30
5.25	Exempt Fund Clients	33
5.26	Information in Proxy Statements	33
5.27	No Undisclosed Liabilities	33
5.28	Transactions with Affiliates	34
5.29	Illegal Payments	34
5.30	Investment Intent; Accredited Investor	34
	ARTICLE VIREPRESENTATIONS AND WARRANTIES OF BUYER34	
6.1	Organization of Buyer	34
6.2	Authority of Buyer; Conflicts	35
6.3	Capital Structure of Parent	35
6.4	No Violation, Litigation or Regulatory Action	36
6.5	Financing	36
6.6	No Brokers	36
6.7	Ineligible Persons	36
6.8	Facts Affecting Regulatory Approvals	37
6.9	Section 15 of the Investment Company Act	37
6.10	Information in Proxy Statements	37
6.11	Investment Representations	37
6.12	No Currently Expected Claims	37
	ARTICLE VIIACTION PRIOR TO THE CLOSING DATE37	
7.1	Access to Information	37
7.2	Notifications	38
7.3	Consents of Third Parties; Governmental Approvals	38
7.4	Operations Prior to the Closing Date	39
7.5	Antitrust Law Compliance	42
7.6	Client Consents	42
7.7	Fund Approvals	43
7.8	Approval of Exempt Fund Clients	43
	ARTICLE VIIIADDITIONAL AGREEMENTS44	
8.1	Employee Matters	44
8.2	Insurance; Risk of Loss	47
8.3	Section 15 of the Investment Company Act	47
8.4	Covenant Not to Compete	47
8.5	Confidentiality; Covenant Not to Solicit	48
8.6	Relief for Violation	49
8.7	Client Fees	49
8.8	Transfer Expenses	49
8.9	Fees of Independent Fiduciary	49
8.10	338(h)(10) Election	49

8.11	Intellectual Property	49
8.12	Audited Financials	50
8.13	AMR Plans	50
8.14	Transition Services Agreement	50
	ARTICLE IXCONDITIONS PRECEDENT TO OBLIGATIONS OF BUYER50	
9.1	No Misrepresentation or Breach of Covenants and Warranties	51
9.2	No Restraint	51
9.3	Governmental Approvals	51
9.4	No Litigation	51
9.5	Closing Revenue Run-Rate	51
9.6	New Management Contracts	51
9.7	Closing Date Deliveries	52
	ARTICLE XCONDITIONS PRECEDENT TO OBLIGATIONS OF SELLER AND THE COMPANY52	
10.1	No Misrepresentation or Breach of Covenants and Warranties	52
10.2	No Restraint	52
10.3	Governmental Approvals	52
10.4	No Litigation	52
10.5	Closing Revenue Run-Rate	53
10.6	New Management Contracts	53
10.7	Section 15 Compliance	53
10.8	Closing Date Deliveries	53
	ARTICLE XIINDEMNIFICATION53	
11.1	Indemnification by Seller	53
11.2	Indemnification by Buyer	55
11.3	Notice of Claims	57
11.4	Determination of Amount	57
11.5	Third Person Claims	59
11.6	Limitations	60
	ARTICLE XIITERMINATION61	
12.1	Termination	61
12.2	Notice of Termination	62
12.3	Effect of Termination	62
12.4	Right to Specific Performance	62
	ARTICLE XIIIGENERAL PROVISIONS62	
13.1	Survival of Representations and Warranties	62
13.2	Governing Law	62
13.3	No Public Announcement	62
13.4	Notices	63
13.5	Successors and Assigns	64
13.6	Access to Records after Closing	64

- 13.7 Entire Agreement; Amendments
- 13.8 Interpretation
- 13.9 Waivers
 - 13.10Expenses66
 - 13.11Partial Invalidity66
 - 13.12Execution in Counterparts66
 - 13.13Further Assurances66
 - 13.14Disclaimer of Warranties66

--

66

Exhibits

Exhibit A	Transition Services
Exhibit B	[Reserved]
Exhibit C	[Reserved]
Exhibit D	Form of Client Consent Notice
Exhibit E	Form of ERISA Client Notice
Exhibit F	Form of Exempt Fund Client Notice
Exhibit G	Travel Privileges Agreement
Exhibit H	[Reserved]
Exhibit I	AAdvantage Participation Agreement

List of Schedules

Schedules

3.3	Net Working Capital
5.2	Qualifications to Conduct Business
5.3	Subsidiary and Investments
5.4(b)	No Conflicts
5.5	Financial Statements; Exceptions
5.6	Operations Since Financial Statements Date
5.7	Taxes
5.8	Governmental Permits
5.9	Assets Under Management
5.10	Real Property
5.11	Personal Property Leases
5.12	Intellectual Property Matters
5.14	Violation, Litigation or Regulatory Action of the Company
5.15	Contracts
5.16	Status of Contracts
5.17(a)	Welfare Plans and Pension Plans
5.17(c)	Compliance of Welfare Plans and Pension Plans
5.17(d)	Other Employee Benefits
5.18	Employee Relations and Agreements
5.19	Insurance
5.20	Broker
5.21	Investment Adviser Registration
5.24(a)	Fund Agreements
5.24(b)	No Regulatory Action
5.24(g)	Fund Taxation
5.27	Undisclosed Liabilities
5.28	Transactions With Affiliates
6.3	Violation, Litigation or Regulatory Action of Buyer
6.5	Broker
7.4(b)	Operations Prior to Closing Date; Methodology for Pre-Closing Dividends
8.1(c)	Buyer's Benefit Programs
8.1(d)	Unused Vacation and Bonus Programs

933225

STOCK PURCHASE AGREEMENT

THIS STOCK PURCHASE AGREEMENT, dated as of April 16, 2008, is by and among American Beacon Advisors, Inc., a Delaware corporation (the "<u>Company</u>"), and Lighthouse Holdings, Inc., a Delaware corporation ("<u>Buyer</u>"), and AMR Corporation, a Delaware corporation and sole stockholder of the Company ("<u>AMR</u>" or "<u>Seller</u>").

PRELIMINARY STATEMENT:

WHEREAS, AMR is the owner of all of the outstanding shares of capital stock of the Company, and AMR desires to sell, and Buyer desires to purchase 1,000 shares (the "<u>Shares</u>") of the issued and outstanding common stock, \$1.00 par value per share (the "<u>Common Stock</u>") constituting all of the issued and outstanding Common Stock of the Company, all on the terms and subject to the conditions set forth herein; and

WHEREAS, the Company is a registered investment adviser under the Advisers Act (as defined below) and is in the business of providing Investment Management Services (the "Business");

NOW, THEREFORE, in consideration of the mutual covenants and agreements hereinafter set forth, it is hereby agreed as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions.</u> In this Agreement, the following terms have the meanings specified or referred to in this <u>Section 1.1</u> and shall be equally applicable to both the singular and plural forms. Any agreement referred to below shall mean such agreement as amended, supplemented and modified from time to time to the extent permitted by the applicable provisions thereof and by this Agreement.

"338(h)(10) Election" has the meaning specified in Section 8.10.

"<u>AA</u>" has the meaning specified in <u>Section 5.18(a)</u>.

"<u>AAdvantage Participation Agreement</u>" means the AAdvantage Participation Agreement dated as of the Closing Date between AA and the Company in the form attached hereto as <u>Exhibit I</u>.

"Accounting Firm" has the meaning specified in Section 3.3(c).

"Adjusted Assets Under Management" means, for any Client account as of any date, the amount, expressed in U.S. dollars, of assets under management by such Person in each such account as of such date, calculated in the same manner as provided for the calculation of base investment management fees payable to such Person or an Affiliate in respect of such account by the terms of the Management Contract applicable to such account; <u>provided</u> in each case that:

(I) additions, contributions, withdrawals, redemptions and repurchases to such Client account shall be taken into account upon the earlier of when notice of such additions, contributions, withdrawals, redemptions or repurchases is provided to the Company (or, with respect to the Funds, to the Funds or their transfer agents) and when funded to or from such account;

(II) any assets under management for any account for which the Person in question or an Affiliate acts as investment adviser and sub-adviser shall be counted only once; and

(III) any assets under management for any set of accounts one of which invests in the other shall be counted only once (except to the extent additional revenue is earned) if the Person in question or an Affiliate acts as investment adviser to both accounts.

"Advisers Act" means the Investment Advisers Act of 1940, as amended.

"Affected Employees" has the meaning specified in Section 8.1(a).

"Affiliate" means, with respect to any Person, any other Person which directly or indirectly controls, is controlled by or is under common control with such Person.

"Alternative Benefit Programs" has the meaning specified in Section 8.1(c).

"<u>American Beacon Funds</u>" means each of the Large Cap Value Fund, International Equity Fund, Small Cap Value Fund, Balanced Fund, Emerging Markets Fund, Mid-Cap Value Fund, Large Cap Growth Fund, Small Cap Value Opportunity Fund, International Equity Index Fund, S&P 500 Index Fund, Small Cap Index Fund, Money Market Fund, U.S. Government Money Market Fund, High Yield Bond Fund, Intermediate Bond Fund, Enhanced Income Fund, Short Term Bond Fund and Treasury Inflation Protected Securities Fund of the American Beacon Funds Trust.

"American Beacon Funds Trust" means the American Beacon Funds, a Massachusetts business trust.

"American Beacon Master Funds" means the Master Money Market Portfolio Fund and the Master U.S. Government Money Market Portfolio Fund of the Master

"AMR" has the meaning specified in the first paragraph of this Agreement.

"AMR Equity Awards" has the meaning specified in Section 8.1(l).

"APEP" collectively means American Private Equity Partners, L.P., a Delaware limited partnership, American Private Equity Partners II, L.P., a Delaware limited partnership and American Private Equity Partners III, L.P., a Delaware limited partnership.

"Applicable Law" means any federal, state, local or foreign law, statute, ordinance, rule or regulation of any Governmental Body.

"Base Date" means March 31, 2008.

Trust.

"Base Net Working Capital Amount" means \$5,000,000. For the avoidance of doubt, the Base Net Working Capital Amount does not include the Company's payment obligations under Section 8.1(k).

"Base Revenue Run-Rate" means the Revenue Run-Rate for the Company and the Subsidiary as of the Base Date.

"Business" has the meaning specified in the second "WHEREAS" clause of this Agreement.

"Business Agreements" has the meaning specified in Section 5.16.

"Business Day" shall mean any day that is not a Saturday, Sunday, legal holiday or other day on which commercial banks in New York, New York are authorized or required by law to close.

"Buyer" has the meaning specified in the first paragraph of this Agreement.

"Buyer Ancillary Agreements" means all agreements, instruments and documents being or to be executed and delivered by Buyer under this Agreement or in connection herewith.

"Buyer Group Member" means Buyer and its Affiliates, their respective directors, officers, employees and agents, each person who controls any of them within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act, and their respective successors and permitted assigns.

"Buyer's Benefit Programs" has the meaning specified in Section 8.1(c).

"Calculation Date" means a date within five (5) Business Days prior to the Closing Date.

"Claim Notice" has the meaning specified in Section 11.3.

"Client" means each Person who is a party to a Management Contract other than the Company or its Subsidiary.

"<u>Client Consent</u>" means the requisite consent or deemed consent of each Client of the Company (other than the Funds and the Exempt Fund Clients) to a change in control of the Company in connection with the transactions contemplated by this Agreement, as required by the Advisers Act or the Investment Company Act, and the respective rules and regulations promulgated thereunder, as interpreted by any relevant no-action letters or similar pronouncements issued by the SEC or its staff; <u>provided</u>, that in the case of the Related Clients and ERISA Clients, "Client Consent" means the entry by such Clients into respective new Management Contracts as contemplated by <u>Sections 9.6</u> and <u>10.6</u>.

"Closing" means the closing of the transfer of the Shares from AMR to Buyer.

"Closing Date" has the meaning specified in Section 4.1.

"Closing Cash Purchase Price" has the meaning specified in Section 3.1.

"<u>Closing Revenue Run-Rate</u>" means the Revenue Run-Rate for the Company and the Subsidiary generated by Adjusted Assets Under Management of the Company as of the Calculation Date held by clients from whom the Company has received Client Consents, held by Funds which have effected Fund Approvals and held by Exempt Fund Clients which have effected Exempt Fund Client Approvals.

"Code" means the Internal Revenue Code of 1986, as amended.

"Common Stock" has the meaning specified in the first "WHEREAS" clause in this Agreement.

"Company" has the meaning specified in the first paragraph of this Agreement.

"<u>Company Ancillary Agreements</u>" means all agreements, instruments and documents being or to be executed and delivered by AMR or the Company under this Agreement or in connection herewith.

"Company Intellectual Property Rights" has the meaning specified in Section 5.12(a).

"Confidentiality Agreement" means that certain letter agreement dated December 20, 2007, as amended.

"Copyrights" has the meaning specified in Section 5.12.

"Court Order" means any judgment, order, award or decree of any foreign, federal, state, local or other court or tribunal and any award in any arbitration proceeding.

"Encumbrance" means any lien, claim, charge, security interest, mortgage, pledge, easement, conditional sale or other title retention agreement, defect in title or other restrictions of a similar kind.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Client" has the meaning specified in Section 5.9(c).

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Exempt Fund Client" means each Client that is excepted from the definition of investment company by virtue of Section 3(c)(1) or 3(c)(7) of the Investment Company Act, including, but not limited to: APEP; Cash Plus Trust, a Massachusetts business trust; American Beacon Global Funds SPC, a Cayman Islands segregated corporation; and American Beacon Global Funds, PLC, a public limited company organized under the laws of Ireland.

"Exempt Fund Client Approval" means the requisite consent of each Exempt Fund Client of the Company to the change in control of the Company in connection with the transactions contemplated by this Agreement as required by the Advisers Act and the rules and regulations promulgated thereunder, as interpreted by any relevant no-action letters or similar pronouncements issued by the SEC or its staff, including, where required, any notice to, or approval of such consent by, Exempt Fund Client Investors as required by Applicable Law or the organizational documents of an Exempt Fund Client.

"Exempt Fund Client Investor" has the meaning specified in Section 7.8.

"Existing Policy" has the meaning specified in Section 8.2(b).

"Expenses" means any and all reasonable out-of-pocket expenses incurred in connection with defending or asserting any claim, action, suit or proceeding incident to any matter indemnified against hereunder (including court filing fees, court costs, arbitration fees or costs, witness fees and reasonable fees and disbursements of legal counsel, expert witnesses, accountants and other professionals).

"Final Cash Purchase Price" has the meaning specified in Section 3.1.

"Financial Statements" means the unaudited consolidated balance sheet of the Company and the Subsidiary as of the Financial Statements Date, and the related statement of income for the fiscal year then ended, included in <u>Schedule 5.5</u>.

"Financial Statements Date" means December 31, 2007.

"Fund Agreements" has the meaning specified in Section 5.24(a).

"Fund Approvals" means the requisite Fund Board Approvals and Fund Shareholder Approvals in connection with the transactions contemplated by this Agreement as required by the Investment Company Act and the rules and regulations thereunder, as interpreted by any relevant no-action letters or similar pronouncements issued by the SEC or its staff.

"Fund Board Approvals" means, with respect to each Fund that is a Client of the Company as of the date of this Agreement, the due consideration and approval by the board of trustees of the Trust under which such Fund has been formed, including where required the requisite approval of a majority of the trustees who are not parties thereto or Interested Persons of any such party, of (i) a new advisory agreement to be in effect with the Company from and after the Closing on terms substantially identical in all material respects (and identical as to fees) to the Management Contract existing between the Company and such Fund as of the date of this Agreement (collectively, the "New Fund Advisory Agreements"), (ii) a new administrative services agreement, if there is in effect as of the date hereof an existing administration agreement to which the Company is a party with respect to such Fund, to be in effect with the Company from and after the Closing on terms substantially identical in all materials as to fees) to such existing administration agreement and (iii) the composition of the board of trustees of the respective Trust under which such Fund has been formed, effective as of (and subject to) the Closing, to satisfy the requirements of Section 15(f) of the Investment Company Act, including by the addition of a sufficient number of trustees who are not Interested Persons of Buyer or the Company who shall have been selected by the existing trustees of such Trust who are not such Interested Persons.

"Fund Financial Statements" has the meaning specified in Section 5.24(h).

"Fund Regulatory Documents" has the meaning specified in Section 5.24(f).

"Fund Shareholder Approvals" means approval of the shareholders of each Fund for which Fund Board Approval is obtained of (i) such New Fund Advisory Agreement approved by the board of trustees of the Trust under which such Fund is formed and (ii) such board composition approved by such board of trustees, in each case as described in the definition of "Fund Board Approvals", including the preparation and mailing to such shareholders a proxy statement describing the transactions contemplated hereby, such New Fund Advisory Agreement and such board composition arrangements, and by holding the shareholder meeting as promptly as practicable.

"Funds" means American Beacon Master Funds, the American Beacon Funds, Mileage Funds and Select Funds.

"GAAP" means United States generally accepted accounting principles, consistently applied in Seller's financial statements, in effect at the date of the financial statement to which it refers.

"Governmental Body." means any foreign, federal, state, local or other governmental authority or regulatory body.

"Governmental Permits" has the meaning specified in Section 5.8.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Immediate Family" means, with respect to any natural person, (a) such person's spouse, parents, grandparents, children, grandchildren and siblings and (b) such person's former spouse(s) and current spouses of such person's children, grandchildren and siblings and (c) estates, trusts, partnerships and other entities of which substantially all of the interest is held directly or indirectly by the foregoing.

"Indebtedness" means (a) all indebtedness for borrowed money, (b) any other indebtedness that is evidenced by a note, bond, debenture or similar instrument, (c) all obligations under financing leases, (d) all obligations in respect of acceptances issued or created, (e) any guarantees of the Indebtedness of other Persons and (f) all restricted cash or cash equivalent balances. For the avoidance of doubt Indebtedness does not include amounts owed pursuant to <u>Section 8.1(k)</u>.

"Indemnified Party" has the meaning specified in Section 11.3.

"Indemnitor" has the meaning specified in Section 11.3.

"Intellectual Property Rights" has the meaning specified in Section 5.12.

"Interested Person" has the meaning specified in the Investment Company Act.

"Investment Company Act" means the Investment Company Act of 1940, as amended.

"Investment Management Services" means any services which involve (i) the management of an investment account or fund (or portions thereof or a group of investment accounts or funds) for compensation, (ii) the giving of advice or the provision of administration services with respect to the investment and/or reinvestment of assets or funds (or any group of assets or funds) for compensation or (iii) otherwise acting as an "investment adviser" within the meaning of the Advisers Act, and performing activities related or incidental thereto.

"Knowledge of the Company" means, as to a particular matter, the actual knowledge of the following persons: William F. Quinn, Douglas G. Herring, Michael Fields, Wyatt Crumpler, Brian Brett, Becky Harris, Rosemary Behan, Terri McKinney, Bo Ragsdale and Donna Merchant.

"Leases" has the meaning specified in Section 5.10(b).

"Losses" means any and all out-of-pocket losses, costs, settlement payments, awards, judgments, fines, penalties, damages, expenses, deficiencies or other charges, it being understood that, Losses shall not include punitive damages other than punitive damages owed to a third party.

"Management Contract" means each investment management, advisory, sub-advisory or administration contract relating to Investment Management Services provided by the Company or the Subsidiary.

"Master Trust, a Massachusetts business trust.

"<u>Material Adverse Effect</u>" means any change, effect, event, occurrence, state of facts or development that (a) is materially adverse to the business, financial condition or results of operations of the Company and its Subsidiary, taking Company and Subsidiary together as a whole, or (b) would materially impair the ability of the Company to perform its obligations under this Agreement; provided, however, that none of the following shall be deemed in themselves, either alone or in combination, to constitute, and none of the following shall be taken into account in determining whether there has been or will be, a Material Adverse Effect: (i) any adverse change, effect, event, occurrence, state of facts or

development to the extent attributable to the announcement or pendency of the transactions contemplated by this Agreement (including any cancellations of or delays in customer orders, any reduction in sales, any disruption in supplier, distributor, partner or similar relationships or any loss of employees); (ii) any adverse change, effect, occurrence, state of facts or developments attributable to conditions affecting the industries in which the Company participates, the U.S. economy as a whole or foreign economies in any locations where the Company or the Subsidiary has material operations or sales to the extent that they do not have a materially disproportionate effect on the Company and the Subsidiary taken as a whole; (iii) any adverse change, effect, occurrence, state of facts or developments resulting from or relating to compliance with the terms of, or the taking of any action required by, this Agreement; or (iv) any adverse change, effect, occurrence, state of facts or developments arising from or relating to any change in GAAP or any change in Applicable Law, or the interpretation thereof to the extent that they do not have a materially disproportionate effect on the Subsidiary taken as a whole.

"Mileage Funds" means the Money Market Mileage Fund series of the Mileage Funds Trust.

"Mileage Funds Trust" means the American Beacon Mileage Funds, a Massachusetts business trust.

"<u>Net Working Capital</u>" means an amount equal to (i) the value of the assets classified as current assets of the Company and the Subsidiary in accordance with GAAP and which are acquired by Buyer hereunder, minus (ii) the value of the liabilities classified as current liabilities of the Company and the Subsidiary in accordance with GAAP as set forth on <u>Schedule 3.3</u>. For this purpose, (i) current assets shall not include (A) deferred Tax assets or (B) the portion of the cash and cash equivalents balance equal to the total amount of the Company's payment obligations under <u>Section 8.1(k)</u>; and (ii) current liabilities shall not include the Company's payment obligations under <u>Section 8.1(k)</u>.

"Net Working Capital Adjustment Amount" has the meaning specified in Section 3.2.

"Net Working Capital Adjustment Report" has the meaning specified in Section 3.3(b).

"New Fund Advisory Agreements" has the meaning specified in the definition of "Fund Board Approvals".

"Notice" has the meaning specified in Section 7.6(a).

"Obligation" has the meaning specified in Section 8.13.

"Parent" shall mean Lighthouse Holdings Parent, Inc. a Delaware corporation and parent company of Buyer.

"<u>Party</u>" means each of Buyer, the Company and AMR.

"Patents" has the meaning specified in Section 5.12.

"Pension Plan" means any pension plan, as defined in Section 3(2) of ERISA.

"Permitted Encumbrances" means (a) liens for Taxes and other governmental charges and assessments which are not yet due and payable, (b) liens of landlords and liens of carriers, warehousemen, mechanics and materialmen and other like liens arising in the ordinary course of business for sums not yet due and payable, (c) Encumbrances identified on the Schedules to this Agreement, (d) source code escrow agreements for Software owned by the Company or the Subsidiary, and (e) other Encumbrances or imperfections on property which are not material in amount or do not materially detract from the value of or materially impair the existing use of the property affected by such lien or imperfection.

"Person" means any individual, corporation, partnership, limited liability company, joint venture, association, joint-stock company, trust, unincorporated organization or Governmental Body.

"Pharos" means Pharos Capital Group, LLC, a Delaware limited liability company and its Affiliates.

"Plans" has the meaning specified in Section 5.17(a).

"Pre-Closing Cash Dividends" means cash dividends paid or to be paid by the Company to AMR after the date hereof and prior to the Closing as permitted under Section 7.4(b)(v).

"Pre-Closing Dividends" means dividends paid or to be paid by the Company to AMR after the date hereof and prior to the Closing in the aggregate amount consistent with Section 7.4(b)(y).

"Pre-Closing Tax Period" means (i) any Tax period ending on or prior to the Closing Date and (ii) the portion of any Straddle Period ending on the Closing Date.

"Princeton Funds Distributors, Inc." means the Princeton Funds Distributors, Inc., a Delaware corporation.

"Properties" has the meaning specified in Section 5.10(b).

"Purchase Price Allocation" has the meaning specified in Section 8.10.

"Quantitative Master Series, LLC" means the Quantitative Master Series Trust, a Delaware limited liability company.

"Related Client" has the meaning specified in Section 5.9(a)(i).

"Revenue Run-Rate" means, as of any date for any Person, the aggregate annualized investment advisory, subadvisory, administration, shareholder servicing, 12b-1 or similar fees computed primarily by reference to assets under management that are payable to such Person or an Affiliate in respect of all Client accounts as to which such Person provides any of the foregoing services, determined by multiplying the Adjusted Assets Under Management for each such account as of such date by the applicable annual fee rate for such account as of such date. For purposes of this definition, the "applicable annual fee rate" for each account shall not include the effect of any performance-based fees or adjustments thereto or any extraordinary revenue items, and shall be reduced to take account of any then applicable fee waiver, expense reimbursement or rebate, or any reallowance of administration or shareholder servicing, 12b-1 or other such fees to any Person in connection with such account.

"SEC" means the United States Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended.

"Select Funds" means each of the Money Market Select Fund and the U.S. Government Money Market Fund series of the Select Funds Trust.

"Select Funds Trust" means the American Beacon Select Funds, a Massachusetts business trust.

"Seller" has the meaning specified in the first paragraph of this Agreement.

"Seller Group Member" means Seller and its Affiliates, their respective directors, officers, employees and agents, each person who controls any of them within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act, and their respective successors and permitted assigns.

"Seller's Accountant" has the meaning specified in Section 3.3(c).

"Seller's Medical Plans" has the meaning specified in Section 8.1(e).

"Seller Party" has the meaning specified in Section 8.4.

"Shares" has the meaning specified in the first "WHEREAS" clause of this Agreement.

"Shortfall" means the amount, if any, by which the amount of debt financing received by Buyer to pay the Closing Cash Purchase Price is less than \$180,000,000.

"<u>Software</u>" means computer software programs and software systems, including, without limitation, all databases, compilations, tool sets, compilers, higher level "proprietary" languages, related documentation and materials, whether in source code, object code or human readable form; <u>provided</u>, <u>however</u>, that Software does not include software that is available generally through consumer retail stores or distribution networks or is otherwise subject to "shrink-wrap" license agreements including, without limitation, any software pre-installed in the ordinary course of business as a standard part of hardware purchased by the Company or the Subsidiary.

"State Street Master Funds, a Massachusetts common law trust.

"Stock Consideration" means a number of shares of Class B Common Stock, \$0.01 par value per share, of Parent that represents, at the Closing, 10% of the issued and outstanding capital stock of Parent (excluding, for the purpose of this calculation, any shares of Class A Common Stock issued as consideration for additional equity to cover a Shortfall).

"Stockholders' Agreement, as may be amended from time to time, dated as of the Closing Date by and among Parent, Seller, TPG, Pharos and TCW.

"Straddle Period" means any Tax period beginning before and ending after the Closing Date.

"Subsidiary" means American Private Equity Management, L.L.C., a Delaware limited liability company, which serves as general partner of, but not investment manager to, APEP.

"Tax" (and, with correlative meaning, "Taxes") means all taxes, however denominated, including any federal, state, local or foreign income, gross receipts, property, sales, use, license, excise, franchise, employment, payroll, withholding, alternative or add-on minimum, ad valorem, value added, transfer or excise tax, or any other tax, custom, duty, governmental fee or other like assessment or charge of any kind whatsoever, together with any interest or penalty (including any penalties imposed in respect of any failure to comply with tax reporting obligations), imposed by any Governmental Body.

"Tax Return" means any return, report or similar statement required to be filed with respect to any Tax (including any attached schedules), including, without limitation, any information return, claim for refund, amended return or declaration of estimated Tax.

"TCW" means TCW/Crescent Mezzanine Partners, V L.P., a Delaware limited partnership, TCW/Mezzanine Partners VB, L.P., a Delaware limited partnership and TCW/Crescent Mezzanine Partners VC, L.P., a Delaware limited partnership and their Affiliates.

"Third Party Rights" has the meaning specified in Section 5.12(c).

"TPG" means TPG Partners V, L.P., a Delaware limited partnership and its Affiliates.

"Trademarks" has the meaning specified in Section 5.12.

"Trade Secrets" has the meaning specified in Section 5.12(g).

"Transition Services Agreement" means the Transition Services Agreement, in form and substance satisfactory to Buyer and Seller, dated as of the Closing Date by and among Buyer, Seller and the Company, pursuant to which Seller shall agree to provide the Company with certain transition services, including, without limitation, those set forth in Exhibit A. Seller shall price each of the services at the direct cost (excluding overhead and allocated costs) of such service to Seller. Annual costs for recurring services shall not exceed \$700,000, exclusive of employee benefit costs and migration costs. The cost for services not reflected on Exhibit A shall be determined consistent with similar services reflected on Exhibit A.

"Transfer Expenses" has the meaning specified in Section 8.8.

"Travel Privileges Agreement" means the Travel Privileges Agreement dated as of the Closing Date between AA and the Company in the form attached hereto as Exhibit G.

"Trusts" means the Master Trust, American Beacon Funds Trust, Mileage Funds Trust and Select Funds Trust.

"Welfare Plan" means any welfare plan, as defined in Section 3(1) of ERISA.

ARTICLE II

PURCHASE AND SALE

2.1 <u>Purchase and Sale of the Shares</u>.

On the terms and subject to the conditions set forth in this Agreement, AMR hereby agrees to sell, assign, transfer, convey and deliver to Buyer, and Buyer hereby agrees to purchase from AMR, the Shares, free and clear of all Encumbrances.

ARTICLE III

PURCHASE PRICE

3.1 Purchase Price.

The purchase price for the Shares shall be an amount in cash equal to \$450,000,000, reduced by Company Indebtedness as of the Closing Date (the "<u>Closing Cash Purchase</u> <u>Price</u>"), subject to adjustment following the Closing in accordance with <u>Sections 3.2</u>, and <u>3.3</u> (as so adjusted, the "<u>Final Cash Purchase Price</u>"), and the Stock Consideration. The Closing Cash Purchase Price shall be paid by Buyer pursuant to <u>Section 4.2</u> hereof.

3.2 Working Capital Adjustment to Purchase Price.

The Closing Cash Purchase Price shall be (i) reduced in the event and to the extent that Net Working Capital, as of the Closing, is less than the Base Net Working Capital Amount and (ii) increased in the event and to the extent that Net Working Capital, as of the Closing, is greater than the Base Net Working Capital Amount, the amount of any such deduction from, or increase in, the Closing Cash Purchase Price being referred to herein as the "<u>Net Working Capital Adjustment Amount</u>."

3.3 Determination of the Working Capital Adjustment Amount.

(a) Attached as <u>Schedule 3.3</u> is a spreadsheet illustrating the calculation of Net Working Capital. The parties agree that the Net Working Capital as of the Closing shall be calculated in a manner consistent with the calculation Net Working Capital as set forth on <u>Schedule 3.3</u>.

(b) On or before the later of January 31, 2009, or the date ninety (90) days following the Closing Date, Buyer shall prepare and deliver to Seller a report (the "<u>Net Working Capital Adjustment Report</u>") setting forth (i) the Net Working Capital and the Net Working Capital Adjustment Amount, if any, as of the Closing Date and (ii) Buyer's computation of each such item.

(c) Seller and/or a firm of independent public accountants designated by Seller ("Seller's Accountant") will be entitled to reasonable access during normal business hours to the relevant records and working papers of the accountants assisting in the preparation of the Net Working Capital Adjustment Report to aid in their review of the Net Working Capital Adjustment Report. The Net Working Capital Adjustment Report will be deemed to be accepted by and shall be conclusive for purposes of determining the Net Working Capital Adjustment Amount except to the extent, if any, that Seller or Seller's Accountant shall have delivered within thirty (30) days after the date on which the Net Working Capital Adjustment Report is delivered to Seller, a written notice to Buyer stating each and every item to which Seller takes exception, specifying in reasonable detail the nature and extent of any such exception (it being understood that any amounts not disputed shall be paid promptly). If a change proposed by Seller is disputed by Buyer, then Buyer and Seller shall negotiate in good faith to resolve such dispute. If, after a period of thirty (30) days following the date on which Seller gives Buyer notice of any such proposed change still remains disputed, then Seller and Buyer shall submit such dispute to Deloitte & Touche, LLP (the "Accounting Firm") for resolution. The Accounting Firm shall be nated solely on presentations by Buyer and Seller, and not by independent review, only those issues still in dispute with respect to calculation of the Net Working Capital Adjustment Amount. The decision of the Accounting Firm shall be in accordance with the provisions of this <u>Section 3.3</u> and shall be the exclusive remedy of the parties with respect to any disputes with respect to the calculation of the Net Working Capital Adjustment Amount. All of the fees and expenses of the Accounting Firm shall be borne equally by Buyer and Seller.

(d) The Net Working Capital Adjustment Amount, if any, shall be paid within fifteen (15) days following the applicable date of final determination of the Net Working Capital Adjustment Amount. Any Net Working Capital Adjustment Amount shall be paid as follows:

(A) if the Net Working Capital Adjustment Amount results in a reduction of the Closing Cash Purchase Price and thus is to be paid by Seller to Buyer, an amount in cash equal to the Net Working Capital Adjustment Amount shall be paid by wire transfer of immediately available funds to an account specified by Buyer in writing to Seller.

(B) if the Net Working Capital Adjustment Amount results in an increase in the Closing Cash Purchase Price and thus is to be paid by Buyer to Seller, an amount in cash equal to the Net Working Capital Adjustment Amount shall be paid by wire transfer of immediately available funds to an account or accounts specified by Seller in writing to Buyer.

(C) Any payment required to be made pursuant to this <u>Section 3.3(d)</u> shall be made together with interest thereon from the Closing Date to the date of payment at the rate of interest per annum equal to the prime rate in effect on the Closing Date as reported in <u>The Wall Street Journal</u>.

ARTICLE IV

CLOSING

4.1 Closing Date.

The Closing shall be consummated on the second Business Day after the conditions set forth in <u>Articles IX</u> and <u>X</u> have been satisfied or waived, at the offices of Kelly Hart & Hallman LLP, 201 Main Street, Suite 2400, Fort Worth, Texas or at such other time and place as shall be agreed upon by Buyer and Seller <u>provided</u>, <u>however</u>, that without the consent of Seller, the Closing shall not be consummated after December 31, 2008. The time and date on which the Closing is actually held is referred to herein as the "<u>Closing</u> <u>Date</u>".

4.2 Payment on the Closing Date.

Subject to fulfillment or waiver (where permissible) of the applicable conditions set forth in <u>Article IX</u>, at the Closing Buyer shall pay Seller an amount equal to the Closing Cash Purchase Price by wire transfer of immediately available funds to a bank account or accounts specified by Seller and shall issue to Seller the Stock Consideration.

4.3 Buyer's Additional Closing Date Deliveries.

Subject to fulfillment or waiver (where permissible) of the applicable conditions set forth in Article IX, at the Closing Buyer shall deliver to Seller all of the following:

- (a) Certificate of Incorporation of Buyer, certified as of a recent date by the Secretary of State of Delaware;
- (b) Certificate of good standing of Buyer, issued as of a recent date by the Secretary of State of Delaware;
- (c) The certificate contemplated by <u>Section 10.1</u>, duly executed by a duly authorized officer of Buyer;

(d) Duly executed original counterparts on behalf of Buyer and/or the Company of the Transition Services Agreement, Travel Privileges Agreement, and AAdvantage Participation Agreement;

- (e) Duly executed original counterparts on behalf of parent of the Stockholders' Agreement; and
- (f) Each of the other instruments or documents required to be delivered by Buyer hereunder.

4.4 Seller's Closing Date Deliveries.

Subject to fulfillment or waiver (where permissible) of the applicable conditions set forth in <u>Article X</u>, at the Closing Seller shall deliver to Buyer all of the following:

(a) Certificates of incorporation and formation of the Company and of the Subsidiary, each certified as of a recent date by the Secretary of State of Delaware;

- (b) Certificates of good standing of the Company and of the Subsidiary, each issued as of a recent date by the Secretary of State of Delaware;
- (c) Certificate of incorporation, as amended, of AMR, certified as of a recent date by the Secretary of State of Delaware;
- (d) A certificate of good standing of AMR, issued as of a recent date by the Secretary of State of Delaware;
- (e) The certificate contemplated by <u>Section 9.1</u>, duly executed by a duly authorized officer of Seller;
- (f) Certificates representing the Shares, duly endorsed in blank or accompanied by stock powers duly executed by AMR;
- (g) Duly executed original counterparts on behalf of AMR of the Transition Services Agreement, and the Stockholders' Agreement;
- (h) Duly executed original counterparts of the AAdvantage Participation Agreement and the Travel Privileges Agreement, executed on behalf of AA;

(i) Certificate of the secretary or an assistant secretary of AMR, dated the Closing Date, in form and substance reasonably satisfactory to Buyer, as to (i) the receipt of Client Consents or Fund approvals contemplated in <u>Sections 7.6</u>, <u>7.7</u> and <u>7.8</u>; and (ii) the amount of the Closing Revenue Run-Rate as contemplated in <u>Section 9.5</u>;

(j) Duly executed employment agreements between the Company and each of William F. Quinn and Michael Fields, on terms and conditions customary in the Business and reasonably acceptable to Buyer, Seller and such persons;

- (k) Duly executed letters of resignation of each of the directors of the Company and Subsidiary; and
- (1) Each of the other instruments or documents required to be delivered by the Company hereunder.

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF SELLER

As an inducement to Buyer to enter into this Agreement and to consummate the transactions contemplated hereby, Seller and the Company, jointly and severally, represent and warrant to Buyer as follows:

5.1 Organization of AMR.

AMR is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.

5.2 Organization; Capital Structure of the Company; Power and Authority.

(a) Each of the Company and the Subsidiary is a corporation or limited liability company, as the case may be, duly organized, validly existing and in good standing under the laws of the State of Delaware. Each of the Company and the Subsidiary is duly qualified to transact business and is in good standing in each jurisdiction listed on <u>Schedule 5.2</u> and in each other jurisdiction where the character of its properties owned or held under lease or the nature of its activities makes such qualifications necessary, except where the failure to be so qualified or in good standing would not reasonably be expected to have a Material Adverse Effect. Each of the Company and the Subsidiary has the corporate or limited liability company, as the case may be, power and authority to own or lease and operate its assets and to carry on its business in the manner that it was conducted immediately prior to the date of this Agreement.

(b) The authorized capital stock of the Company consists of 1,000 shares of Common Stock, par value \$1.00 per share, of which 1,000 shares are issued and outstanding. AMR owns beneficially and of record all the outstanding capital stock of the Company, free and clear of all Encumbrances. All such capital stock is duly authorized, validly issued and outstanding, fully paid and nonassessable, and free of preemptive rights. Except for this Agreement, there are no commitments to issue or sell any shares of capital stock or any securities or obligations convertible into or exercisable or exchangeable for, or giving any Person any right to acquire from AMR or the Company, any shares of capital stock of the Company, and no such securities or obligations are outstanding. There are no voting trusts, voting agreements or other similar agreements or understandings outstanding with respect to the Shares.

5.3 Subsidiaries and Investments.

Except for (i) ownership of capital stock or equity interests of the Subsidiary as set forth on <u>Schedule 5.3</u>, (ii) any investments not in excess of 1% in the aggregate of any class of capital stock or other equity interest of any Person which may be held in the investment portfolio of the Company, and (iii) ownership of capital stock or equity interests of APEP, the Company does not, directly or indirectly, own, of record or beneficially, or have any right to acquire any outstanding equity interests in any corporation, partnership, joint venture or other entity. The Company owns, beneficially and of record, all the outstanding membership interests of the Subsidiary, free and clear of all Encumbrances. All such membership interests are duly authorized, validly issued and outstanding, fully paid and nonassessable, and free of preemptive rights. The Subsidiary has no commitment to issue or sell any equity interests or any securities or obligations convertible into or exercisable or exchangeable for, or giving any Person any right to acquire from the Subsidiary, any equity interests and no such securities or obligations are outstanding. The sole business of the Subsidiary is to serve as general partner of APEP for which it receives a management and administration fee and is entitled to an incentive reallocation of certain profits.

5.4 Authority of Seller; Conflicts.

(a) Each of AMR and the Company has the corporate power and corporate authority to execute, deliver and perform this Agreement and each of the Company Ancillary Agreements to which it is a party. The execution, delivery and performance of this Agreement and the Company Ancillary Agreements to which it is a party by each of AMR and the Company have been duly authorized and approved by the board of directors of AMR and the Company and do not require any further authorization or consent of the Company or AMR or their respective stockholders. This Agreement has been duly authorized, executed and delivered by AMR and the Company and (assuming the valid authorization, execution and delivery of this Agreement by Buyer) is the legal, valid and binding obligation of AMR and the Company enforceable against each in accordance with its terms, and each of the Company Ancillary Agreements to which it is a party has been duly authorized by AMR and/or the Company, as applicable, and upon execution and delivery by AMR and/or the Company, as applicable, will be (assuming the valid authorization, execution and delivery by Buyer, where Buyer is a party, or the other party or parties thereto) a legal, valid and binding obligation of AMR and/or the Company, as applicable, enforceable against each in accordance with its terms, subject, in the case of the Agreement and each of the Company Ancillary Agreements, to bankruptcy, insolvency, reorganization, moratorium and similar laws of general application relating to or affecting creditors' rights and to general equity principles.

(b) Except as set forth in <u>Schedule 5.4(b)</u>, neither the execution and delivery of this Agreement or any of the Company Ancillary Agreements or the consummation of any of the transactions contemplated hereby or thereby nor compliance with or fulfillment of the terms, conditions and provisions hereof or thereof will:

(i) assuming the receipt of all necessary consents and approvals and the filing of all necessary documents as described in <u>Section 5.4(b)(ii)</u>, result in a breach of the terms, conditions or provisions of, or constitute a default, an event of default or an event creating rights of acceleration, termination or cancellation or a loss of rights under, or result in the creation or imposition of any Encumbrance upon any of the securities or any of the assets of the Company or the Subsidiary, under (1) the charter or by-laws or other applicable organizational documents of AMR, the Company or the Subsidiary, (2) any material contract, agreement, note, bond, instrument, mortgage, lease, license, franchise or financial obligation to which AMR, the Company or the Subsidiary is a party or any of their respective properties is subject or by which AMR, the Company or the Subsidiary is bound, (3) any Court Order to which AMR, the Company or the Subsidiary or any of their respective properties is subject or by which AMR, the Company or the Subsidiary is bound or (4) any Applicable Law affecting AMR, the Company or the Subsidiary or any of their respective properties, or

(ii) require the approval, consent, authorization or act of, or the making by AMR, the Company or any Subsidiary of any notice, declaration, filing, permit or registration with, any Person, except (1) in connection, or in compliance, with the provisions of the HSR Act, and (2) such approvals, consents, authorizations, declarations, filings or registrations the failure of which to be obtained or made would not reasonably be expected to materially impair the Company's or the Subsidiary's ability to conduct its business in the manner that it was conducted immediately prior to the date hereof, materially impair the ability of the Company to perform its obligations hereunder or prevent the consummation of any of the transactions contemplated hereby.

5.5 Financial Statements.

Schedule 5.5 contains (i) the unaudited consolidated balance sheets of the Company and the Subsidiary as of the Financial Statements Date, December 31, 2004, December 31, 2005 and December 31, 2006 and the unaudited consolidated statements of income of the Company and the Subsidiary for the years then ended, and (ii) the unaudited consolidated balance sheets of the Company and the Subsidiary as of March 31, 2007 and 2008 and the unaudited statements of income of the Company and the Subsidiary for the three-month periods then ended. Such balance sheets and statements of income have been prepared in conformity with GAAP (except that the financial statements do not contain footnotes), and such balance sheets and related statements of income present fairly in accordance with GAAP, in all material respects, the financial position and results of operations of the Company and the Subsidiary, as of their respective dates and for the respective periods covered thereby subject, in the case of the financial statements referred to in clause (ii), to normal year-end adjustments (provided that such adjustments are not material, either individually or in the aggregate).

5.6 **Operations Since Financial Statements Date**.

Except as set forth in <u>Schedule 5.6</u>, since the Financial Statements Date, there have been no changes in the business, assets, results of operations or condition (financial or other) of the Company and the Subsidiary which have had or would reasonably be expected to have a Material Adverse Effect. Except as set forth in <u>Schedule 5.6</u>, from the Financial Statements Date, the Company and the Subsidiary have conducted the Business only in the ordinary course and neither the Company nor the Subsidiary has taken any action that, if taken during the period from the date hereof through the Closing Date, would constitute a breach of <u>Section 7.4</u>; provided, that, for the purposes of this <u>Section 5.6</u> references to "the date hereof" in <u>Section 7.4</u> shall be deemed to refer to the Financial Statements Date.

5.7 <u>Taxes</u>.

Except as set forth on <u>Schedule 5.7</u>, (i) the Company and the Subsidiary have filed or participated in the timely filing of all material Tax Returns required to have been filed on or before the date hereof and have paid in a timely manner (or there has been timely paid on its behalf) all Taxes required to be paid by or in respect of it (including withholding Taxes required to be paid or withheld) and have established reserves in accordance with GAAP for any other Taxes in respect of Pre-Closing Tax Periods; (ii) neither the Company nor the Subsidiary has waived any statute of limitations in respect of Taxes of the Company or the Subsidiary, pursuant to a waiver currently in effect; (iii) no issues that have been raised in writing against the Company or the Subsidiary, or in respect of which the Company or the Subsidiary could be liable for Tax, by any taxing authority in connection with the Company or the Subsidiary are currently pending; and (iv) all deficiencies asserted in writing or assessments made in writing by any taxing authority have been fully satisfied by payment or fully withdrawn. Except for the tax sharing arrangement between the Company or Subsidiary could be liable for any Taxes. As of the Closing Date, the Company will have paid all amounts owed by it under the tax sharing arrangement with AMR and that arrangement will be terminated on or prior to the Closing Date.

5.8 Governmental Permits.

Except as set forth in <u>Schedule 5.8</u>, the Company and the Subsidiary own, hold or possess all licenses, franchises, permits, privileges, immunities, approvals and other authorizations from a Governmental Body that are necessary to entitle them to own or lease, operate and use their assets and to carry on and conduct the Business substantially as conducted immediately prior to the date of this Agreement (herein collectively called "<u>Governmental Permits</u>"), except for such Governmental Permits as to which the failure to so own, hold or possess would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Each of the Company and the Subsidiary has complied in all material respects with all terms and conditions of the Governmental Permits and, to the Knowledge of the Company, no condition exists that with notice or lapse of time or otherwise would constitute a default under or would result in non-compliance in any material respect with the terms of any such Governmental Permit. The Governmental Permits are in full force and effect and are not subject to any suspension, cancellation, modification, revocation or any proceedings or investigations related thereto, and, to the Knowledge of the Company, no such suspension, cancellation, modification, proceedings or investigation is threatened.

5.9 Assets Under Management.

(a) The aggregate Adjusted Assets Under Management by the Company and the Subsidiary as of the Base Date are accurately set forth in <u>Schedule 5.9</u>. Also set forth in <u>Schedule 5.9</u> is a list as of the Base Date of all Management Contracts, setting forth with respect to each such Management Contract:

(i) the name of the Client under such Management Contract, indicating (A) any such Client (other than any ERISA Client) that is AMR or an Affiliate of AMR, and (B) any ERISA Client that is, holds the "plan assets" of, or acts on behalf of, an employee benefit plan maintained by AMR or an Affiliate of AMR (any Person described in clause (A) or (B), a "<u>Related Client</u>");

(ii) the state (or, if such Client is not a U.S. citizen, the country) of which such Client is a citizen or resident (in the case of individuals) or domiciled (in the case of entities);

(iii) the amount of Adjusted Assets Under Management pursuant to such Management Contract at the Base Date, and the nature of the Investment Management Services provided (e.g., discretionary or non-discretionary, advisory or administrative, whether any of such assets are the subject of a sub-advisory arrangement and, if so, the identity of the sub-adviser, etc.); and

(iv) the fee schedule in effect with respect to such Management Contract (including identification of any applicable sub-components of such fees, e.g., investment management fees, fees for any other services, etc., as applicable), and a description of any fees payable by the underlying Client in connection with Investment Management Services (or other services) provided by the Company or the Subsidiary other than pursuant to such Management Contract.

Except as set forth in <u>Schedule 5.9</u>, there are no contracts, agreements, arrangements or understandings pursuant to which the Company or any Subsidiary has undertaken or agreed to cap, waive, offset, reimburse or otherwise reduce any or all fees or charges payable by or with respect to any of the Clients set forth in <u>Schedule 5.9</u> or pursuant to any of the contracts set forth in <u>Schedule 5.9</u>. Except as set forth in <u>Schedule 5.9</u>, the Company has not received written notice from any Client of the Company (or, in the case of any Clients that are collective investments vehicles, underlying investors therein, as applicable) of its intention to terminate or materially reduce its investment relationship with the Company or the Subsidiary, or adjust the fee schedule with respect to any Management Contract in a manner which would reduce the fees payable to the Company or the Subsidiary in connection with such Client relationship and, to the Knowledge of the Company, no client has any intention to terminate or materially reduce its investment

relationship with the Company or the Subsidiary, or adjust the fee schedule with respect to any Management Contract in a manner which would reduce the fees payable to the Company or the Subsidiary in connection with such Client relationship.

(b) Except as set forth in <u>Schedule 5.9</u>, neither the Company nor the Subsidiary has any Management Contract or other arrangement with respect to which amounts payable to the Company or the Subsidiary are based on performance or otherwise provide for compensation on the basis of a share of capital gains upon or capital appreciation of the funds (or any portion thereof) of any Client.

(c) Each Client to which the Company provides Investment Management Services that is (i) an employee benefit plan, as defined in Section 3(3) of ERISA, that is subject to Title I of ERISA, (ii) a person acting on behalf of such a plan or (iii) an entity whose assets include the assets of such a plan, within the meaning of ERISA and applicable regulations (an "<u>ERISA Client</u>") has been managed by the Company such that the Company in the exercise of management is in compliance in all material respects with the applicable requirements of ERISA. <u>Schedule 5.9</u> identifies each Client that is an ERISA Client. The Company is, and there has been no event or occurrence that has resulted or would reasonably be expected to result in the Company failing to continue to qualify as, a qualified professional asset manager within the meaning of Part V(a) of Prohibited Transaction Class Exemption 84-14, as amended. With the Company as the qualified professional asset manager, the condition of Part I(g) of Prohibited Transaction Class Exemption 84-14, as amended, is, and there has been no event or occurrence that has resulted or would reasonably be expected to result in such condition failing to be, satisfied.

(d) Each Client to which the Company provides Investment Management Services that is registered as an investment company under the Investment Company Act (including, in the case of any "series" investment company, each series thereof) and each Exempt Fund Client is so identified on <u>Schedule 5.9</u>. Other than the Funds and the Exempt Fund Clients, neither the Company nor the Subsidiary provides Investment Management Services to or through (i) any issuer or other Person that is an investment company (within the meaning of the Investment Company Act), (ii) any issuer or other Person that would be an investment company (within the meaning of the Investment Company Act), Section 3(c)(7), the final clause of Section 3(c)(3) (common trust funds) or the third or fourth clauses of Section 3(c)(11) (single or collective trusts for qualified plans) of the Investment Company Act, or (iii) any issuer or other Person that is or is required to be registered under the laws of the appropriate securities regulatory authority in the jurisdiction in which the issuer is domiciled (other than the United States or the states thereof), which is or holds itself out as engaged primarily in the business of investing, reinvesting or trading in securities.

(e) The Company has not received written notice from any Client of any material dispute between the Company or the Subsidiary and any Client of the Company or the Subsidiary and, to the Knowledge of the Company, no material dispute exists between the Company or the Subsidiary and any Client of the Company or the Subsidiary.

(f) Except as set forth in <u>Schedule 5.9</u>, no exemptive orders, "no-action" letters or similar exemptions or regulatory relief have been obtained, nor are any requests pending therefor, by or with respect to the Company, AMR or any Fund, or any officer, director, partner or employee of the Company or the Funds or employees of AA seconded to the Company, in connection with the business of the Company or the Funds, or by or with respect to any Client of the Company or the Subsidiary in connection with the provision of Investment Management Services to such Client.

(g) All performance information provided, presented or made available by the Company to any Client or potential Client has complied in all material respects with Applicable Law and the Company maintains all documentation necessary to form the basis for, demonstrate or recreate the calculation of the performance or rate of return of all accounts that are included in a composite (current and historical performance results) as required by Applicable Law.

5.10 Real Property.

(a) Neither the Company nor the Subsidiary owns any real property or holds any option to acquire any real property.

(b) <u>Schedule 5.10</u> sets forth a list of each lease or similar agreement under which the Company or the Subsidiary is lessee of, or holds or operates, any real property owned by any third Person (the "<u>Properties</u>"). Copies of all leases of Properties (the "<u>Leases</u>") have been made available to Buyer. Each Lease is valid, binding and enforceable in accordance with its terms, is in full force and effect, affords the Company or the Subsidiary, as applicable, peaceful and undisturbed possession of the subject matter thereof and has not been modified, amended, or altered, in writing or otherwise. There is not with respect to any such Leases any existing default or event that, with notice or lapse of time or both, would constitute a default or event of default on the part of the Company or the Subsidiary or, to the Knowledge of the Company, on the part of any other party thereto, except such defaults or events of default or other events that would not result in a Material Adverse Effect, and, to the Knowledge of the Company, there is no existing default by any other party to any Lease. The performance by the Company of this Agreement will not result in the termination of, or in any increase of any amounts payable under, any of the Leases. Except as set forth in <u>Schedule 5.10</u>, none of AMR, the Company or the Subsidiary has assigned any of the Leases and has not sublet all or any portion of its leased premises under any of the Leases.

5.11 Personal Property Leases.

<u>Schedule 5.11</u> contains as of the date of this Agreement a list of each lease or other agreement or right under which either the Company or the Subsidiary is lessee of, or holds or operates, any machinery, equipment, vehicle or other tangible personal property owned by a third Person, except those which are terminable by the Company or the Subsidiary without penalty on notice of sixty (60) days or less or which provide for annual rental payments of less than \$100,000. The Company or the Subsidiary, as applicable, holds a valid leasehold interest or other right in the properties leased or held pursuant to such leases, agreements or rights.

5.12 Intellectual Property.

<u>Schedule 5.12</u> contains a complete and accurate list of all registered and unregistered copyrights and applications therefor (including renewals) (the "<u>Copyrights</u>"), registered and unregistered trademarks and applications therefor (including renewals) (the "<u>Trademarks</u>"), registered and unregistered service marks and applications therefor (including renewals), patent disclosures (together with all re-issuances, continuations, continuations in part, revisions and extensions), patent rights (collectively, the "<u>Patents</u>"), inventions and discoveries (whether or not patentable and whether or not reduced to practice), trade names, trade name rights, trade dress and corporate names (together with all translations, derivations and combinations thereof) or other similar rights used or useable in the Business, including, specifically, the name (and any similar name), logos, trademarks, service marks, or other similar rights used in connection with the Business (whether or not fucures), and all licenses, sublicenses, agreements and permissions to or from third parties with respect to the foregoing or rights related thereto and, in each case, which is material to the Business (collectively, the "<u>Intellectual Property Rights</u>"). Such list includes, where applicable, the record owner, jurisdiction and registration and/or application number, and date issued (or filed) for each of the foregoing. As to the Trademarks, <u>Schedule 5.12</u> describes (i) those marks currently owned by AA that are being assigned to AMR. Except as set forth in <u>Schedule 5.12</u>:

(a) the Company exclusively owns or possesses adequate and enforceable rights to use, without payment to a third party, all of the Intellectual Property Rights and Trade Secrets (as defined below), that are used in or necessary for the conduct of the Business ("Company Intellectual Property Rights") free and clear of all Encumbrances;

(b) all Patents, Trademarks and Copyrights included in the Company Intellectual Property Rights which are issued by or registered with, as applicable, the U.S. Patent and Trademark Office, the U.S. Copyright Office or any similar office or agency anywhere in the world are currently in compliance in all material respects with formal legal requirements and are valid and enforceable;

(c) there are no pending, or, to the Knowledge of the Company, threatened claims against the Company alleging that the operation of the Business or any of the Company Intellectual Property Rights infringes, misappropriates or conflicts with the rights of others ("Third Party Rights");

(e) the Company has not received any communications alleging that the Company or the Subsidiary has violated or, by conducting its business, would infringe, misappropriate or violate any Third Party Rights or that any of the Company Intellectual Property Rights are invalid or unenforceable;

(f) to the Knowledge of the Company, no third party is violating, misappropriating or infringing any of the Company's Intellectual Property Rights;

(g) the Company has taken all reasonable security measures to protect the secrecy, confidentiality and value of any trade secrets it may own (the "<u>Trade Secrets</u>"). No material Trade Secrets have been disclosed by the Company to any person or entity other than employees or contractors performing services for the Company who had a need to know and use such Trade Secrets in the course of their employment or contract performance. The Company has the right to use, free and clear of claims of third parties, all of the Trade Secrets. To the Knowledge of the Company, no third party has asserted that the use by either the Company or the Subsidiary of any Trade Secrets misappropriates or violates any Third Party Right;

(h) To the Knowledge of the Company, any Software owned or used by the Company and the Subsidiary is free from any material uncorrected defects or programming or documentation errors, and the Company is not aware of the existence of any material defects or viruses that would cause an interruption to the operation of such Software;

(i) The Company and the Subsidiary have complied and are complying in all material respects with all Applicable Laws and contractual obligations and their own internal policies applicable to privacy, data security and data protection. The Company and the Subsidiary have implemented reasonable backup and disaster recovery technology consistent with normal industry practice. The Company and the Subsidiary have not experienced any losses or theft of data; and

(j) Except for the Intellectual Property Rights licensed to the Company pursuant to the AAdvantage Participation Agreement, Seller does not own or control any Company Intellectual Property Rights material to the operation of the business.

5.13 Assets.

(a) Except for assets disposed of in the ordinary course of business since the Financial Statements Date, the Company and the Subsidiary have valid title to each item of machinery, equipment and other tangible or intangible property reflected on the Financial Statements as owned by the Company and the Subsidiary, free and clear of all Encumbrances, except for Permitted Encumbrances. All items of machinery, equipment or other tangible personal property of the Company are usable in the ordinary course of business.

(b) The Company and the Subsidiary taken as a whole have ownership of or rights in, free and clear of Encumbrances (except Permitted Encumbrances), all of the assets, properties and rights (including all books and records), and have access to all services provided by the Seller or by contractors of the Seller to the Company as of the date hereof, used in or necessary for the conduct of their business in all material respects as currently conducted or reasonably anticipated to be conducted, including without limitation those assets set forth in <u>Schedule 5.13</u>.

5.14 No Violation, Litigation or Regulatory Action.

Except as set forth in <u>Schedule 5.14</u>:

(a) the Company and the Subsidiary have complied, in all material respects, with all Applicable Laws and Court Orders;

(b) there are no lawsuits, claims, suits, proceedings or investigations pending or, to the Knowledge of the Company, threatened against the Company or the Subsidiary or affecting the properties or assets of the Company or the Subsidiary, or, as to matters related to the Company or the Subsidiary, against any officer, director, stockholder or employee performing services for the Company or the Subsidiary in their respective capacities in such positions;

(c) there are no lawsuits, claims, suits, proceedings or investigations pending or, to the Knowledge of the Company, threatened relating to the termination of or limitation of the rights of the Company under or with respect to the Company's registration under the Advisers Act, the Company's ability to act as a qualified professional asset manager within the meaning of Part V(a) of Prohibited Transaction Class Exemption 84-14, as amended, or any similar or related rights under any registrations or qualifications with various states or other jurisdictions; and

(d) there are no lawsuits, claims, suits, proceedings or investigations pending or, to the Knowledge of the Company, threatened against the Company or the Subsidiary that question the legality of the transactions contemplated by this Agreement or any of the Company Ancillary Agreements or that are reasonably expected to materially impair the ability of the Company to perform its obligations hereunder or prevent the consummation of the transactions contemplated hereby.

5.15 <u>Contracts</u>. Except as set forth in <u>Schedule 5.15</u> or any other Schedule hereto, as of the date of this Agreement, neither the Company nor the Subsidiary is a party to or bound by:

(a) any Management Contract or any other contract for the provision of Investment Management Services or other similar services;

(b) any contract or agreement for the provision of services to Clients of the Company or the Subsidiary, other than Investment Management Services (e.g., tax preparation or similar services);

(c) other than Management Contracts, any contract or agreement providing for the indemnification by the Company or the Subsidiary of any Person with respect to liabilities, whether absolute, accrued, contingent or otherwise that would be material to the Business;

(d) any plan or contract providing for bonuses, pensions, options, stock (or other beneficial interest) purchases (or other securities or phantom equity purchases), incentive compensation, deferred compensation, retirement payments, profit sharing, severance or the like;

(e) any employment contract or contract for services which is not terminable at will by the Company or the Subsidiary without liability for any penalty or severance payment (excluding any liability or obligation imposed by statute (e.g., COBRA));

(f) any contract or agreement with any investment or research consultant, solicitor or sales agent, or otherwise with respect to the referral of business to the Company or the Subsidiary (including without limitation any agreement with respect to solicitation of prospective investors in any of the Funds);

(g) any license agreement (as licensor or licensee) (other than any such agreement the subject matter of which is off-the-shelf, commercially available software), including without limitation any source code agreements for Software owned by the Company or the Subsidiary;

(h) any contract for the purchase by the Company or the Subsidiary of services, supplies, components or equipment which involved the payment of more than \$50,000 in the year ended December 31, 2007;

(i) any loan agreements, promissory notes, indentures, bonds, security agreements, guarantees or obligations for Indebtedness or other instruments involving Indebtedness in an amount in excess of \$100,000;

(j) any partnership, joint venture or other similar agreement or arrangement;

(k) any agreement containing any covenant or provision prohibiting the Company or the Subsidiary or any employee who performs services for the Company from engaging in any line or type of business, competing with any Person or operating in any geographical area, including by means of a grant of exclusivity (except for such agreements which shall not apply to the Company or the Subsidiary or any such employee upon Closing);

(l) any contract or agreement to cap fees, share fees or other payments, share expenses, waive fees or to reimburse or assume any or all fees or expenses thereunder that would be material to the Business, taken as a whole;

(m) any contract or agreement (other than shareholder servicing agreements, the revenues and expenses of which are accurately reflected in the management projections shown to Buyer, consistent with the assumptions underlying the composition and amount of AUM contained therein, which are consistent with historical trends) that provides for aggregate annual future payments by the Company or the Subsidiary of more than \$50,000 or that has an unexpired term exceeding one year and that may not be canceled upon sixty (60) days' or less notice without any liability, penalty or premium, or that provides for the payment of royalties to a third party;

(n) any contract or agreement that was entered into by the Company or the Subsidiary with a stockholder, officer, director or employee of the Company or the Subsidiary or, to the Knowledge of the Company, any family member or Affiliate of any of the foregoing;

(o) any collective bargaining or similar agreement; or

(p) any contract or agreement relating to the redemption or purchase of capital stock or all or substantially all of the assets of the Company or equity interests of the Subsidiary or any acquisition, merger or similar agreement.

5.16 Status of Contracts.

Except as set forth in <u>Schedule 5.16</u> or in any other Schedule hereto, each of the leases, contracts, licenses and other agreements listed in <u>Schedules 5.9</u>, <u>5.10</u>, <u>5.11</u>, <u>5.12</u> and <u>5.15</u> (collectively, the "<u>Business Agreements</u>") is in full force and effect. Neither the Company nor the Subsidiary is in or, to the Knowledge of the Company, alleged to be in breach or default under any of the Business Agreements, and, to the Knowledge of the Company, no other party to any such Business Agreement is in or is alleged to be in breach or default in any material respect thereunder.

5.17 ERISA.

(a) Each Welfare Plan, Pension Plan, bonus, profit sharing, deferred compensation, incentive compensation, stock ownership, stock purchase, stock option, phantom stock or other equity-based or other employee benefit plan and each employment, retention, consulting, change in control, termination or severance plan, program, arrangement or agreement, maintained or established in connection with the Business, in which at least one employee performing services for the Company participates (whether or not subject to ERISA) is listed on <u>Schedule 5.17(a)</u> (the "<u>Plans</u>"), and the Company has made available to Buyer either a true and correct copy of each such Plan, all material amendments or modifications to any such Plan or a current summary plan description used in connection with such Plan. With respect to any Pension Plan maintained by AMR or the Company that is subject to Title IV of ERISA, there has not been a plan termination or any other event that would cause the Company or the Subsidiary to incur liability under Sections 4062, 4063, 4064, 4067 or 4069 of ERISA or to have a lien imposed on its assets under Section 4068 of ERISA. The Company has never maintained and has never been obligated to contribute to a multiemployer plan (as defined in Section 3(37) of ERISA).

(b) No event or condition has occurred that would reasonably be expected to give rise to a material liability of the Company as a result of the termination of a pension plan subject to Title IV of ERISA, any failure to contribute to any pension plan subject to Section 412 of the Code or the withdrawal or any other event with respect to a multiemployer plan or multiple employer plan subject to Title IV of ERISA.

(c) Except as set forth in <u>Schedule 5.17(c)</u>, with respect to each Welfare Plan and Pension Plan listed on <u>Schedule 5.17(a)</u>, (i) each such plan has been maintained and operated in compliance in all material respects with the applicable requirements of the Code and ERISA and the regulations issued thereunder and (ii) no litigation or asserted claims against the Company exist with respect to any such plan other than claims for benefits in the normal course of business.

(d) Any employee benefits other than those listed in <u>Schedule 5.17(a)</u> relating to the Business which are in effect on the Closing Date and as to which the Company has or may have in the future any liability (other than regular wages or salary), such as any fringe benefits described in Section 132 of the Code and any education assistance plans under Section 127 of the Code are listed in <u>Schedule 5.17(d)</u>, and any written document which exists with respect to any such employee benefit has been made available to Buyer by the Company.

5.18 Employee Relations and Agreements.

(a) <u>Schedule 5.18</u> contains a true and complete listing, as of a recent date, of all employees of American Airlines, Inc. ("<u>AA</u>") who are seconded to the Company and whose annual base salaris exceed \$100,000, their annual base salary and date of hire. Since the Financial Statements Date, except as disclosed on <u>Schedule 5.18</u> or as has occurred in the ordinary course of business and consistent as to timing and amount with past practices, AA has not: (i) increased the compensation payable or to become payable to or for the benefit of any of the employees seconded to the Company, (ii) provided any of the employees seconded to the Company with increased security or tenure of employment, (iii) increased the amount payable to any of the employees seconded to the Company upon the termination of such persons' employment, or (iv) increased, augmented or improved benefits granted to or for the benefit of any of the employees seconded to the Company under any bonus, incentive compensation (including any equity-based compensation), profit sharing, pension, retirement, deferred compensation, insurance or other direct or indirect benefit plan or arrangement. Neither the Company nor the Subsidiary has any employees.

(b) Except as set forth in <u>Schedule 5.18</u>, neither AMR, AA nor the Company is a party to or bound by any labor contract or collective bargaining agreement affecting employees seconded by AA to the Company.

(c) Except as set forth in <u>Schedule 5.18</u>, no union or similar organization represents employees of AA seconded to the Company and, to the Knowledge of the Company, no such organization is attempting to organize such employees.

(d) Except as set forth in <u>Schedule 5.18</u>, no director, officer or employee of AA seconded to the Company or the Subsidiary is a party to any Plan that entitles him or her to compensation or other consideration upon the acquisition by any Person of control of the Company or any Subsidiary or upon the closing of the transactions contemplated hereby.

(e) Neither the Company nor the Subsidiary is a party to, or otherwise bound by, any consent decree with, or citation by, any Governmental Body relating to employee or employment practices.

(f) The Company and the Subsidiary are in compliance with all Applicable Laws relating to employment, including, without limitation, Applicable Laws relating to discrimination, hours of work and the payment of wages or overtime wages and there are no complaints, lawsuits, or other proceedings pending or, to the Knowledge of the Company, threatened against the Company or the Subsidiary brought by or on behalf of any applicant for employment, any current or former employee, any current or former employee of AA seconded to the Company or any class of the foregoing, relating to any such Applicable Law or regulation, or alleging breach of any express or implied contract of employment, wrongful termination of employment, or alleging any other discriminatory, wrongful or tortuous conduct in connection with the employment relationship.

(g) Since December 31, 2007, no key employee of AA seconded to the Company has terminated, or given notice of his or her intent to terminate, employment.

5.19 Insurance.

AMR or the Company currently maintain the insurance policies and coverages set forth in <u>Schedule 5.19</u>. The insurance provided under such insurance policies is in such amounts, with such deductibles and against such risks and losses as are reasonable in respect of the operations of the Business. Such insurance policies are in full force and effect and all premiums due and payable thereon have been paid in full, and no written notice of cancellation or termination has been received with respect to any such insurance policy, which has not been replaced on substantially similar terms prior to the date of such cancellation. Except as set forth in <u>Schedule 5.19</u>, there are no material outstanding claims under any such insurance policy.

5.20 No Brokers.

Except for the fees and commissions set forth in <u>Schedule 5.20</u>, for which AMR is solely responsible, none of AMR, the Company or the Subsidiary nor any Person acting on behalf of any of them has paid or become obligated to pay any fee or commission to any broker, finder or intermediary for or on account of the transactions contemplated by this Agreement.

5.21 <u>Registration as Investment Adviser</u>.

(a) The Company is duly registered as an investment adviser under the Advisers Act and the rules and regulations thereunder and has completed notice filings for each state listed in <u>Schedule 5.21</u>, which jurisdictions are the only jurisdictions wherein the Company is required to make such filings.

(b) The Company has filed a Form ADV with the SEC in accordance with the Advisers Act, which Form at the time of filing was, and as amended and supplemented is, in effect pursuant to the requirements of the Advisers Act. The Company has heretofore made available to Buyer copies of its Form ADV as amended or supplemented. The Company and each of its investment adviser representatives (as such term is defined in Rule 203A-3(a) under the Advisers Act) have, and until the time of the Closing will have, all material Governmental Permits required in order for them to conduct the Business in the manner presently conducted. None of the Company, the Subsidiary or any of such representatives is subject to any material limitation imposed in connection with one or more of the Governmental Permits. Except as set forth on Schedule 5.21, none of the Company, the Subsidiary or, in connection with their service to the Company or the Subsidiary, any of their respective directors or officers or the employees of AA seconded to them, is registered or required to be registered as a broker or dealer, a commodity trading adviser, a commodity pool operator, a futures commission merchant, an introducing broker, a registered representative or associated person, a counseling officer, an insurance agent, a sales person or in any similar capacity with the SEC, the Commodity Futures Trading Commission, the National Futures Association, the Financial Industry Regulatory Authority, Inc. (or its predecessor, the National Association of Securities Dealers, Inc.), the securities commission of any state or any other self-regulatory body. Except as set forth on Schedule 5.9, no person, other than full-time employees of AA seconded to the Company, renders Investment Management Services to or on behalf of Clients or solicits Clients with respect to the provision of Investment Management Services by either the Company or the Subsidiary. Those officers and employees of the Company and the Subsidiary who are required to be licensed or registered for the activities conducted by them in respect of the Business are and at all times since January 1, 2003 have been duly licensed or registered in each state or jurisdiction in which and with each Governmental Body with whom such licensing or registration is so required. Each such registration or license is in full effect, except where the failure to be so licensed or registered, individually or in the aggregate, would not reasonably be expected to have an adverse impact on the Business in any material respect. None of these officers and employees is, or, since January 1, 2003, has been subject to any material disciplinary or other material regulatory compliance action or material complaint by a Governmental Body or Client.

(c) None of the Company, the Subsidiary or any person "associated" (as defined under both the Investment Company Act and the Advisers Act) with either the Company or the Subsidiary has been (i) convicted of any crime or is or has engaged in any conduct that would be a basis for (A) denial, suspension or revocation of registration of an investment adviser under Section 203(e) of the Advisers Act, or (B) ineligibility to serve as an investment adviser (or in any other capacity contemplated by the Investment Company Act) to a registered investment company pursuant to Section 9(a) or 9(b) of the Investment Company Act and, to the Knowledge of the Company, there is no proceeding or investigation that is reasonably likely to become a basis for any such ineligibility, disqualification, denial, suspension or revocation or (ii) subject to voluntary or involuntary petition under the federal bankruptcy laws or any state insolvency law or the appointment of a receiver, fiscal agent or similar officer by a court for his or her business or property or that of any partnership of which he or she was a general partner or any corporation or business association of which he or she was an executive officer.

(d) The Company has adopted a written policy regarding insider trading and a Code of Ethics which complies in all material respects with all applicable provisions of the Advisers Act (including without limitation with respect to insider trading and personal trading under Section 204A thereof and Rule 204A-1 thereunder) and the Investment Company Act (including without limitation Section 17(j) thereof), a copy of which has been made available to Buyer. All employees of AA seconded to the Company have executed acknowledgments that they are bound by the provisions of such Code of Ethics and insider trading policies. Each Fund has adopted a Code of Ethics which complies in all material respects with all applicable provisions of the Investment Company Act (including without limitation Section 17(j) thereof), copies of which have been made available to Buyer. Except as set forth on <u>Schedule 5.21</u>, during the past three (3) years, there have been no actions taken against any employee with regard to any violations of such Code of Ethics or insider trading policies that resulted in the termination of employment or the imposition of monetary penalties.

5.22 Filing of Reports; Maintenance of Records. Except for immaterial delays in filing,

the Company (a) has timely made all filings, including reports and other documents, required by the Advisers Act, (b) has timely made all disclosures and delivered all documents required by the Advisers Act to be delivered to its Clients and (c) has maintained all books and other records required by the Advisers Act. All filings, reports, disclosures and other documents were in compliance, in all material respects, with the Advisers Act at the time filed or made. The Company has heretofore made available to Buyer copies of all such filings, disclosure documents and books and records.

5.23 Fee Arrangements, Expenses and Custody of Funds.

The Company has complied in all material respects with the fee arrangement restrictions set forth in the Advisers Act. The Company does not maintain custody or constructive custody of any funds or securities of any Client except to the extent that it may be deemed to have custody by virtue of arrangements for the payment of fees from Client accounts.

5.24 Fund Matters.

(a) <u>Schedule 5.24(a)</u> describes each of the investment advisory agreements, distribution or underwriting contracts, plans adopted pursuant to Rule 12b-1 under the Investment Company Act, arrangements for the payment of service fees (as such term is defined in Rule 2830 of the NASD Conduct Rules), administrative services agreements and other agreements and contracts (other than agreements and contracts entered into by the Funds in the ordinary course of business in connection with the making of portfolio investments) (collectively, the "<u>Fund Agreements</u>") pertaining to any of the Funds. As to each Fund, there is in effect an investment advisory, distribution or underwriting agreement (as applicable), except that, in the case of the American Beacon Master Funds, there is no distribution or underwriting agreement. Each Fund Agreement pursuant to which the Company has received compensation with respect to its activities in connection with any of the Funds was duly approved in accordance with the applicable provisions of the Investment Company Act.

(b) Except as set forth in <u>Schedule 5.24(b)</u>, there are no special restrictions, consent judgments or SEC or judicial orders on or with regard to any of the Funds currently in effect.

(c) Each of the Trusts is duly organized, validly existing and in good standing in the jurisdiction in which it is organized and has all requisite trust power and authority to conduct its business in the manner and in the places where such business is currently conducted. Each Fund is and has been, since its inception, engaged solely in the business of an investment company.

(d) Each of the American Beacon Master Funds, American Beacon Funds, Mileage Funds and Select Funds is, and at all times required under the securities laws has been, a series of, respectively, the Master Trust, American Beacon Funds Trust, Mileage Funds Trust and Select Funds Trust. Each such Fund is a registered investment company under the Investment Company Act and is in compliance with Applicable Laws of the SEC and any other Governmental Body or self-regulatory body having jurisdiction over such Fund in all material respects. Since their initial offering, shares of each of the Funds have been duly qualified for sale under the securities laws of each jurisdiction in which they have been sold or offered for sale at such time or times during which such qualification was required. The shares of each of the Funds (excluding the American Beacon Master Funds) have been registered under the Securities Act during such period or periods for which such registration was required, the related registration statement has become effective under the Securities Act, no stop order suspending the effectiveness of any such registration statement nor any amendments thereto contained, at the time such registration statement or amendment became effective, any untrue statement of a material fact or omitted to state a material fact required to be stated therein or such shares under the Securities Act. Copies of the current registration statement of each of the Funds under the Investment Company Act and, with the exception of the American Beacon Master Funds, under the Securities Act have been made available to Buyer by the Company. All of the outstanding shares of beneficial interest of each Fund are duly authorized, validly issued, fully paid and non-assessable.

(e) Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, or subject the Funds to any regulatory sanctions or a valid Fund shareholder claim, each of the Funds' investments has been made in accordance with its investment policies and restrictions set forth in its registration statement in effect at the time the investments were made and has been held in accordance with its respective investment policies and restrictions, to the extent applicable and in effect at the time such investments were held.

(f) (i) Except for immaterial delays in filing, each of the Funds has timely filed (other than in respect of Taxes, which are the subject of separate representations and warranties set forth herein) all reports, registration statements and other documents, together with any amendments required to be made with respect thereto, that were required to be filed with any Governmental Body, including the SEC (the "<u>Fund Regulatory Documents</u>"), and has paid all fees and assessments due and payable in connection therewith, and (ii) as of their respective dates, each of the foregoing filings complied in all material respects with the requirements of the securities laws and the rules and regulations of the SEC promulgated thereunder applicable to such Fund Regulatory Documents, and none of the Fund Regulatory Documents or related prospectuses, as of their respective dates, contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The Company has made available to Buyer a copy of each Fund Regulatory Document filed with the SEC by a Fund after the date hereof and prior to the Closing.

(g) Except as set forth on <u>Schedule 5.24(g)</u>, (i) for all taxable years since 2002, each of the Funds (other than the American Beacon Master Fund, which is taxable as a partnership) has elected to be treated as, and has qualified as, a regulated investment company taxable under Subchapter M of Chapter 1 of the Code and under any similar provisions of state or local law in any jurisdictions in which such Fund filed, or is required to file, a Tax Return; (ii) each of the Funds has filed or participated in the timely filing of all material Tax Returns required to have been filed on or before the date hereof and has paid in a timely manner (or there has been timely paid on its behalf) all material Taxes required to have been paid by or in respect of it (including withholding Taxes required to be paid or withheld) or have established reserves in accordance with GAAP for any other Taxes in respect of Pre-Closing Tax Periods; (iii) no Fund has waived any statute of limitations in respect of Taxes of such Fund, pursuant to a waiver currently in effect; (iv) no issues that have been raised in writing by any taxing authority in connection with the Funds are currently pending; and (v) all deficiencies asserted in writing or assessments made in writing by any taxing authority have been fully satisfied by payment or fully withdrawn.

(h) The Company has made available to Buyer copies of the most recently available audited financial statements, prepared in accordance with GAAP, of each of the Funds, and unaudited financial statements, prepared in accordance with GAAP, of each of the Funds for the first six months of its most recent fiscal year if the ending date of such six-month period occurred more than sixty (60) days prior to the date of this Agreement (each hereinafter referred to as a "Fund Financial Statement"). To the Knowledge of the Company, each of the Fund Financial Statements is consistent with the books and records of the related Fund, and presents fairly the consolidated financial position of the related Fund in accordance with GAAP applied on a consistent basis (except as otherwise noted therein) at the respective date of such Fund Financial Statement and the results of operations and cash flows for the respective periods indicated. The Fund Financial Statements reflect and disclose all material changes in accounting principles and practices adopted by each of the Funds during the periods covered by each Fund Financial Statement.

(i) Except as currently disclosed in Item 11 to Part I of the Company's Form ADV, there is no litigation or legal action, suit, proceeding or investigation at law or in equity pending in any court or before or by any governmental agency or instrumentality, department, commission, board, bureau or agency, or before any arbitrator, by or against any of the Funds or the Trusts, or any officer or trustee thereof relating to the activities of the Funds or the Trusts, any disqualification of the Company under Section 9(a) of the Investment Company Act, or any event which would require the Company to give an affirmative response to any of the questions in Item 11 to Part I of its Form ADV (or any similar or successor form) and, to the Knowledge of the Company, no such litigation, legal action, suit, proceeding or investigation at law or in equity is threatened. There are no judgments, injunctions, orders or other judicial or administrative mandates outstanding against or affecting any of the Funds or the Trusts or any officer or trustee thereof relating to the activities of or affecting the Funds or the Trusts.

5.25 Exempt Fund Clients.

(a) To the Knowledge of the Company, (x) each Exempt Fund Client has been duly organized and is validly existing and in good standing under the laws of the jurisdiction of its organization and has all requisite corporate, partnership, limited liability company or similar power and authority, and possesses all rights, licenses, authorizations and approvals necessary to entitle it to use its name, to own, lease or otherwise hold its properties and assets and to carry on its business as it is now conducted, and is duly qualified, licensed or registered to do business in each jurisdiction where it is required to do so under Applicable Law (except where the failure to do so is not material to its business); and (y) each Exempt Fund Client is in compliance in all material respects with the terms and conditions of its constituent documents.

(b) All outstanding shares or units of each Exempt Fund Client have been issued and sold in substantial compliance with Applicable Law; and each Exempt Fund Client, since inception of operations, has been operated and is currently operating in compliance in all material respects with its respective investment objectives and policies and Applicable Law.

(c) None of the Exempt Fund Clients has been enjoined, indicted, convicted or made the subject of disciplinary proceedings, consent decrees or administrative orders on account of any violation of the rules or orders of any Governmental Body or self-regulatory body having jurisdiction over the Exempt Fund Client.

5.26 Information in Proxy Statements.

None of the information in any proxy statement described in <u>Section 7.7</u> hereof (other than any information to be supplied by or on behalf of Buyer or its Affiliates for inclusion therein) will, at the time any such proxy statement is mailed to the shareholders of the applicable fund, or at the time of the meeting of the shareholders of such fund, contain, any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

5.27 No Undisclosed Liabilities.

Neither the Company nor the Subsidiary has any material liabilities or obligations, whether accrued, absolute, contingent or otherwise, asserted or unasserted, required to be on a balance sheet prepared in accordance with GAAP except liabilities or obligations (a) stated or adequately reserved against in the unaudited consolidated balance sheet included in the Financial Statements, (b) set forth on <u>Schedule 5.27</u> or any other Schedule hereto, or (c) incurred since the Financial Statements Date in the ordinary course of business consistent (as to character, manner and amount) with past practice, which are completely and accurately reflected on the Company's books and records.

5.28 Transactions with Affiliates.

Except as set forth on <u>Schedule 5.28</u>, there are no loans, leases or other agreements or transactions involving in excess of \$50,000 annually between the Company, the Subsidiary or AMR, on the one hand, and any present or former director, officer or employee of the Company, the Subsidiary or AA seconded to the Company, or any member of such officer's, director's or employee's Immediate Family, or any person controlled by such officer, director or employee or his or her Immediate Family, on the other hand, other than payments for services rendered in the ordinary course. Any item listed on <u>Schedule 5.28</u> was entered into on an arm's-length basis and is on terms and conditions no less favorable to the Company than could be obtained from nonrelated persons.

5.29 Illegal Payments.

Neither the Company, the Subsidiary nor, to the Knowledge of the Company, any Person affiliated with the Company or the Subsidiary has ever offered, made or received on behalf of the Company or the Subsidiary any illegal payment or contribution of any kind, directly or indirectly, including, without limitation, payments, gifts or gratuities, to any person, entity, or United States or foreign national, state or local government officials, employees or agents or candidates therefor or other Persons.

5.30 Investment Intent; Accredited Investor.

(a) Seller is acquiring the shares issued as the Stock Consideration for its own account, for investment and without any intention of distributing or selling such shares in violation of the Securities Act or any applicable state securities law.

- (b) Seller qualifies as an "accredited investor," as such term is defined in Rule 501(a) promulgated pursuant to the Securities Act.
- (c) Seller qualifies as a "qualified purchaser" within the meaning of Sections 3(c)(7) and 2(a)(51) of the Investment Company Act and the related rules thereunder.

(d) Seller understands that the shares issued as the Stock Consideration have not been registered under the Securities Act. Seller acknowledges that such shares may not be transferred, sold, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act and any other provision of applicable state securities Laws or pursuant to an applicable exemption therefrom.

ARTICLE VI

REPRESENTATIONS AND WARRANTIES OF BUYER

As an inducement to Seller and the Company to enter into this Agreement and to consummate the transactions contemplated hereby, Buyer hereby represents and warrants to Seller and the Company as follows:

6.1 Organization of Buyer.

Buyer is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. Buyer has the corporate power and authority to own or lease and operate its assets and to carry on its businesses in the manner that they were conducted immediately prior to the date of this Agreement.

6.2 Authority of Buyer; Conflicts.

(a) Buyer has the corporate or other power and authority to execute, deliver and perform this Agreement and each of the Buyer Ancillary Agreements to which it is a party. The execution, delivery and performance of this Agreement and the Buyer Ancillary Agreements by Buyer have been duly authorized and approved by Buyer's board of directors and do not require any further authorization or consent of Buyer or its members. This Agreement has been duly authorized, executed and delivered by Buyer and (assuming the valid authorization, execution and delivery of this Agreement by AMR and the Company) is the legal, valid and binding agreement of Buyer enforceable in accordance with its terms, and each of the Buyer Ancillary Agreements has been duly authorized by Buyer and upon execution and delivery by AMR and/or the Company, where AMR and/or the Company is a party, or the other party or parties thereto) a legal, valid and binding obligation of Buyer enforceable in accordance with its terms, subject, in the case of the Agreement and each of the Buyer Ancillary Agreements, to bankruptcy, insolvency, reorganization, moratorium and similar laws of general application relating to or affecting creditors' rights and to general equity principles.

(b) Neither the execution and delivery of this Agreement or any of the Buyer Ancillary Agreements or the consummation of any of the transactions contemplated hereby or thereby nor compliance with or fulfillment of the terms, conditions and provisions hereof or thereof will:

(i) assuming the receipt of all necessary consents and approvals and the filing of all necessary documents as described in <u>Section 6.2(b)(ii)</u>, result in a breach of the terms, conditions or provisions of, or constitute a default, an event of default or an event creating rights of acceleration, termination or cancellation or a loss of rights under (1) the certificate of formation or operating agreement or other similar organizational documents of Buyer, (2) any material contract, agreement, note, bond, instrument, mortgage, lease, license, franchise or financial obligation to which Buyer is a party or any of their respective properties is subject or by which Buyer is bound, (3) any Court Order to which Buyer is a party or subject or (4) any Applicable Law affecting Buyer, or

(ii) require the approval, consent, authorization or act of, or the making by Buyer of any declaration, filing or registration with, any Person, except (1) in connection, or in compliance, with the provisions of the HSR Act and (2) such approvals, consents, authorizations, declarations, filings or registrations the failure of which to be obtained or made would not materially impair the ability of Buyer to perform its obligations hereunder or prevent the consummation of any of the transactions contemplated hereby.

6.3 <u>Capital Structure of Parent</u>. At the Closing, the shares of Class B Common Stock issued to AMR as the Stock Consideration pursuant to Section 3.1 shall constitute 100% of the issued and outstanding Class B Common Shares. As of the Closing, the Class B Common Stock shall represent 10% of the issued and outstanding capital stock of Parent; provided, however, that in the event and to the extent that there is a Shortfall, additional shares of Class A Common Stock shall be issued at the Closing, at the same purchase price per share of Class A Common Stock as originally paid by TPG and Pharos, to make up such Shortfall and no additional shares of Class B Common Stock shall be issued as Stock Consideration as a consequence of the issuance of additional shares of Class A Common Stock issued in connection with covering the Shortfall. All issued and outstanding Class B Common Stock issued to AMR as the Stock Consideration shall be duly authorized, validly issued and outstanding, fully paid and nonassessable, and free of preemptive rights. Immediately prior to the Closing, Buyer shall own beneficially and of record the shares of Class B Common Stock constituting the Stock Consideration, free and clear of all Encumbrances, and upon issuance of the Stock Consideration to AMR pursuant to Section 3.1, AMR shall receive such shares free and clear of all Encumbrances. At the Closing, except as provided in this Agreement or the Stockholders' Agreement. As of the Closing, Parent will own 100% of the outstanding capital stock of Buyer, free and clear of all Encumbrances. At the Closing, except as provided in this Agreement the Subscription Agreement by and between Parent and TCW, neither Parent nor Buyer is a

party to, or bound by, any agreement, arrangement or commitments to issue or sell any shares of capital stock or any securities or obligations convertible into or exercisable or exchangeable for, or giving any Person any right to acquire from Parent or Buyer, any shares of capital stock of Parent or Buyer, and no such securities or obligations are outstanding, other than the options issued at the Closing to the Company's management.

6.4 No Violation, Litigation or Regulatory Action.

Except as set forth in Schedule 6.3:

(a) there are no lawsuits, claims, suits, proceedings or investigations pending or, to the knowledge of Buyer, threatened against Buyer, which are reasonably expected to materially impair the ability of Buyer to perform its obligations hereunder or prevent the consummation of any of the transactions contemplated hereby; and

(b) there is no action, suit or proceeding pending or, to the knowledge of Buyer, threatened that questions the legality of the transactions contemplated by this Agreement or any of the Buyer Ancillary Agreements.

6.5 Financing.

Buyer will have sufficient funds available at Closing for it to pay the Closing Cash Purchase Price.

6.6 No Brokers.

Except as set forth in <u>Schedule 6.6</u>, neither Buyer nor any Person acting on its behalf has paid or become obligated to pay any fee or commission to any broker, finder or intermediary for or on account of the transactions contemplated by this Agreement.

6.7 Ineligible Persons.

Neither Buyer nor any "affiliated person" (as defined in the Investment Company Act) thereof, as applicable, is ineligible pursuant to Section 9(a) or 9(b) of the Investment Company Act to serve as an investment adviser (or in any other capacity contemplated by the Investment Company Act) to a registered investment company. Neither Buyer nor any "person associated with" Buyer (as defined in the Advisers Act), as applicable, is ineligible pursuant to Section 203 of the Advisers Act to serve as an investment adviser or as an associated person to a registered investment adviser or has failed to disclose as required any act enumerated in Rule 206(4)-4(b) under the Advisers Act.

6.8 Facts Affecting Regulatory Approvals.

To the knowledge of Buyer, there is no fact, event or condition applicable to Buyer which would reasonably be expected to adversely affect the likelihood of Buyer securing the requisite approvals or consents of any Governmental Body to the transactions contemplated by this Agreement.

6.9 Section 15 of the Investment Company Act.

Neither Buyer nor any of its Affiliates has any express or implied understanding or arrangement which would impose an unfair burden on any of the Funds or would in any way cause or result in a failure of the conditions contained in Section 15(f) of the Investment Company Act with respect to the transactions contemplated hereby.

6.10 Information in Proxy Statements.

None of the information to be supplied in writing by or on behalf of Buyer or its Affiliates expressly for inclusion in any proxy statement described in <u>Section 7.7</u> hereof will, at the time any such proxy statement is mailed to the shareholders of the applicable fund, or at the time of the meeting of the shareholders of such fund, contain, any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

6.11 <u>Investment Representations</u>. Buyer is acquiring the Shares for its own account, for investment and without any intention of distributing such Shares in violation of the Securities Act of 1933 or any applicable state securities law.

6.12 <u>No Currently Expected Claims</u>. Based on Buyer's current analysis of facts revealed to Buyer by means of due diligence materials, disclosure schedules or other information made available to Buyer by Seller and the Company, (i) Buyer is not aware of any breach of any representation made by Seller in <u>Section 5</u> and (ii) currently does not expect to make a claim for a breach of any representation made in <u>Section 5</u>. This representation is based upon Buyer's current knowledge of facts revealed to Buyer by means of due diligence materials, disclosure schedules or otherwise, and does not foreclose Buyer from making any claim of breach on any basis after the Closing.

ARTICLE VII

ACTION PRIOR TO THE CLOSING DATE

The respective Parties hereto covenant and agree to take the following actions between the date hereof and the Closing Date:

7.1 Access to Information.

The Company shall afford to the officers, employees and authorized representatives of Buyer (including, without limitation, independent public accountants and attorneys) reasonable access during normal business hours, upon reasonable advance notice, to the offices, properties, and business and financial records (including computer files, retrieval programs and similar documentation) of the Company, the Subsidiary and the Funds and the employees of AA seconded to the Company, to the extent Buyer shall reasonably deem necessary or desirable, and shall furnish to Buyer or its authorized representatives such additional information concerning the Company, the Subsidiary, the Clients and the Funds as shall be reasonably requested; <u>provided</u>, <u>however</u>, that the Company shall not be required to violate any obligation of confidentiality to which the Company, the Subsidiary or the Funds is subject or to waive any privilege which any of them may possess in discharging their obligations pursuant to this <u>Section 7.1</u>. Buyer agrees that such investigation shall be conducted in such a manner as not to interfere unreasonably with the operations of the Company, the Subsidiary, AMR or the Funds. Notwithstanding the foregoing, the obligations of the Company pursuant to this <u>Section 7.1</u> (other than with respect to the delivery of requested financial information) shall be subject to the right of the Company to determine, in its reasonable discretion, the appropriate timing of the disclosure of information it deems proprietary commercial information or privileged information.

7.2 Notifications.

(a) Each of Seller and Buyer shall promptly notify the other of any action, suit or proceeding that shall be instituted or threatened against such party to restrain, prohibit or otherwise challenge the legality of any transaction contemplated by this Agreement. Each Party hereto shall promptly notify the other of any lawsuit, claim, proceeding or investigation that may be threatened, brought, asserted or commenced against the Subsidiary, the Company or Buyer, as the case may be, that would have been listed in <u>Schedule 5.14</u> or <u>Schedule 6.3</u>, as the case may be, if such lawsuit, claim, proceeding or investigation had arisen prior to the date hereof.

(b) Each of Seller and Buyer shall promptly notify the other upon obtaining knowledge of (a) the occurrence or failure to occur of any event which occurrence or failure would be likely to cause any representation or warranty contained in this Agreement to be untrue or inaccurate in a manner reasonably likely to result in the failure of a condition set forth in <u>Article IX</u> or <u>Article X</u> and (b) any failure of such Party to comply with or satisfy any covenant or agreement to be complied with or satisfied by it under this Agreement in a manner reasonably likely to result in the failure of a condition set forth in <u>Article IX</u> or <u>Article X</u> and (b) any failure of a condition set forth in <u>Article IX</u> or <u>Article X</u>.

(c) No notice under this Section 7.2 shall be deemed to have modified any representation and/or warranty, cured any breach of covenant or otherwise affected any of the rights or remedies of the Parties to this Agreement.

7.3 Consents of Third Parties; Governmental Approvals.

(a) The Company and Buyer will act diligently and reasonably in attempting to secure, before the Closing Date, the consent, approval or waiver, in form and substance reasonably satisfactory to the other Party, required to be obtained from any party (other than a Governmental Body) to consummate the transactions contemplated by this Agreement; provided, however, that such action shall not include any requirement of the Company or any of its Affiliates (including the Company and the Subsidiary) to commence or participate in any litigation, or offer or grant any accommodation (financial or other) to any third party that it is not contractually obligated to offer or grant.

(b) During the period prior to the Closing Date, Buyer shall act diligently and reasonably, and the Company and Seller, upon the request of Buyer, shall use commercially reasonable efforts to cooperate with Buyer, in attempting to secure any consents and approvals of any Governmental Body required to be obtained by Buyer in order to permit the consummation of the transactions contemplated by this Agreement.

(c) Upon the terms and subject to the conditions set forth in this Agreement, each of the Parties agrees to use commercially reasonable efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other Party in doing all things necessary, proper or advisable to consummate, in the most expeditious manner practicable, the transactions contemplated by this Agreement, including but not limited to: (i) obtaining all necessary actions or non-actions, waivers, consents and approvals from all Governmental Bodies and making all necessary registrations and filings (including filings with Governmental Bodies) and taking all steps as may be necessary to obtain an approval or waiver from, or to avoid an action or proceeding by, any Governmental Body (including those in connection with the HSR Act), (ii) defending any lawsuits or other legal proceedings, whether judicial or administrative, challenging this Agreement or the consummation of the transactions contemplated hereby, including seeking to have any stay or temporary restraining order entered into by any court or other Governmental Body vacated or reversed, (iii) in the case of Buyer, promptly, if required by any Governmental Body in order to consummate the transactions contemplated hereby, taking all steps and making all undertakings to secure antitrust clearance (provided that in no event shall Buyer be required to effect the sale or other disposition of material properties of Buyer or its subsidiaries), (iv) keeping the other Party informed in all material respects of any material communication received by such Party from, or given by such Party to, any Governmental Body and of any material communication received or given in connection with any proceeding by a private party relating to the transactions contemplated by this Agreement, in each case regarding any of the transactions contemplated hereby, (v) permitting the other Party to review any material communication delivered to, and consulting with the other Party in advance of any meeting or conference with, any Governmental Body relating to the transactions contemplated by this Agreement or in connection with any proceeding by a private party, and giving the other Party the opportunity to attend and participate in such meetings and conferences (to the extent permitted by such Governmental Body or private party), (vi) obtaining of all necessary consents, approvals or waivers from third parties (other than ERISA Clients), including but not limited to those referred to in Sections 7.6, 7.7 and 7.8, and (vii) executing and delivering any additional instruments necessary to consummate the transactions contemplated by this Agreement. No Party to this Agreement shall consent to any voluntary delay of the consummation of the transactions contemplated hereby at the behest of any Governmental Body without the consent of the other Party to this Agreement, which consent shall not be unreasonably withheld.

7.4 **Operations Prior to the Closing Date**.

(a) From the date hereof to the Closing Date, Seller shall use commercially reasonable efforts to cause the Company and the Subsidiary to, and the Company shall and shall cause the Subsidiary to, operate and carry on the Business (and will use commercially reasonable efforts to cause the Funds and the Exempt Fund Clients to conduct their business) in the ordinary course and substantially as operated immediately prior to the date of this Agreement. Consistent with the foregoing, the Company and the Subsidiary shall use commercially reasonable efforts contractors, licensors, employees, Clients, investors, distributors or others having business relations with the Company or the Subsidiary.

(b) Notwithstanding Section 7.4(a), except as set forth on Schedule 7.4(b), except as contemplated by this Agreement or except with the express written approval of Buyer (which Buyer agrees shall not be unreasonably withheld or delayed), the Company and the Subsidiary shall not and, where applicable, AMR shall cause AA to not:

(i) make any material change in the Business or its operations, except such changes as may be required to comply with any Applicable Law;

(ii) make any capital expenditure or enter into any contract or commitment therefor in excess of \$50,000 (in the aggregate for both the Company and the Subsidiary);

(iii) enter into any contract for the purchase of real property or exercise any option to extend a lease listed in <u>Schedule 5.10</u>;

(iv) create, incur or assume, or agree to create, incur or assume, any Indebtedness or enter into, as lessee, any capitalized lease obligations (as defined in Statement of Financial Accounting Standards No. 13) other than money borrowed or advances from any of its Affiliates in the ordinary course of business;

(v) declare, set aside or make, or agree to make, any dividend or other payment or distribution of cash or assets to AMR or any of its Affiliates (other than the Company or the Subsidiary) or any other Person, other than Pre-Closing Cash Dividends that would not cause the cash and cash equivalents balance of the Company as of the Closing to be less than an amount equal to the sum of (A) \$500,000; and (B) the total amount of the payments that the Company will be obligated to make under <u>Section 8.1(k)</u>;

(vi) institute any new, or permit any increase in any existing, profit-sharing, bonus, incentive compensation (including any equity-based compensation), deferred compensation, insurance, pension, retirement, medical, hospital, disability, welfare or other employee benefit plan with respect to the employees of AA seconded to it, other than as required by any such plan or Applicable Law;

(vii) make any change in the compensation of the employees of AA seconded to it, other than normal increases in annual salary to Persons who are not officers or directors in the ordinary course of business consistent with past practices and that, in the aggregate, do not result in a material increase in benefits or compensation expense of the Company;

(viii) terminate any employee of AA seconded to the Company, except as a direct result of such employee's (i) willful failure to perform the duties or responsibilities of his or her employment, (ii) engaging in serious misconduct, or (iii) being convicted of or entering a plea of guilty to any crime;

(ix) make any change in the accounting policies applied in the preparation of the Financial Statements, unless such change is required by GAAP;

(x) make advertising, marketing or similar types of expenditures other than in the ordinary course of business and in amounts consistent with the practices of the Company during the twelve-month period immediately preceding the date of this Agreement (and the Company will use commercially reasonable efforts to prevent any Fund from making such expenditures);

(xi) make any change in the charter or by-laws of the Company or the Subsidiary or issue, sell, transfer, pledge, dispose of or encumber any capital stock or other equity interests (or securities exchangeable, convertible or exercisable for capital stock or other equity interests);

(xii) modify the terms of any Indebtedness or other liability, other than modifications of short term debt in the ordinary course of business, or assume or guarantee the obligations of any other Person, or pay, discharge or satisfy any claims, liabilities or obligations (absolute, accrued, contingent or otherwise), other than the payment, discharge or satisfaction of liabilities in the ordinary course of business;

(xiii) sell, lease (as lessor), transfer or otherwise dispose of (including any transfers to any Affiliate of the Company), or mortgage or pledge, or impose or suffer to be imposed any Encumbrance on, a material amount of the assets reflected on the consolidated balance sheet of the Company and the Subsidiary as of the Financial Statements Date or any assets acquired by the Company or the Subsidiary after the Financial Statements Date, except for Permitted Encumbrances;

(xiv) cancel any debts owed to or claims held by the Company or the Subsidiary (including the settlement of any claims or litigation) that were material to the Company and the Subsidiary taken as a whole other than in the ordinary course of the business consistent with past practices;

(xv) pay or agree to pay any pension, retirement allowance or other employee benefit not required by any of the plans or programs described in <u>Schedules 5.17(a)</u> or <u>5.17(c)</u> to any director, officer, employee, consultant or employee of AA seconded to the Company, whether past or present;

(xvi) enter into any new, or amend, terminate or renew any existing, employment, consulting, salary continuation, severance or termination agreement with any director, officer, employee or consultant or any employee of AA seconded to the Company;

(xvii) forgive any loans to any employee of AA seconded to the Company or any member of such employee's Immediate Family, or any person controlled by such employee or his or her Immediate Family;

(xviii) enter into any new loans, leases or other agreements or transactions which, if entered into as of the date hereof, would be required to be listed on <u>Schedule 5.28</u>, or, except as and to the extent contemplated by <u>Section 9.6</u> and <u>Section 10.6</u>, materially amend any such item listed on <u>Schedule 5.28</u>;

(xix) except as may be required to comply with Applicable Law, become obligated under any new Pension Plan, Welfare Plan or other employee benefit plan, which was not in existence on the date hereof, or amend any such plan in existence on the date hereof if such amendment would have the effect of materially enhancing any benefits thereunder; or

(xx) (A) make, change or revoke any material election in respect of Taxes; (B) prepare any Tax Returns in a manner which is not consistent in all material respects with the past practice of the Company and the Subsidiary with respect to the treatment of items on such Tax Returns; (C) file any material amendment to a Tax Return that will or may increase the Tax liability of the Company or the Subsidiary before or after the Closing; or (D) settle any claim or assessment in respect of Taxes, in each case, which would materially and adversely affect the Company and the Subsidiary, taken as a whole, before or after the Closing Date.

7.5 <u>Antitrust Law Compliance</u>.

As promptly as practicable after the date hereof, if determined to be required by Applicable Law, Seller and the Company shall file with the Federal Trade Commission and the Antitrust Division of the Department of Justice the notifications and other information required to be filed under the HSR Act, or any rules and regulations promulgated thereunder, with respect to the transactions contemplated hereby. Each Party warrants that all such filings by it will be, as of the date filed, true and accurate in all material respects and in material compliance with the requirements of the HSR Act and any such rules and regulations. Each of Buyer and Seller agrees to make available to the other such information as each of them may reasonably request relative to its business, assets and property as may be required of each of them to file any additional information requested by such agencies under the HSR Act and any such rules and regulations.

7.6 Client Consents.

(a) As soon as reasonably practicable following the date hereof, the Company and Buyer shall send notices substantially in the form of <u>Exhibit D</u> hereto (each, a "<u>Notice</u>") to each of the Company's Clients (excluding the Funds, the Exempt Fund Clients and the ERISA Clients), (i) informing each such Client of the transactions contemplated by this Agreement which would result in a change in control of the Company and (ii) requesting the consent or approval of the change in control of the Company if Client Consent to such change in control is required under the respective Management Contract.

(b) As soon as reasonably practicable following the date hereof, the Company and Buyer shall send notices substantially in the form of Exhibit <u>E</u> hereto to each of the Company's ERISA Clients (i) informing each such Client of the transactions contemplated by this Agreement which would result in a change in control of the Company and (ii) requesting that such Client enter into a new Management Contract, in form and substance satisfactory to Buyer, (subject to approval thereof by the independent fiduciary of such Clients).

7.7 Fund Approvals.

Each of Buyer and the Company shall, as promptly as practicable after the date hereof, use reasonable best efforts to assure that prior to the Closing Date, (i) the board of trustees of each of the Master Trust, American Beacon Funds Trust, Mileage Funds Trust and the Select Funds Trust, acting on behalf of, respectively, the American Beacon Master Funds, American Beacon Funds, Mileage Funds and Select Funds, including a majority of the trustees who are not Interested Persons of the Company or Buyer or any of their respective Affiliates, or, as applicable, any of the subadvisers referred to below, or any of their respective Affiliates, vote in person (A) for the Fund Board Approvals, (B) for a new sub-advisory agreement, if there is in effect as at the date hereof an existing sub-advisory agreement between the Company, as adviser, and a sub-adviser with respect to such Fund, to be in effect from and after the Closing with the Company on terms substantially identical in all material respects (and identical as to fees) to such existing subadvisory agreement, (C) for a new service or distribution agreement, if the Company is a party to any thereof with any such Fund, to be in effect from and after the Closing with the Company on terms substantially identical in all material respects (and identical as to fees) to such existing service or distribution agreement (to the extent required under the existing agreement or Applicable Law), (D) for new master-feeder participation agreements for the American Beacon Small Cap Index Fund, American Beacon International Equity Index Fund and the American Beacon S&P 500 Index Fund to be in effect from and after the Closing with the Company on terms substantially identical in all material respects (and identical as to fees) to the master-feeder participation agreements for the American Beacon Small Cap Index Fund, the American Beacon International Equity Index Fund and the American Beacon S&P 500 Index Fund existing with respect to such Funds, the Company and Quantitative Master Series, LLC, Princeton Funds Distributors, Inc., and State Street Master Funds, (E) to approve the preparation of proxy statements for meetings of the shareholders of the respective Funds at which the Fund Shareholder Approvals will be recommended and sought, and (F) to approve such other matters as may be required by the Investment Company Act, including those specifically relating to Section 15(f) thereof relating to the sale of investment advisers, and (ii) the Fund Shareholder Approvals are obtained at shareholder meetings of the Funds. Buyer and Seller shall each bear fifty percent (50%) of the costs, fees and expenses incurred by the Funds in connection with the proxy statements described above, including fees of accountants and attorneys, except to the extent such costs, fees and expenses are borne by the Funds in accordance with Applicable Law.

7.8 Approval of Exempt Fund Clients.

As soon as reasonably practicable following the date hereof, the Company or the Subsidiary, as appropriate, shall send a notice substantially in the form of <u>Exhibit F</u> hereto to each Exempt Fund Client, (a) informing each such Exempt Fund Client of the transactions contemplated by this Agreement which would result in a change in control of the Company, and (b) requesting the consent of such Exempt Fund Client to the change in control of the Company if required by Applicable Law or under the respective Management Contract or the organizational documents of such Exempt Fund Client and, if such consent by such Exempt Fund Client is required to be approved, under Applicable

Law or under the respective Management Contract or the organizational documents of such Exempt Fund Client, by each investor in each Exempt Fund Client (each, an "Exempt Fund Client Investor"), requesting such approval. If an Exempt Fund Client is required by Applicable Law or its organizational documents to obtain the consent or approval of its shareholders, it will seek to obtain such consent or approval in a manner reasonably satisfactory to Seller and Buyer.

ARTICLE VIII

ADDITIONAL AGREEMENTS

8.1 Employee Matters.

(a) On the Closing Date, and effective as of the Closing, the Company shall offer immediate employment to substantially all employees of AA who are seconded to the Company (including employees who are not actively at work on account of vacation, illness, short-term disability, which is not expected to result in long-term disability, or leave of absence) (all employees actually hired by the Company and who commence employment with the Company, the "<u>Affected Employees</u>"). For a period commencing on the Closing Date and ending no earlier than the first anniversary of the Closing Date, Buyer shall, and shall cause the Company to, take any action necessary so that each Affected Employee shall continue to receive his or her base wage or salary at a rate no less favorable to such Affected Employee than the rates of base wage or salary paid to such Affected Employees on the date of this Agreement, as long as such Affected Employee remains an employee of Buyer. As to any employees who are not offered employment by the Company, Seller shall cause AA to either reassign such employees or terminate their employment with AA and provide termination and severance benefits available to them under AA's employment policies and procedures. Nothing in this <u>Section 8.1(a)</u> shall limit the right of the Buyer or any of its subsidiaries to terminate the employment of any Affected Employee at any time.

(b) AA and the Company shall jointly give notice to all employees of AA seconded to the Company on the Closing Date that the active participation of Affected Employees in the Pension Plans and Welfare Plans and other employee benefit programs and plans of Seller and its Affiliates shall terminate on the Closing Date, as of which date such Affected Employees shall be eligible to participate in benefit programs provided by Buyer or the Company pursuant to <u>Section 8.1(c)</u>. Except as otherwise expressly provided in this <u>Section 8.1</u>, in no event shall any Affected Employee be entitled to accrue any benefits under any Pension Plan or Welfare Plan or other benefit plan or program of Seller and its Affiliates with respect to services rendered or compensation paid by Buyer or the Company after the Closing.

(c) Buyer and the Company agree to provide Affected Employees with full participation in any benefit plans offered by Buyer to its employees ("<u>Buyer's Benefit</u> <u>Programs</u>") on the same terms and conditions as similarly situated employees of Buyer. If Buyer does not maintain any Buyer's Benefit Programs at the date hereof, or if Buyer's Benefit Programs are not available to the Affected Employees, then for a period commencing on the Closing Date and ending no earlier than December 31, 2008, Buyer shall provide the Affected Employees with benefit plans on terms no less favorable in the aggregate than benefit plans maintained for and provided to Affected Employees immediately prior to the Closing Date (excluding, for this purpose, equity-based compensation, retiree medical and retiree travel benefits) as listed on <u>Schedule 8.1(c)</u> (the "<u>Alternative Benefit</u> <u>Programs</u>"). Such Affected Employees shall be credited for their length of service with Seller and its Affiliates for all purposes (but not the actual accrual of benefits) under Buyer's Benefit Programs or the Alternative Benefit Programs.

(d) Buyer and the Company shall recognize all unused vacation and sick leave of the Affected Employees as of the Closing Date as estimated on <u>Schedule 8.1(d)</u>, and Buyer and the Company shall provide such paid vacation and sick leave, including the responsibility to make immediate cash payments for such accrued vacation amounts at any time on or after the Closing Date to the extent required under Applicable Law, but in no event shall Buyer be required to provide to any single Affected Employee both accrued vacation or sick leave, as applicable, and a cash payment in lieu of such accrued vacation or sick leave, as applicable. Buyer and the Company shall recognize the short term incentive and commission bonus programs for Affected Employees existing as of the Closing Date as listed on <u>Schedule 8.1(d)</u> and maintain such programs without amendment through the end of fiscal 2008 and pay to the Affected Employees the bonuses they have earned thereunder in accordance with the terms thereof; <u>provided</u>, <u>however</u>, subject to the foregoing, that nothing herein shall prevent the amendment or termination of any such program or interfere with the Buyer's right or obligation to make such changes as are necessary to conform with Applicable Law. It is agreed by Buyer and the Company that costs incurred by the Company in connection with the transactions contemplated by this Agreement shall be considered a special nonrecurring expense or loss for purpose of such programs.

(e) Seller or its Affiliates shall retain the responsibility for payment of all medical and dental claims or expenses covered under medical and dental plans maintained by Seller and its Affiliates (the "<u>Seller's Medical Plans</u>") incurred by any Affected Employee on and prior to the Closing Date, and neither Buyer nor the Company shall assume or be responsible for any liability with respect to such claims. Buyer shall remit to Seller all Affected Employee premiums due for medical and dental benefit coverage under Seller's Medical Plans attributable to the period on and prior to the Closing Date, but which, as of the Closing Date, had not been collected and remitted to the Company. Any preexisting condition clause in any of the health coverage (including medical, dental and disability coverage) included in Buyer's Benefits Programs shall be waived for the Affected Employees to the extent such Affected Employee would have been entitled to coverage under the corresponding Seller's Medical Plans in which he or she participated prior to the Closing Date. Buyer and the Company shall credit Affected Employees with any amounts paid under Seller's Medical Plans on and prior to the Closing Date toward satisfaction of the applicable deductible amounts and copayment minimums under the corresponding benefit plans of Buyer to the extent such payments would be taken into account under the benefit plans maintained by Buyer with respect to similarly situated employees.

(f) Buyer and the Company shall be responsible for providing any Affected Employee whose "qualifying event," within the meaning of Section 4980B(f) of the Code, occurs on or after the Closing Date (and such employees' "qualified beneficiaries" within the meaning of Section 4980B(g) of the Code) with the continuation of group health coverage required by Section 4980B(f) of the Code to the extent required by law.

(g) Buyer and the Company shall assume all liability for severance pay and obligations payable to any Affected Employee who is terminated by Buyer or the Company after the Closing. For a period commencing on the Closing Date and ending no earlier than the first anniversary of the Closing Date, Buyer and the Company agree to provide notice of termination or pay in lieu thereof and severance pay, if applicable, to Affected Employees that is no less favorable than the current practices of Seller and its Affiliates as in effect as of the date hereof.

(h) After the Closing, Buyer and the Company shall have the liability and obligation for, and neither Seller nor any of its Affiliates shall have any liability or obligation for, short-term disability benefits, long-term disability benefits, sick pay and salary continuation (and any medical dental and health benefits or claims incurred after the Closing) for the Affected Employees.

(i) With respect to the Affected Employees, Buyer and the Company shall be responsible for all liabilities or obligations under the Worker Adjustment and Retraining Notification Act and similar Applicable Law resulting from Buyer's or the Company's actions following the Closing.

(j) With respect to each Affected Employee, Buyer and the Company shall have the obligation and liability for any workers' compensation or similar workers' protection claims with respect to any such individual incurred after the Closing.

(k) Any amounts payable to the Affected Employees under the Company's 2005 Long-Term Incentive Plan or 1995 Stock Appreciation Rights Plan, as amended, as a result of the Pre-Closing Dividends referenced in <u>Section 7.4(b)(y)</u> or the transactions contemplated hereby shall be paid by the Company promptly after the Closing. Once payment has been made, such plans will be terminated by Buyer and the Company.

(1) For any Affected Employee who has been previously awarded stock-based awards under (a) AMR's 1998 Long Term Incentive Plan, as amended, and its predecessor and (b) the 2003 Employee Stock Incentive Plan (collectively referred to as, the "<u>AMR Equity Awards</u>"), as of the Closing Date, such Affected Employee shall be deemed to have been terminated not for cause for purposes of vesting and/or exercise under the terms and conditions of such AMR Equity Awards. From and after the Closing Date, Affected Employees will not be eligible to receive AMR Equity Awards into the future.

(m) From and after the Closing Date, Affected Employees will cease to participate in all Welfare Plans, Pension Plans and other employee benefit plans or programs established or maintained by the Seller or its Affiliates and if applicable will be deemed to have been terminated not for cause for purposes of vesting, if any, under such plan or program; provided, if a plan or program is part of the Transition Services Agreement, the Affected Employee will continue to participate in such plan or program until the termination of such participation as contemplated by the Transition Services Agreement.

8.2 Insurance; Risk of Loss.

(a) The Company shall and AMR shall cause the Company to keep insurance policies currently maintained by the Company covering its Business and current or former employees, as the case may be, or suitable replacements therefor, in full force and effect through the close of business on the Closing Date. AMR shall keep all insurance covering the Company and the Business in full force and effect through the close of business on the Closing Date. AMR shall become solely responsible for insurance coverage and related risk of loss based on events occurring after the Closing with respect to the Company and the Business. Seller shall be solely responsible for losses based on events occurring prior to the Closing that are within the scope of the Company's insurance policies and AMR's insurance policies that cover the Company as in effect on the date hereof. To the extent that after the Closing any Party hereto requires any information regarding claim data, payroll or other information in order to make any filing with insurance carriers or self insurance regulators from another Party hereto, the other Party will promptly supply such information.

(b) AMR shall procure an insurance and indemnification policy, that provides "run-off coverage", for six (6) years from the Closing Date, for events occurring on or before the Closing Date, that is no less favorable in scope (with commercially reasonable liability limits and deductibles) than AMR's claims-made policies that insure the Company, with respect to directors and officers liability in effect as of the date hereof (the "<u>Existing Policy</u>"), covering each person currently covered by the Existing Policy. Alternatively, for a period of six (6) years from the Closing Date, AMR, at its sole cost, may elect to continue AMR's existing claims-made policies to provide coverage for events occurring on or before the Closing Date on terms no less favorable than those afforded to other individuals insured by AMR's claims-made directors and officers liability policy then in effect and covering each person currently covered by the Existing Policy.

8.3 Section 15 of the Investment Company Act.

From and after the Closing Date, Buyer shall use reasonable best efforts to cause the Company to conduct its business so as to assure that:

(a) for a period of not less than three (3) years following the Closing Date, at least seventy-five percent (75%) of the members of the board of trustees of each Trust are not Interested Persons of Buyer, AMR or the Company or any of their respective Affiliates or any other investment adviser or predecessor investment adviser (within the meaning of the Investment Company Act) to any of the Funds and to the extent applicable, such non-Interested Persons shall have been nominated and elected in accordance with Section 16(b) of the Investment Company Act; and

(b) for a period of not less than two (2) years following the Closing Date, there is not imposed on any Fund an "unfair burden" (within the meaning of Section 15(f) of the Investment Company Act) with respect to the transactions contemplated by this Agreement, or any express or implied terms, conditions or understanding applicable thereto.

8.4 Covenant Not to Compete.

Seller agrees that, for a period of five (5) years following the Closing Date, neither Seller nor any Affiliate of Seller (any "<u>Seller Party</u>") shall, without Buyer's prior written consent, directly or indirectly, engage or participate in, or acquire an ownership interest in a Person providing, Investment Management Services anywhere in the United States in competition with the Company or the Subsidiary; provided that the foregoing shall not prohibit any Seller Party from:

(i) acquiring, in the aggregate, less than a five percent (5%) equity interest in any Person which provides Investment Management Services, it being understood that a Seller Party shall not be considered to have participated in an acquisition of an aggregate interest in any such Person of five percent (5%) or more if such Seller Party (i) acquired less than a five percent (5%) interest in such Person and (ii) acted independently and without knowledge of any other Seller Party acquiring an interest in such Person whose acquired interest in such Person, when added to such Seller Party's acquired interest, is less than a ten percent (10%) interest in such Person and upon discovery of such aggregate interest equal to or greater than five percent (5%) Seller Party uses commercially reasonable efforts to divest the portion of such interest in excess of five percent (5%);

(ii) acquiring an equity interest in any Person which provides Investment Management Services through an account or entity over which such Seller Party has no direct or indirect investment discretion or control;

(iii) acquiring any equity interest in any Person that, among other businesses, provides Investment Management Services; provided that (x) the annual revenue of such company or business derived from the provision of Investment Management Services is less than ten percent (10%) of such company's annual revenues and (y) such Seller Party uses commercially reasonable efforts to divest the portion of any such acquired business providing Investment Management Services as soon as reasonably practicable; or

(iv) being acquired, directly or indirectly, by any third party engaged in Investment Management Services acquires, directly or indirectly, in a transaction in which such third party acquires substantially all of the assets or a majority of the equity interests of a Seller Party.

8.5 Confidentiality; Covenant Not to Solicit.

(a) Following the Closing Date, neither Seller nor any Affiliate of Seller shall, directly or indirectly, make public comments, statement or communication or disclose any confidential information with respect to any matters regarding Buyer, the Company or the Business, or any of the terms, conditions or aspects of the transactions contemplated by this Agreement, except as and to the extent that any such Person shall be so obligated by law or the rules of any stock exchange, in which case Buyer shall be advised and the Parties shall use their best efforts to cause a mutually agreeable release or announcement to be issued.

(b) AMR agrees that, for a period of five (5) years following the Closing Date, neither AMR nor any other Seller Party shall, without Buyer's prior written consent, directly or indirectly, solicit for employment or as a consultant any individual who is at the time of, or was within a one (1) year period prior to, the solicitation, an employee of the Company or the Subsidiary receiving a salary of at least \$100,000 and an Affected Employee (it being understood that the publication of a solicitation for employment in a newspaper or magazine of general circulation shall not alone be deemed a solicitation for employment in violation of this <u>Section 8.5</u>).

8.6 <u>Relief for Violation</u>.

AMR recognizes that the performance of its obligations under <u>Sections 8.4</u> and <u>8.5</u> are special, unique and extraordinary in character and that, in the event of any actual or threatened violation of any provision of <u>Sections 8.4</u> or <u>8.5</u>. Buyer shall be entitled to institute and prosecute proceedings in any court of competent jurisdiction, either in law or in equity, to obtain damages for any actual violation, enforce the specific performance and/or enjoin any actions in violation of <u>Sections 8.4</u> or <u>8.5</u>.

8.7 Client Fees.

Buyer agrees that, for a period of five (5) years after the Closing Date, Buyer shall not make or propose any increases in the fee schedules contained in the Management Contracts of any Related Client as such fee schedules are set forth on <u>Schedule 5.9</u>.

8.8 Transfer Expenses.

All federal, state, local and foreign sales, transfer, stamp and notarial taxes, fees and other duties (collectively, "<u>Transfer Expenses</u>"), if any, under Applicable Law incurred in connection with the sale and transfer of the Shares pursuant to this Agreement will be borne and paid fifty percent (50%) by Buyer and fifty percent (50%) by Seller, and Buyer shall promptly reimburse Seller for fifty percent (50%) of any Transfer Expenses which Seller and/or any of its affiliates is required to pay under Applicable Law.

8.9 Fees of Independent Fiduciary.

AMR agrees that it or its Affiliates shall pay or cause to be paid all expenses incurred up to, on and after the Closing Date for engaging an independent fiduciary with respect to any ERISA Client that is, holds the "plan assets" of, or acts on behalf of, any employee benefit plan maintained by AMR or any Affiliate of AMR.

8.10 <u>338(h)(10) Election</u>.

Buyer and Seller shall join in making a valid, timely election under Section 338(h)(10) of the Code (and any corresponding elections under state and/or local tax law, collectively, a "338(h)(10) Election") with respect to the Buyer's acquisition of the Shares in accordance with applicable statutes, rules and regulations, including IRS Form 8023 and Treasury Regulation Section 1.338(h)(10)-1. Seller shall deliver or cause to be delivered to Buyer at Closing IRS Form 8023 properly and fully executed and completed with respect to Buyer's acquisition of the Shares. Buyer shall prepare a properly and fully completed IRS Form 8083 pursuant to Treasury Regulation Section 1.1060-1 or shall take such other action required pursuant to the Treasury Regulations under Section 338(h)(10) of the Code to report the allocation of the Purchase Price (the "Purchase Price Allocation"). Seller agrees that, within fifteen (15) days of receiving the Purchase Price Allocation, complying with the requirements of the preceding sentences it shall sign the Purchase Price Allocation and return an executed copy thereof to the Buyer. Neither Seller nor Buyer shall, nor shall permit any of their Affiliates to, file any Tax Return, or take any position with a taxing authority, that is inconsistent with the Purchase Price Allocation.

8.11 Intellectual Property.

From and after the Closing, AMR agrees, at Buyer's request and sole expense, to provide reasonable assistance with filing registrations for Company Intellectual Property or for causing Company Intellectual Property to be registered.

8.12 Audited Financials.

From and after the Closing, AMR agrees, at Buyer's request and sole expense, to provide reasonable assistance to Buyer, including reasonable access during normal business hours, upon reasonable prior notice and in such manner as will not unreasonably interfere with operations or the conduct of the business of such persons, to the officers and employees of Seller and its Affiliates, for preparing audited financial statements for periods prior to or for fiscal periods that include the Closing Date and/or for preparing registration materials for a public offering of Buyer's securities.

8.13 AMR Plans.

(a) During the five (5) year period commencing on the Closing Date, with respect to any ERISA Client that is, holds the "plan assets" of, or acts on behalf of, any employee benefit plan maintained by AMR or any Affiliate of AMR and is or was a Related Client, AMR shall pay or shall cause to be paid from assets other than the assets of such plans, a portion of the total investment management fees charged by the managers of all such plans, whether any such manager is the Company or a third party, in an amount not less than \$5,000,000 per year (the "Obligation"). Solely for purposes of this calculation to the extent any such fees are paid to a manager other than the Company, such Obligation shall be reduced by a percentage amount equal to AMR's average fully diluted equity interest in the Company during such year. The purpose of the adjustment shall be to ensure that AMR is economically indifferent in terms of amounts of any fees paid in meeting this obligation between payments to the Company and payments to other managers.

8.14 Transition Services Agreement.

(a) From the date hereof to the Closing Date, each of the parties to this Agreement will use commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary or advisable to ensure that the Company owns and possesses or continues to use and possess, as applicable, all, tangible and intangible, assets and rights and has access to all services provided by the Seller or by contractors of the Seller to the Company, as are necessary to operate and carry on the Business as conducted immediately prior to the date of this Agreement.

(b) The parties agree to negotiate in good faith the preparation and execution of the Transition Services Agreement.

ARTICLE IX

CONDITIONS PRECEDENT TO OBLIGATIONS OF BUYER

The obligations of Buyer under this Agreement shall, at the option of Buyer (to the extent permissible under Applicable Law), be subject to the satisfaction, on or prior to the Closing Date, of the following conditions:

9.1 No Misrepresentation or Breach of Covenants and Warranties. Each of the Company and

Seller shall have performed or complied in all material respects with all of its covenants and agreements herein, each of the representations and warranties of Seller and the Company contained in this Agreement shall be true and correct on the Closing Date as though made on the Closing Date (except to the extent that they expressly relate to an earlier date) without giving effect to any materiality or Material Adverse Effect qualification set forth therein, except for changes therein specifically resulting from any transaction expressly consented to in writing by Buyer and other than breaches of representations and warranties which, individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect; and there shall have been delivered to Buyer a certificate to such effect, dated the Closing Date, signed on behalf of Seller by a duly authorized officer.

9.2 No Restraint.

The waiting period under the HSR Act (if applicable) shall have expired or been terminated, and no injunction or restraining order shall have been issued by any court of competent jurisdiction and be in effect which restrains or prohibits any material transaction contemplated hereby or imposes conditions materially adverse, to any Party hereunder, including requiring any divestiture of assets or imposing any restrictions on operations.

9.3 Governmental Approvals.

The Parties shall have made all filings with, and shall have received all approvals and actions of or by, all Governmental Bodies necessary to consummate the transactions contemplated hereby without imposing any conditions materially adverse to any Party, including requiring any divestiture of assets or imposing any restrictions on operations, and to allow the Company to continue the Business immediately following the Closing Date in substantially the manner it is currently conducted, other than any filings, approvals and actions which are not required to be made or obtained prior to the Closing by Applicable Law.

9.4 <u>No Litigation</u>.

No action, suit or proceeding shall be pending before any court or quasi-judicial or administrative agency of any federal, state, local, or foreign jurisdiction or before any arbitrator wherein an unfavorable judgment, decree, injunction, order or ruling would prevent the performance of any material transaction contemplated hereby, declare unlawful the transactions contemplated by this Agreement or cause such transactions to be rescinded, and no judgment, decree, injunction, order or ruling shall have been entered which has any of the foregoing effects.

9.5 <u>Closing Revenue Run-Rate</u>.

The Closing Revenue Run-Rate shall comprise at least eighty percent (80%) of the Base Revenue Run-Rate.

9.6 New Management Contracts.

Each ERISA Client shall have entered into a new Management Contract with the Company, in form and substance satisfactory to Buyer, which new Management Contracts shall have been duly authorized and approved under all Applicable Law (including, without limitation, with respect to each ERISA Client, by its independent fiduciary).

9.7 Closing Date Deliveries.

Seller shall have delivered to Buyer each of the closing date deliveries listed in Section 4.4.

ARTICLE X

CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLER AND THE COMPANY

The obligations of Seller and the Company under this Agreement shall, at the option of Seller (to the extent permissible under Applicable Law), be subject to the satisfaction, on or prior to the Closing Date, of the following conditions:

10.1 No Misrepresentation or Breach of Covenants and Warranties.

Buyer shall have performed or complied in all material respects with all of its covenants and agreements herein, each of the representations and warranties of Buyer contained in this Agreement shall be true and correct on the Closing Date as though made on the Closing Date (except to the extent they expressly relate to an earlier date) without giving effect to any materiality limitation or qualification, except for changes therein specifically resulting from any transaction expressly consented to in writing by the Seller and other than breaches of representations and warranties which, individually or in the aggregate, have not had and would not reasonably be expected to have a material adverse effect on Buyer's ability to consummate the transactions contemplated hereby; and there shall have been delivered to Seller a certificate to such effect, dated the Closing Date, signed on behalf of Buyer by a duly authorized officer of Buyer.

10.2 No Restraint.

The waiting period under the HSR Act (if applicable) shall have expired or been terminated, and no injunction or restraining order shall have been issued by any court of competent jurisdiction and be in effect which restrains or prohibits any material transaction contemplated hereby or imposes conditions materially adverse to any Party hereunder, including requiring any divestiture of assets or imposing any restrictions on operations.

10.3 <u>Governmental Approvals</u>.

The Parties shall have made all filings with, and shall have received all approvals and actions of or by, all Governmental Bodies necessary to consummate the transactions contemplated hereby without imposing any conditions materially adverse to any Party, including requiring any divestiture of assets or imposing any restrictions on operations, and to allow the Company to continue the Business immediately following the Closing Date in substantially the manner it is currently conducted, other than any filings, approvals and actions which are not required to be obtained prior to the Closing by Applicable Law.

10.4 No Litigation.

No action, suit or proceeding shall be pending before any court or quasi-judicial or administrative agency of any federal, state, local, or foreign jurisdiction or before any arbitrator wherein an unfavorable judgment, decree, injunction, order or ruling would prevent the performance of any material transaction contemplated hereby, declare unlawful the transactions contemplated by this Agreement or cause such transactions to be rescinded, and no judgment, decree, injunction, order or ruling shall have been entered which has any of the foregoing effects.

10.5 Closing Revenue Run-Rate.

The Closing Revenue Run-Rate shall comprise not more than one hundred twenty percent (120%) of the Base Revenue Run-Rate.

10.6 New Management Contracts.

The Company shall have entered into a new Management Contract with each ERISA Client, in form and substance satisfactory to Buyer, which new Management Contracts shall have been duly authorized and approved under all Applicable Law (including without limitation, with respect to each ERISA Client, by its independent fiduciary).

10.7 Section 15 Compliance.

Effective upon the Closing, no more than twenty-five percent (25%) of the members of the board of trustees (as applicable) of any Trust shall be Interested Persons of Buyer, AMR, the Company or any of their respective Affiliates or any other investment adviser or predecessor investment adviser (within the meaning of the Investment Company Act) to any of the Funds.

10.8 Closing Date Deliveries.

Buyer shall have delivered to Seller each of the closing date deliveries listed in Section 4.3.

ARTICLE XI

INDEMNIFICATION

11.1 Indemnification by Seller.

(a) From and after the Closing, Seller agrees to indemnify and hold harmless each Buyer Group Member from and against any and all Losses and Expenses incurred by such Buyer Group Member in connection with or arising from:

(i) any breach of any warranty or the inaccuracy of any representation of Seller and the Company contained or referred to in this Agreement or the certificate delivered by or on behalf of Seller and the Company pursuant to <u>Section 9.1(a)</u>;

(ii) any breach by Seller and the Company of, or failure by Seller and the Company to perform, any of their respective covenants or obligations contained in this Agreement;

(iii) any claim brought by an ERISA Client who is or was prior to the Closing Date a Related Client for events that occurred in whole or in part on or prior to the Closing Date;

(iv) any Taxes imposed on or in respect of the Company, the Subsidiary or any of the Funds in respect of any Pre-Closing Tax Period (including as a result of the Company, the Subsidiary or any of the Funds being treated prior to the Closing as a member of any consolidated, combined, unitary or similar group with respect to which the Company is not the common parent pursuant to the laws of the United States, any foreign jurisdiction or any state or locality), as reasonably determined in accordance with the Company's ordinary and customary Tax reporting practices, and based on a closing of the books of the Company, the Subsidiary or any of the Funds at the end of the Closing Date, and

(v) any Losses or Expenses relating to or arising in connection with the claims against the Company by Ronald A. Katz Technology Licensing, L.P. or its Affiliates, successors or assigns, including the matter described in Schedule 5.12.

(b) Notwithstanding anything in <u>Section 11.1(a)</u> to the contrary, Seller shall be required to indemnify and hold harmless each Buyer Group Member in respect of claims for indemnification made under <u>Section 11.1(a)(i)</u> for breaches of representations and warranties only to the extent that:

(i) the amount of Loss and Expense suffered by Buyer Group Members related to such individual claim under <u>Section 11.1(a)(i)</u> exceeds \$25,000 (it being understood that such \$25,000 shall be a deductible for which Seller shall bear no indemnification responsibility);

(ii) the aggregate amount of all Losses and Expenses for claims under <u>Section 11.1(a)(i)</u> exceeds one percent (1%) of the Final Cash Purchase Price; provided that once such Losses and Expenses equal or exceed such threshold, subject to <u>Sections 11.1(b)(i)</u> and <u>11.1(b)(iii)</u>, Seller shall be liable for all Losses and Expenses (including any Losses and Expenses previously not indemnified pursuant to this <u>Section 11.1(b)(i)</u>; and

(iii) the aggregate amount required to be paid by Seller related to such claims under <u>Section 11.1(a)(i)</u> shall not exceed ten percent (10%) of the Final Cash Purchase Price.

Notwithstanding anything to the contrary herein, the limitations contained in clauses (i), (ii) and (iii) shall not apply to any Loss or Expense incurred by any Buyer Group Member: (A) in connection with or arising from fraud, (B) any breach of any representation or warranty in <u>Sections 5.2(b), 5.3, 5.4(a), 5.7</u> and <u>5.24(g)</u> or (C) for claims pursuant to <u>Sections 11.1(a)(ii), 11.1(a)(iii), 11.1(a)(ii)</u> and <u>11.1(a)(v)</u>. In no event shall the aggregate amount required to be paid by Seller without such limitations pursuant to this <u>Section 11.1(a)</u> exceed the Final Cash Purchase Price.

(c) The indemnification provided for in <u>Section 11.1(a)(i)</u> shall terminate eighteen months after the Closing Date (and no claims shall be made by any Buyer Group Member under <u>Section 11.1(a)(i)</u> thereafter), except that the indemnification by the Company under <u>Section 11.1(a)(i)</u> shall continue as to:

(i) the representations and warranties of Seller and the Company set forth in <u>Sections 5.2</u>, <u>5.3</u>, <u>5.4(a)</u>, <u>5.7</u>, <u>5.9(g)</u>, <u>5.14</u>, <u>5.17(c)</u>, <u>5.24(d)</u>, and <u>5.24(g)</u>, which shall survive until the expiration of the relevant statutory period of limitations applicable to the underlying claim, giving effect to any waiver, mitigation or extension thereof;

(ii) the representations and warranties of Seller and the Company set forth in Sections 5.9 (other than 5.9(g)), 5.17 (other than 5.17(d)), 5.21 and 5.24 (other than 5.24(g)), which shall survive until the earlier of (x) the fifth anniversary of the Closing Date and (y) the relevant statutory period of limitations applicable to the underlying claim, giving effect to any waiver, mitigation or extension thereof;

(iii) any Losses or Expenses of which any Buyer Group Member has validly given a Claim Notice to Seller in accordance with the requirements of <u>Section 11.3</u> on or prior to the date such indemnification would otherwise terminate in accordance with this <u>Section 11.1(c)</u>, as to which the obligation of Seller shall continue solely with respect to the specific matters in such Claim Notice until the liability of Seller shall have been determined pursuant to this <u>Article XI</u>, and Seller shall have reimbursed all Buyer Group Members for the full amount of such Losses and Expenses that are payable with respect to such Claim Notice in accordance with this <u>Article XI</u>; and

(iv) claims for indemnification based on fraud.

(d) The indemnification provided for in Section 11.1(a)(ii) shall survive the execution and delivery of this Agreement and the Closing until such obligations and covenants identified therein are performed or the obligation to so perform shall have expired and for a period of ninety (90) days thereafter, except the covenants of Seller set forth in <u>Articles II</u> and <u>III</u> and <u>Section 8.2</u> shall survive until the expiration of the relevant statutory period of limitations applicable to the underlying claim, giving effect to any waiver, mitigation or extension thereof; and, <u>provided</u>, that any Losses or Expenses of which any Buyer Group Member has validly given a Claim Notice to Seller in accordance with the requirements of <u>Section 11.3</u> on or prior to the date such indemnification would otherwise terminate in accordance with this <u>Section 11.1(d)</u>, as to which the obligation of Seller shall continue solely with respect to the specific matters in such Claim Notice until the liability of Seller shall have been determined pursuant to this <u>Article XI</u>, and Seller shall have reimbursed all Buyer Group Members for the full amount of such Losses and Expenses that are payable with respect to such Claim Notice in accordance with this <u>Article XI</u>.

(e) The indemnification provided for in <u>Section 11.1(a)(iii)</u> shall survive until the expiration of the relevant statutory period of limitations applicable to the underlying claim (and no claims shall be made by any Buyer Group Member under <u>Section 11.1(a)(iii)</u> thereafter).

(f) The indemnification provided for in Section 11.1(a)(iv) shall survive the Closing Date and continue until thirty (30) days following the expiration of the statute of limitations on assessment of any Taxes. Notwithstanding the foregoing, any claim for indemnification shall survive such termination date if any Buyer Group Member, prior to such termination date, shall have advised Seller in writing of facts that constitute or may give rise to an alleged claim for indemnification under Section 11.1(a)(iv), specifying in reasonable detail the basis under this Agreement for such claim.

(g) The indemnification provided for in Section 11.1(a)(y) shall survive the Closing Date and continue until the liability of Seller shall have been determined pursuant to this Article XI.

(a) From and after the Closing, Buyer agrees to indemnify and hold harmless each Seller Group Member from and against any and all Losses and Expenses incurred by such Seller Group Member in connection with or arising from:

(i) any breach of any warranty or the inaccuracy of any representation of Buyer contained or referred to in this Agreement or the certificate delivered by or on behalf of Buyer pursuant to <u>Section 10.1(a)</u>; and

(ii) any breach by Buyer of, or failure by Buyer to perform, any of its covenants and obligations contained in this Agreement.

(b) Notwithstanding anything in <u>Section 11.2(a)</u> to the contrary, Buyer shall be required to indemnify and hold harmless each Seller Group Member in respect of claims for indemnification made under <u>Section 11.2(a)(i)</u> for breaches of representations and warranties only to the extent that:

(i) the amount of Loss and Expense suffered by Seller Group Members related to each individual claim under <u>Section 11.2(a)(i)</u> exceeds \$25,000 (it being understood that such \$25,000 shall be a deductible for which Buyer shall bear no indemnification responsibility);

(ii) the aggregate amount of all Losses and Expenses for claims under <u>Section 11.2(a)(i)</u> exceeds one percent (1%) of the Final Cash Purchase Price; provided that once such Losses and Expenses equal or exceed such threshold, subject to <u>Sections 11.2(b)(ii)</u> and <u>11.2(b)(iii)</u>, Buyer shall be liable for all Losses and Expenses (including any Losses and Expenses previously not indemnified pursuant to this <u>Section 11.2(b)(ii)</u>; and

(iii) the aggregate amount required to be paid by Buyer related to such claims under <u>Section 11.2(a)(i)</u> shall not exceed ten percent (10%) of the Final Cash Purchase Price.

Notwithstanding anything to the contrary herein, the limitations contained in clauses (i), (ii) and (iii) shall not apply to any Loss or Expense incurred by any Seller Group Member: (A) in connection with or arising from fraud, (B) any breach of any representation or warranty in <u>Section 6.2(a)</u> or (C) for claims pursuant to <u>Section 11.2(a)(ii)</u>. In no event shall the aggregate amount required to be paid by Buyer without such limitations pursuant to this <u>Section 11.2(a)</u> exceed the Final Cash Purchase Price.

(c) The indemnification provided for in <u>Section 11.2(a)(i)</u> shall terminate eighteen months after the Closing Date (and no claims shall be made by any Seller Group Member under <u>Section 11.2(a)(i)</u> thereafter), except that the indemnification by Buyer shall continue as to:

(i) the representations and warranties of Buyer set forth in <u>Section 6.2(a)</u>, which shall survive until the expiration of the relevant statutory period of limitations applicable to the underlying claim, giving effect to any waiver, mitigation or extension thereof;

(ii) the representations and warranties of Buyer set forth in <u>Sections 6.6, 6.8</u> or <u>6.10</u> which shall survive until the earlier of (x) the fifth anniversary of the Closing Date and (y) the relevant statutory period of limitations applicable to the underlying claim, giving effect to any waiver mitigation or extension thereof;

(iii) any Losses or Expenses of which any Seller Group Member has validly given a Claim Notice to Buyer in accordance with the requirements of <u>Section 11.3</u> on or prior to the date such indemnification would otherwise terminate in accordance with this <u>Section 11.2(c)</u>, as to which the obligation of Buyer shall continue solely with respect to the specific matters in such Claim Notice until the liability of Buyer shall have been determined pursuant to this <u>Article XI</u>, and Buyer shall have reimbursed all Seller Group Members for the full amount of such Losses and Expenses that are payable with respect to such Claim Notice in accordance with this <u>Article XI</u>; and

(iv) claims for indemnification based on fraud.

(d) The indemnification provided for in Section 11.2(a)(ii) shall survive the execution and delivery of this Agreement and the Closing until such covenants and obligations identified therein are performed or the obligation to so perform shall have expired and for a period of ninety (90) days thereafter, except the covenants of Buyer set forth in Articles II and III and Sections 8.1, 8.2 and 8.3 shall survive until the expiration of the relevant statutory period of limitations applicable to the underlying claim, giving effect to any waiver, mitigation or extension thereof; provided, that any Losses or Expenses of which any Seller Group Member has validly given a Claim Notice to Buyer in accordance with the requirements of Section 11.3 on or prior to the date such indemnification would otherwise terminate in accordance with this Section 11.2(d), as to which the obligation of Buyer shall continue solely with respect to the specific matters in such Claim Notice until the liability of Buyer shall have been determined pursuant to this Article XI, and Buyer shall have reimbursed all Seller Group Members for the full amount of such Losses and Expenses that are payable with respect to such Claim Notice in accordance with this Article XI.

11.3 Notice of Claims.

Any Buyer Group Member or Seller Group Member seeking indemnification hereunder (the "<u>Indemnified Party</u>") shall give promptly to the party obligated to provide indemnification to such Indemnified Party (the "<u>Indemnitor</u>") a notice (a "<u>Claim Notice</u>") describing in reasonable detail the facts giving rise to the claim for indemnification hereunder and shall include in such Claim Notice (if then known) the amount or the method of computation of the amount of such claim, and a reference to the provision of this Agreement or any other agreement, document or instrument executed hereunder or in connection herewith upon which such claim is based; <u>provided</u>, <u>however</u>, that a Claim Notice in respect of any action at law or suit in equity by or against a third Person as to which indemnification will be sought shall be governed by <u>Section 11.5</u>.

11.4 Determination of Amount.

(a) For purposes of determining Losses and Expenses pursuant to Section 11.1(a)(i) and Section 11.2(a)(i), representations and warranties shall be read without regard to any materiality, Material Adverse Effect or knowledge limitation or qualification contained therein.

(b) In calculating any Loss or Expense there shall be deducted any insurance proceeds which the Indemnified Party has received. If an Indemnitor is required to indemnify an Indemnified Party pursuant to the provisions of <u>Sections 11.1</u> or <u>11.2</u>, and the cost, expense or liability for which the indemnification is sought has provided such Indemnified Party with a Tax benefit that is actually recognized currently, the amount of such Tax benefit shall reduce the Indemnitor's liability to indemnify such Indemnified Party.

(c) After the giving of any Claim Notice pursuant to <u>Section 11.3</u>, the amount of indemnification to which an Indemnified Party shall be entitled under this <u>Article XI</u> shall be determined: (i) by the written agreement between the Indemnified Party and the Indemnifier, (ii) by a final judgment or decree of any court, arbitration board or administrative agency of competent jurisdiction; or (iii) by any other means to which the Indemnified Party and the Indemnitor shall agree. The judgment or decree of a court, arbitration board or administrative agency shall be deemed final when the time for appeal, if any, shall have expired and no appeal shall have been taken or when all appeals taken shall have been finally determined.

(d) Buyer and Seller agree that, for purposes of computing the amount of any indemnification payment under this <u>Article XI</u>, the parties shall treat any such indemnification payment as an adjustment to the Final Cash Purchase Price for all Tax purposes. If, contrary to the position of the parties, any payment made pursuant to this Agreement is required by any taxing authority to be treated as taxable income of the recipient, then the payor shall indemnify and hold harmless the recipient on an after-Tax basis from any liability for Taxes attributable to the receipt of such payment.

11.5 Third Person Claims.

(a) Any party seeking indemnification provided for under this Agreement in respect of, arising out of or involving a claim or demand made by any third Person against the Indemnified Party shall notify the Indemnitor in writing, and in reasonable detail, of the third Person claim within thirty (30) days after receipt by such Indemnified Party of written notice of the third Person claim. Any notice of a claim by reason of any of the representations, warranties or covenants contained in this Agreement shall contain a reference to the provision of this Agreement or any other agreement, document or instrument executed hereunder or in connection herewith upon which such claim is based, the facts giving rise to an alleged basis for the claim and (if then known) the amount or method of computation of the liability asserted against the Indemnitor by reason of the claim. Thereafter, the Indemnified Party shall deliver to the Indemnitor, within 10 Business Days after the Indemnified Party's receipt thereof, copies of all notices and documents (including court papers) received by the Indemnified Party relating to the third Person claim. Notwithstanding the foregoing, should a party be physically served with a complaint with regard to a third Person claim, the Indemnified Party shall provide the Indemnitor with a copy of the complaint within five (5) Business Days after receipt thereof and shall deliver to the Indemnified Party shall provide the Indemnitor opies of notices and documents (including court papers) received by the Indemnified Party shall provide the receipt of such complaint copies of notices and documents (including court papers) received by the Indemnified Party shall provide the Indemnitor with a copy of the complaint within five (5) Business Days after receipt thereof and shall deliver to the Indemnitor within seven (7) Business Days after the receipt of such complaint copies of notices and documents (including court papers) received by the Indemnified Party relating to the third Person claim. The failure t

(b) Within ten (10) Business Days after receiving a Claim Notice for indemnification or reimbursement under <u>Section 11.3</u> or this <u>Section 11.5</u>, the Indemnitor shall, by written notice to the Indemnified Party, either (i) concede or deny liability for the claim in whole or in part, or (ii) in the case of a claim asserted by a third party, advise that the matters set forth in the notice are, or will be, subject to contest or legal proceedings not yet finally resolved. If the Indemnitor concedes liability in whole or in part, it shall, within thirty (30) Business Days of such concession, make payment of the amount of the claim to the Indemnified Party to the extent of the liability conceded, in immediately available funds equal to the amount of such claim so payable. If the Indemnitor shall make no payment (except for the amount of any conceded liability payable as set forth above) until the matter is resolved in accordance with this Agreement.

(c) In the case of any third party claim, if within ten (10) Business Days after receiving the notice described in the preceding paragraph (a), the Indemnitor gives written notice to the Indemnified Party stating that the Indemnitor would be liable under the provisions hereof for indemnity in the amount of such claim if such claim were valid and that the Indemnitor disputes and intends to defend against such claim, liability or expense at the Indemnitor's own cost and expense then counsel for the defense shall be selected by the Indemnitor (subject to the consent of such Indemnified Party which consent shall not be unreasonably withheld) and the Indemnitor shall not be required to consent to any payment to the Indemnified Party with respect to such claim, liability or expense as long as the Indemnitor is conducting a good faith and diligent defense at its own expense; provided, however, that the assumption of defense of any such matters by the Indemnitor shall relate solely to the claim, liability or expense that is subject or potentially subject to indemnification If the Indemnitor assumes such defense in accordance with the preceding sentence, it shall have the right, with the consent of such Indemnified Party, which consent shall not be unreasonably withheld, to settle all indemnifiable matters related to claims by third parties which are susceptible to being settled provided the Indemnitor's obligation to indemnify such Indemnified Party therefor will be fully satisfied only by payment of money by the Indemnitor pursuant to a settlement which includes a complete release of such Indemnified Party. Notwithstanding anything herein stated, such Indemnified Party shall at all times have the right to fully participate in such defense at its own expense directly or through counsel; provided, however, if the named parties to the action or proceeding include both the Indemnitor and the Indemnified Party and representation of both parties by the same counsel would be inappropriate under applicable standards of professional conduct, the reasonable expense of separate counsel for such Indemnified Party shall be paid by the Indemnitor provided that such Indemnitor shall be obligated to pay for only one counsel for the Indemnified Party. If no such notice of intent to dispute and defend is given by the Indemnitor, or if such diligent good faith defense is not being or ceases to be conducted, such Indemnified Party may undertake the defense of (with counsel selected by such Indemnified Party), and shall have the right to compromise or settle, such claim, liability or expense (exercising reasonable business judgment) with the consent of the Indemnitor, which consent shall not be unreasonably withheld. After any final judgment or award shall have been rendered by a court, arbitration board or administrative agency of competent jurisdiction and the time in which to appeal therefrom has expired, or a settlement shall have been consummated, or the Indemnified Party and the Indemnitor shall arrive at a mutually binding agreement with respect to each separate matter to be indemnified by the Indemnitor hereunder, the Indemnified Party shall forward to the Indemnitor notice of any sums due and owing by it with respect to such matter and the Indemnitor shall pay all of the sums so owing to the Indemnified Party by wire transfer, certified or bank cashier's check within thirty (30) days after the date of such notice.

11.6 Limitations.

(a) If the Indemnified Party receives insurance proceeds with respect to an indemnified Loss or Expense after the full amount of such indemnified Loss has been paid by the Indemnifying Party, then such Indemnified Party shall promptly remit to the Indemnifying Party an amount equal to the excess (if any) of (i) the amount theretofore paid by the Indemnifying Party in respect of such indemnified Loss or Expense, less (ii) the amount of the indemnity payment that would have been due if such insurance proceeds in respect thereof had been received before the indemnity payment was made, net of any expenses incurred by such Indemnified Party in collecting such insurance proceeds.

(b) Seller shall not be required to indemnify and hold harmless any Buyer Group Member pursuant to <u>Section 11.1(a)</u> if, and only to the extent that, any Losses or Expenses incurred by such Buyer Group Member were received by, or otherwise credited to, Buyer in the computation of the Final Cash Purchase Price pursuant to <u>Section 3.3</u> (including by means of any reserve set forth in the Net Working Capital Adjustment Report with respect to the type of such matter).

(c) Except for remedies that cannot be waived as a matter of law and injunctive and provisional relief (including, but not limited to, specific performance), this <u>Article</u> <u>XI</u> shall be the exclusive remedy for breaches of this Agreement (including any covenant, obligation, representation or warranty contained in this Agreement or in any certificate delivered pursuant to this Agreement) or otherwise in respect of the sale of the Shares contemplated hereby.

(d) Notwithstanding anything to the contrary contained herein, the Seller shall be liable, without any right of claim against the Company, for any breach of the representations and warranties by the Company contained herein made as of the date hereof or as of the Closing Date or pursuant to the certificate delivered pursuant to Section 9.1. After the Closing, Seller shall have no rights against the Company, its Subsidiary or any director, officer, or employee thereof (in their capacity as such), whether by reason of contribution, indemnification, reimbursement, subrogation, or otherwise, in respect of any payment by Seller for any Losses or Expenses claimed by any Buyer Party pursuant to this Agreement and Seller shall not take any action against such Person with respect thereto. The Company, its Subsidiary and any director, officer or employee thereof shall have no liability or obligation to Seller with respect to any representation by the Company contained herein, and Seller has no right of indemnification, reimbursement contribution, subrogation or otherwise from the Company, its Subsidiary or any director, officer or employee thereof.

ARTICLE XII

TERMINATION

12.1 <u>Termination</u>.

Anything contained in this Agreement to the contrary notwithstanding, this Agreement may be terminated at any time prior to the Closing Date:

(a) by the mutual consent of Buyer and Seller;

(b) by Buyer in the event of any breach by Seller or the Company of any of Seller's and the Company's covenants or agreements set forth in this Agreement, or if any of the representations and warranties of Seller or the Company becomes untrue without giving effect to any materiality limitation or qualification, except for changes therein specifically resulting from any transaction expressly consented to in writing by Buyer and other than breaches of representations and warranties which, individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect and such breach or untruth (i) cannot be cured within sixty (60) days of the date on which Seller receives written notice thereof or (ii) has not been cured within thirty (30) days after receipt of notice by Seller from Buyer requesting such breach to be cured;

(c) by Seller in the event of any breach by Buyer of any of Buyer's covenants or agreements set forth in this Agreement, or if any of the representations and warranties of Buyer becomes untrue without giving effect to any materiality limitation or qualification, except for changes therein specifically resulting from any transaction expressly

consented to in writing by Seller and other than breaches of representations and warranties which, individually or in the aggregate, have not had and would not reasonably be expected to have a material adverse effect on Buyer's ability to consummate the transactions contemplated hereby and such breach or untruth (i) cannot be cured within sixty (60) days of the date on which the Buyer receives written notice thereof or (ii) has not been cured within thirty (30) days after receipt of written notice by Buyer from Seller requesting such breach to be cured;

(d) by Buyer or Seller and the Company if any court of competent jurisdiction in the United States or other United States Governmental Body shall have issued a final and non-appealable order, decree or ruling permanently restraining, enjoining or otherwise prohibiting the consummation of the transactions contemplated hereby; <u>provided</u> that a party may not terminate this Agreement pursuant to this <u>Section 12.1(d)</u> if it or its Affiliates' failure to perform its obligations under this Agreement resulted in or substantially contributed to the issuance of such judgment, injunction or decree; or

(e) by Buyer or Seller and the Company if the Closing shall not have occurred on or before December 31, 2008 (or such later date as may be agreed to in writing by Buyer and Seller and the Company).

12.2 Notice of Termination.

Any Party desiring to terminate this Agreement pursuant to Section 12.1 shall give written notice of such termination to the other Parties to this Agreement.

12.3 Effect of Termination. In the event that this Agreement shall be terminated pursuant to this <u>Article XII</u>, all further obligations of the Parties under this Agreement (other than <u>Sections 13.3</u> and <u>13.10</u>) shall be terminated without further liability of any Party to the other; <u>provided</u>, <u>however</u>, that nothing herein shall relieve any Party from liability for its willful breach of this Agreement.

12.4 <u>Right to Specific Performance</u>. Notwithstanding anything to the contrary contained in this Agreement, it is specifically understood and agreed that any material breach by either Party of its obligations with respect to the purchase and sale of the Shares will result in irreparable injury to the other Party, that the remedies available to such other Party at law alone will be an inadequate remedy for such breach, and that, in addition to any other legal or equitable remedies which such other Party may have, such other Party may enforce its rights in court by an action for specific performance and the Parties expressly waive the defense that a remedy in damages will be adequate.

ARTICLE XIII

GENERAL PROVISIONS

13.1 Survival of Representations and Warranties.

All representations and warranties contained in this Agreement shall survive the consummation of the transactions contemplated by this Agreement through the period during which claims for indemnification may be made for such representations and warranties pursuant to <u>Article XI</u> (at which time such representations and warranties shall terminate).

13.2 Governing Law.

This Agreement shall be governed by and construed in accordance with the internal laws (as opposed to the conflicts of law provisions) of the State of Delaware.

13.3 No Public Announcement.

Neither Buyer nor the Company shall, without the approval of the other, make any press release or other public announcement concerning the transactions contemplated by this Agreement, except as and to the extent that any such Party shall be so obligated by law, in which case the other Party shall be advised and the Parties shall use their reasonable best efforts to cause a mutually agreeable release or announcement to be issued; <u>provided</u>, <u>however</u>, that the foregoing shall not preclude communications or disclosures necessary to implement the provisions of this Agreement or to comply with the accounting and the SEC disclosure obligations or the rules of any stock exchange.

13.4 Notices.

All notices or other communications required or permitted hereunder shall be in writing and shall be deemed given or delivered when delivered personally or upon electronic confirmation of receipt when transmitted by facsimile transmission (but only if followed by transmittal by overnight courier or hand delivery on the next Business Day) or on receipt after sent by registered or certified mail or by private courier addressed as follows:

If to Buyer, to:

TPG Capital, L.P. 301 Commerce Street Suite 3300 Fort Worth, Texas 76102 Fax: (817) 871-4088 Attention: Clive D. Bode, Esq.

Pharos Capital Group, LLC 300 Crescent Court, Suite 1380 Dallas, TX 75201 Fax: (214) 855-1230 Attention: Kneeland Youngblood

with a copy to (which shall not constitute notice):

Cleary Gottlieb Steen & Hamilton LLP One Liberty Plaza New York, New York 10006 Fax: (212) 225-3999 Attention: Richard S. Lincer

If to the Company or AMR, to:

AMR Corporation American Beacon Advisors, Inc. MD 5566 4333 Amon Carter Blvd. Ft. Worth, TX 76155 with a copy to (which shall not constitute notice):

Kelly Hart & Hallman LLP 201 Main Street Suite 2500 Fort Worth, TX 76102 Fax: (817) 878-9709 Attention: F. Richard Bernasek

or to such other address as such party may indicate by a notice delivered to the other party hereto.

13.5 Successors and Assigns.

(a) The rights of any Party under this Agreement shall not be assignable by such Party hereto prior to the Closing without the written consent of the other Parties, except that Buyer may assign any or all of its rights and interests hereunder to any Affiliate and Buyer may collaterally assign its rights to any lender providing financing in connection with the transaction contemplated hereby. Any such assignment shall not relieve Buyer of any obligations hereunder and the obligations of Buyer shall continue with respect to any such assignee.

(b) Except as provided above, following the Closing, neither Party may assign any of its rights hereunder to any third Person without the written consent of the other Party, except that either Party may assign its rights hereunder to an Affiliate. Any assignment hereunder (whether before or after the Closing) shall not relieve the assigning Party of its obligations hereunder. This Agreement shall be binding upon and inure to the benefit of the Parties hereto and their successors and permitted assigns.

(c) Other than the Parties and successors and assigns permitted by this <u>Section 13.5</u> and except as provided in <u>Section 11</u> with respect to the Indemnified Party, this Agreement is for the sole benefit of the Parties to this Agreement and their permitted successors and assigns and nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person, including any union or any employee or former employee of Seller or it subsidiaries, any legal or equitable right, benefit or remedy of any nature whatsoever, including any rights of employment for any specified period, under or by reason of this Agreement. Notwithstanding anything herein to the contrary, nothing contained herein shall (i) be treated as an amendment to any particular Welfare Plan, Pension Plan or other employee benefit plan or arrangement, (ii) obligate Buyer or any of its subsidiaries to (A) maintain any particular benefit plan or arrangement or (B) retain the employment of any particular employee of AA seconded to the Company, or (iii) prevent Buyer or any of its subsidiaries from amending or terminating any benefit plan or arrangement.

13.6 Access to Records after Closing.

(a) For a period of six (6) years after the Closing Date, AMR and its representatives shall have reasonable access to all of the books and records of the Company and the Subsidiary to the extent that such access may reasonably be required by AMR in connection with matters relating to or affected by the operations of the Company and the Subsidiary prior to the Closing Date. Such access shall be afforded by Buyer upon receipt of reasonable advance notice and during normal business hours. AMR shall be solely responsible for any costs or expenses incurred by it pursuant to this <u>Section 13.6</u>. If Buyer, the Company or the Subsidiary shall desire to dispose of any of such books and records prior to the expiration of such six-year period, Buyer shall, prior to such disposition, give AMR a reasonable opportunity, at AMR's expense, to segregate and remove such books and records as AMR may select.

(b) For a period of six (6) years after the Closing Date, Buyer and its representatives shall have reasonable access to all of the books and records relating to the Company and the Subsidiary and the Funds which AMR or any of its Affiliates may retain after the Closing Date. Such access shall be afforded by AMR and its Affiliates upon receipt of reasonable advance notice and during normal business hours. Buyer shall be solely responsible for any costs and expenses incurred by it pursuant to this <u>Section</u> <u>13.6(b)</u>. If AMR or any of its Affiliates to dispose of any of such books and records prior to the expiration of such six year period, AMR shall, prior to such disposition, give Buyer a reasonable opportunity, at Buyer's expense, to segregate and remove such books and records as Buyer may select.

13.7 Entire Agreement; Amendments.

This Agreement, the Exhibits and Schedules referred to herein, the documents delivered pursuant hereto and the Confidentiality Agreement contain the entire understanding of the Parties hereto with regard to the subject matter contained herein or therein, and supersede all other prior representations, warranties, agreements, understandings or letters of intent between or among any of the Parties hereto. This Agreement shall not be amended, modified or supplemented except by a written instrument signed by an authorized representative of each of the Parties hereto.

13.8 Interpretation.

Articles, titles and headings to sections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement. The Schedules and Exhibits referred to herein shall be construed with and as an integral part of this Agreement to the same extent as if they were set forth verbatim herein. Disclosure of any fact or item in any Schedule hereto referenced by a particular section in this Agreement shall be deemed to have been disclosed with respect to every other section in this Agreement but only to the extent it is reasonably apparent on its face that such disclosure is applicable with respect to such other section. Neither the specification of any dollar amount in any representation or warranty contained in this Agreement nor the inclusion of any specific item in any Schedule hereto is intended to imply that such amount, or higher or lower amounts, or the item so included or other items, are or are not material, and no Party shall use the fact of the setting forth of any such amount or is not material for purposes of this Agreement. Unless this Agreement specificatily provides otherwise, neither the specification of any item or matter in any representation or warranty contained in the ordinary course of business, and no Party shall use the fact of the setting forth or the inclusion of any such item or matter, or other items or matters, are or are not in the ordinary course of business, and no Party shall use the fact of the setting forth or the inclusion of any such item or matter in any dispute or controversy between the Parties as to whether any obligation of any such item or matter or there in the ordinary course of business for purposes of this Agreement nor the inclusion of any such item or matter, or other items or matters, are or are not in the ordinary course of business, and no Party shall use the fact of the setting forth or the inclusion of any such item or matter in any dispute or controversy between the Parties as to whet

13.9 Waivers.

Any term or provision of this Agreement may be waived, or the time for its performance may be extended, by the Party or Parties entitled to the benefit thereof. Any such waiver shall be validly and sufficiently authorized for the purposes of this Agreement if, as to any Party, it is authorized in writing by an authorized representative of such Party. The failure of any Party hereto to enforce at any time any provision of this Agreement shall not be construed to be a waiver of such provision, nor in any way to affect the validity of this Agreement or any part hereof or the right of any Party thereafter to enforce each and every such provision. Except as otherwise provided in <u>Articles IX</u> and <u>X</u>, no waiver of any breach of this Agreement shall be held to constitute a waiver of any other or subsequent breach.

13.10 Expenses.

Except as expressly set forth herein, each Party hereto will pay all costs and expenses incident to its negotiation and preparation of this Agreement and to its performance and compliance with all agreements and conditions contained herein on its part to be performed or complied with, including the fees, expenses and disbursements of its counsel and independent public accountants.

13.11 Partial Invalidity.

Wherever possible, each provision hereof shall be interpreted in such manner as to be effective and valid under Applicable Law, but in case any one or more of the provisions contained herein shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, such provision shall be ineffective to the extent, but only to the extent, of such invalidity, illegality or unenforceability without invalidating the remainder of such invalid, illegal or unenforceable provision or provisions or any other provisions hereof, unless such a construction would be unreasonable.

13.12 Execution in Counterparts.

This Agreement may be executed in one or more counterparts, each of which shall be considered an original instrument, but all of which shall be considered one and the same agreement, and shall become binding when one or more counterparts have been signed by each of the Parties hereto and delivered to the Company and Buyer.

13.13 Further Assurances.

On and after the Closing Date each Party hereto shall take such other actions and execute such other documents and instruments of conveyance and transfer as may be reasonably requested by the other Party hereto from time to time to effectuate or confirm the transfer of the Shares to Buyer in accordance with the terms of this Agreement.

13.14 Disclaimer of Warranties.

EXCEPT AS TO THOSE MATTERS EXPRESSLY COVERED BY THE REPRESENTATIONS AND WARRANTIES IN THIS AGREEMENT AND THE CERTIFICATE DELIVERED BY SELLER PURSUANT TO <u>SECTION 9.1</u> AND THE CERTIFICATE DELIVERED BY BUYER PURSUANT TO <u>SECTION 10.1</u> AND THE RESPECTIVE SCHEDULES RELATED THERETO, EACH OF THE COMPANY, SELLER AND BUYER DISCLAIMS ALL OTHER REPRESENTATIONS AND WARRANTIES WHETHER EXPRESS OR IMPLIED. Buyer acknowledges that neither Seller and the Company nor any of their representatives or any other Person has made any representation or warranty, express or implied, as to the accuracy or completeness of any memoranda, charts or summaries heretofore made available by Seller and the Company or their representatives to Buyer or any other information which is not included in this Agreement or the Schedules hereto, and neither Seller and the Company nor any of their representatives or any other Person will have or be subject to any liability to Buyer, any Affiliate of Buyer or any other Person resulting from the distribution of any such information to, or use of any such information by, Buyer, any Affiliate of Buyer or any of their agents, consultants, accountants, counsel or other representatives.

[Remainder of Page Intentionally Left Blank-Signature Page Follows]

933225

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the day and year first above written.

COMPANY:

AMERICAN BEACON ADVISORS, INC.

By: <u>/s/</u> Name: William F. Quinn Title: Chief Executive Officer

Signature Page to Stock Purchase Agreement

SELLER:

AMR CORPORATION

By: <u>/s/</u> Name: Thomas W. Horton Title: Executive Vice President and Chief Financial Officer

Signature Page to Stock Purchase Agreement

BUYER:

LIGHTHOUSE HOLDINGS, INC.

By: <u>/s/</u> Name: Kneeland Youngblood Title:

Signature Page to Stock Purchase Agreement

The following Exhibits and Schedules are omitted pursuant to Item 601(b)(2) of Regulation S-K. A supplemental copy of such Exhibits and Schedules shall be furnished to the Securities and Exchange Commission upon request.

Exhibits

Exhibit A	Transition Services
Exhibit B	[Reserved]
Exhibit C	[Reserved]
Exhibit D	Form of Client Consent Notice
Exhibit E	Form of ERISA Client Notice
Exhibit F	Form of Exempt Fund Client Notice
Exhibit G	Travel Privileges Agreement
Exhibit H	[Reserved]
Exhibit I	AAdvantage Participation Agreement

Schedules

3.3Net Working Capital5.2Qualifications to Conduct Business5.3Subsidiary and Investments5.4(b)No Conflicts5.5Financial Statements; Exceptions
5.3Subsidiary and Investments5.4(b)No Conflicts
5.6 Operations Since Financial Statements Date
5.7 Taxes
5.8 Governmental Permits
5.9 Assets Under Management
5.10 Real Property
5.11 Personal Property Leases
5.12 Intellectual Property Matters
5.14 Violation, Litigation or Regulatory Action of the Company
5.15 Contracts
5.16 Status of Contracts
5.17(a) Welfare Plans and Pension Plans
5.17(c) Compliance of Welfare Plans and Pension Plans
5.17(d) Other Employee Benefits
5.18 Employee Relations and Agreements
5.19 Insurance
5.20 Broker
5.21 Investment Adviser Registration
5.24(a) Fund Agreements
5.24(b) No Regulatory Action
5.24(g) Fund Taxation
5.27 Undisclosed Liabilities
5.28 Transactions With Affiliates
6.3 Violation, Litigation or Regulatory Action of Buyer
6.5 Broker
7.4(b) Operations Prior to Closing Date; Methodology for Pre-Closing Dividends
8.1(c) Buyer's Benefit Programs
8.1(d) Unused Vacation and Bonus Programs

AMR Corporation

July 15, 2008

Mr. Thomas W. Horton 4333 Amon Carter Boulevard Fort Worth, Texas 76155

Re: Employment Agreement Extension

Dear Tom:

You, AMR Corporation and American Airlines, Inc. are parties (the "Parties") to an Employment Agreement dated as of March 29, 2006 (the "Employment Agreement"). The Parties have agreed to extend the term of the Employment Agreement and to make other conforming changes as described in this letter.

The Employment Agreement is amended as follows:

- 1. In Section 2 of the Employment Agreement, the words "third anniversary" are replaced with the words "sixth anniversary".
- 2. The third and fourth sentences of Section 3(b)(iv) are deleted and replaced with the following: "The Executive shall be provided with one and one/third additional years of age and service credit for each year worked during the first three years of the Employment Period (for up to a maximum of 3.9 years of additional age and service credit) for all purposes of American's Supplemental Executive Retirement Program (the "SERP"), all with the effect that the Executive shall be deemed to have served continuously with American since August 1985. The additional age and service credit under the SERP shall not be provided if the Executive's employment is terminated by American for Cause or by the Executive without Good Reason during the first three years of the Employment Period."

Except as expressly amended by this letter, the Employment Agreement shall continue in full force and effect in accordance with its terms.

If you agree that this letter sets forth our understanding with regard to the extension of the term of the Employment Agreement, please sign this letter where indicated below.

AMR CORPORATION

By:_/s/ Gerard J. Arpey

AMERICAN AIRLINES, INC.

By: /s/ Gerard J. Arpey

Agreed to and Accepted:

THOMAS W. HORTON

Dated: ____

AMR CORPORATION Computation of Ratio of Earnings to Fixed Charges (in millions)

	Three Months Ended June 30,				Six Months Ended June 30,			
	2008		2007		2008		2007	
Earnings (loss): Earnings (loss) before income taxes	\$	(1,448)	\$	317	\$	(1,776)	\$	398
Add: Total fixed charges (per below)		397		459		806		938
Less: Interest capitalized Total earnings (loss) before income taxes	\$	8 (1,059)	\$	5 771	\$	13 (983)	\$	14 1,322
Fixed charges: Interest	\$	172	\$	220	\$	352	\$	447
Portion of rental expense representative of the interest factor		206		221		419		454
Amortization of debt expense Total fixed charges	\$	19 397	\$	18 459	\$	35 806	\$	37 938
Ratio of earnings to fixed charges		-		1.68		-		1.41
Coverage deficiency	\$	1,456	\$		\$	1,789	\$	

I, Gerard J. Arpey, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of AMR Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Gerard J. Arpey

Date: July 17, 2008 Gerard J. Arpey Chairman, President and Chief Executive Officer I, Thomas W. Horton, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of AMR Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

 Date: July 17, 2008
 /s/ Thomas W. Horton

 Thomas W. Horton
 Executive Vice President and Chief Financial Officer

AMR CORPORATION Certification Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Subsections (a) and (b) of Section 1350, Chapter 63 of Title 18, United States Code)

Pursuant to section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code), each of the undersigned officers of AMR Corporation, a Delaware corporation (the Company), does hereby certify, to such officer's knowledge, that:

The Quarterly Report on Form 10-Q for the quarter ended June 30, 2008 (the Form 10-Q) of the Company fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934 and information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: July 17, 2008 Gerard J. Arpey Chairman, President and Chief Executive Officer

/s/ Thomas W. Horton

/s/ Gerard J. Arpey

Date: July 17, 2008 Thomas W. Horton Executive Vice President and Chief Financial Officer

The foregoing certification is being furnished solely pursuant to section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code) and is not being filed as part of the Form 10-Q or as a separate disclosure document.